

# TSS 6220 Transport Stream Server

# **User Manual**

**5**5



ncore TSS 6220					
inin	Flue Admin Describes Alterna			Time: 14:09:54	Disk Usage: 69.3 GB / 3
Play Record Scheou	e riles Admin Reporting About				
Schedule Control Panel					
Physical Connector eth0					•
Add Schedule			Tx Bitra	le: 0.017 Mbps	•
Physical Connector eth1					0
Add Schedule			Tx Bitra	le: 69.243 Mbps	•
@ () IP (Schedule 1)	Current Source: GAP(starz_hevc_1080p30 👼	00:00:53	239.192.0.170:10001	4.500 Mbps	•
IP (Schedule 2)	Current Source: GAP(No Output)	00:00:00	239.192.0.200.10000	0.000 Mbps	۲
	Current Source: GAP(No Output)	00:00:00	239.192.0.201:10000	0.000 Mbps	۲
IP (Schedule 4)	Current Source: GAP(SILENT_BOB.trp)	01:38:24	239.192.0.205:10000	18.500 Mbps	٠
IP (Schedule 5)	Current Source: GAP(No Output)	00:00:00	239.192.0.202.10000	0.000 Mbps	۲
IP (Schedule 6)	Current Source: GAP(Null Packets)	00:00:00	239.192.0.203:10000	10.000 Mbps	٠
IP (Schedule 7)	Current Source: GAP(No Output)	00:00:00	239.192.0.204:10000	0.000 Mbps	۲
IP (Schedule 8)	Current Source: GAP(No Output)	00:00:00	239.192.0.207:10000	0.000 Mbps	۲
IP (Schedule 9)	Current Source: GAP(No Output)	00:00:00	239.192.0.208.10000	0.000 Mbps	۲
B D (Schedule 10)	Current Source: GAP(Null Packets)	00:00:00	239.192.0.206:10000	10.000 Mbps	۲



### Copyright

© 2019 Sencore, Inc. All rights reserved. 3200 Sencore Drive, Sioux Falls, SD USA www.sencore.com

This publication contains confidential, proprietary, and trade secret information. No part of this document may be copied, photocopied, reproduced, translated, or reduced to any machine-readable or electronic format without prior written permission from Sencore. Information in this document is subject to change without notice and Sencore Inc. assumes no responsibility or liability for any errors or inaccuracies. Sencore, Sencore Inc, and the Sencore logo are trademarks or registered trademarks in the United States and other countries. All other products or services mentioned in this document are identified by the trademarks, service marks, or product names as designated by the companies who market those products. Inquiries should be made directly to those companies. This document may also have links to third-party web pages that are beyond the control of Sencore. The presence of such links does not imply that Sencore endorses or recommends the content on those pages. Sencore acknowledges the use of third-party open source software and licenses in some Sencore products. This freely available source code can be obtained by contacting Sencore Inc.

### **About Sencore**

Sencore is an engineering leader in the development of high-quality signal transmission solutions for the broadcast, cable, satellite, IPTV, and telecommunications markets. The company's world-class portfolio includes video delivery products, system monitoring and analysis solutions, and test and measurement equipment, all designed to support system interoperability and backed by best-in-class customer support. Sencore products meet the rapidly changing needs of modern media by ensuring the efficient delivery of high-quality video from the source to the home. More information about Sencore is available at the company's website, <u>www.sencore.com</u>.

All trademarks and registered trademarks mentioned herein are the property of their respective owners.



# **Revision History**

Date	Version	Description	Author
2017/10/31	1.0	Manual for the Sencore TSS 6220	GAK
2018/01/05	1.2	Add Play, Record, Licensing, features to the TSS 6220 Manual	GAK
2018/07/20	1.3	Add Disaster Recovery and Time Delay Features	GAK
2019/03/13	1.4	Add ASI I/0 Features	GAK



## **Safety Instructions**

- Read these instructions
- Keep these instructions
- Heed all warnings
- Follow all instructions
- Do not use this apparatus near water
- Clean only with dry cloth
- Do not block any ventilation openings. Install in accordance with the manufacturer's instructions
- Do not install near any heat sources such as radiators, heat registers, stoves, or other apparatus (including amplifiers) that produce heat
- Do not defeat the safety purpose of the polarized or grounding-type plug. A polarized plug has two blades with one wider than the other. A grounding type plug has two blades and a third grounding prong. The wide blade or the third prong is provided for your safety. If the provided plug does not fit into your outlet, consult an electrician for replacement of the obsolete outlet.
- Protect the power cord from being walked on or pinched particularly at plugs, convenience receptacles, and the point where they exit from the apparatus.
- Only use attachments/accessories specified by the manufacturer.
- Unplug this apparatus during lightning storms or when unused for long periods of time.
- Refer all servicing to qualified service personnel. Servicing is required when the apparatus has been damaged in any way, such as power-supply cord or plug is damaged, liquid has been spilled or objects have fallen into the apparatus, the apparatus has been exposed to rain or moisture, does not operate normally, or has been dropped.
- Do not expose this apparatus to dripping or splashing and ensure that no objects filled with liquids, such as vases, are placed on the apparatus.
- To completely disconnect this apparatus from the AC Mains, disconnect the power supply cord plug from the AC receptacle.
- The mains plug of the power supply cord shall remain readily operable.
- **Damage Requiring Service**: Unplug this product from the wall outlet and refer servicing to qualified service personnel under the following conditions:
  - When the power-supply cord or plug is damaged.
  - If liquid has been spilled, or objects have fallen into the product.
  - If the product has been exposed to rain or water.
  - If the product does not operate normally by following the operating instructions. Adjust only those controls that are covered by the operating instructions as an improper adjustment of the controls may result in damage and will often require extensive work by a qualified technician to restore the product to its normal operation.
  - If the product has been dropped or damaged in any way.
  - The product exhibits a distinct change in performance.
- **Replacement Parts**: When replacement parts are required, be sure the service technician uses replacement parts specified by Sencore, or parts having the same operating characteristics as the original parts. Unauthorized part substitutions made may result in fire, electric shock or other hazards.



# SAFETY PRECAUTIONS

### There is always a danger present when using electronic equipment.

Unexpected high voltages can be present at unusual locations in defective equipment and signal distribution systems. Become familiar with the equipment that you are working with and observe the following safety precautions.

- Every precaution has been taken in the design of your TSS 6220 to ensure that it is as safe as possible. However, safe operation depends on you the operator.
- Always be sure your equipment is in good working order. Ensure that all points of connection are secure to the chassis and that protective covers are in place and secured with fasteners.
- Never work alone when working in hazardous conditions. Always have another person close by in case of an accident.
- Always refer to the manual for safe operation. If you have a question about the application or operation call Sencore for assistance.
- WARNING To reduce the risk of fire or electrical shock never allow your equipment to be exposed to water, rain or high moisture environments. If exposed to a liquid, remove power safely (at the breaker) and send your equipment to be serviced by a qualified technician.
- To reduce the risk of shock the TSS 6220 must be connected to a mains socket outlet with a protective earth ground connection.
- For the TSS 6220 the mains plug is the main disconnect and should remain readily accessible and operable at all times.
- To reduce the risk of shock and damage to equipment, it is recommended that the chassis grounding screw located on the rear of the TSS 6220 be connected to the installation's rack, the vehicle's chassis, the battery's negative terminal, and/or earth ground.



# **Table of Contents**

1	Introduction	9
2	Specifications	10
3	Getting Started	12
	3.1 Introduction	. 12
	3.2 Package Contents	. 12
	3.3 Installation	. 12
	3.4 Rear Panel Connections	. 14
	3.5 Power Connections - Installation	. 15
	3.6 Front Panel Features	. 15
	3.7 Obtaining the TSS 6220 IP Address	. 17
	3.8 Unit Networking and DNS Conliguration	. 18
	3.10 Simplified Startup - Getting a Stream Plaving	. 20
٨	Diav Control Danel	28
4	1 Day Control Panel Overview	20
	4.2 Stream/PCAP Information Overview	29
	4.3 Plaver - Adding a Transport Stream - IP	. 30
	4.3.1 Add Transport Stream - IP - General Configuration	. 31
	4.3.2 Add Transport Stream - Advanced IP Configuration	32
	4 3 3 Add Transport Stream – IP - File Configuration	- <u>-</u>
		. 00
	4.4 Adding a PCAP Play File	. 34
	4.4.1 Add PCAP - General PCAP Play Settings	. 35
	4.4.2 Add PCAP - Advanced PCAP Settings	. 36
	4.4.3 Add PCAP - File Play Settings	. 36
	4.5 IP Stream and PCAP Status & Configuration Information	. 37
	4.6 IP and PCAP Monitor Panel	. 40
	4.7 Play Control Panel - ASI Output	. 41
	4.7.1 Add Transport Stream - ASI - General Configuration	. 42
	4.7.2 Add Transport Stream - File - Configuration	. 43
5	Schedule Panel	44
	5.1 Schedule Panel Overview	. 45
	5.2 Schedule - Information Fields	. 46
	5.3 Adding or Creating a Schedule Configuration - General	. 47
	5.3.1 Add Schedule - General Configuration	. 47
	5.3.2 Add Schedule - Advanced IP Configuration	. 49
	5.3.3 Add Schedule – ASI - General Configuration	. 50
	5.4 Schedule Configuration & Information Window	. 51
	5.5 Scheduling Configuration Panel	. 52
	5.5.1 Scheduling Configuration Panel - Colors	. 53
	5.5.2 Scheduling Configuration Panel – Moving Cursor	. 54
	5.5.3 Scheduling Configuration Panel - Loading Event Lists	. 54
6	Delay Viewing Panel	55
	6.1 Delay Panel Overview	. 56
	6.2 Delay Panel – Receive Fields	. 57
	6.3 Delay – Butter Information Fields	. 58
	0.4 Delay - Halistill Information Fields	. 59



	<ul> <li>6.5 Delay – Adding or Creating a Delay Line</li> <li>6.5.1 Add Delay – General Configuration - IP</li> </ul>	59 60
	6.5.2 Add Delay – Receive Configuration - IP	61
	6.5.3 Add Delay – Transmit Configuration - IP	62
	6.5.4 Add Delay – General Configuration – ASI In/Out	64
	6.5.5 Add Delay – Receive Configuration – ASI In/Out	65
	6.5.6 Add Delay – Transmit Configuration – ASI In/Out	66
7	<ul> <li>6.6 Delay Buffer – Extract Buffer To File</li></ul>	68 70 71 72 73 <b> 75</b> 76 77 77
	7.2.2 Disaster Recovery Panel – Buffer Descriptions	78
	7.2.3 Disaster Recovery Panel – Transmit Descriptions	79
	7.3.1 Adding a Disaster Recovery – General Menu	80
	7.3.2 Adding a Disaster Recovery – Receive Menu	82
	7.3.3 Adding a Disaster Recovery – Transmit Menu	84
	7.3.4 Adding a Disaster Recovery – General Menu – ASI In/Out	86
	7.3.5 Adding a Disaster Recovery – Receive Menu – ASI In/Out	88
	7.3.6 Adding a Disaster Recovery – Transmit Menu – ASI In/Out	89
	<ul> <li>7.4 Delay Buffer – Extract Buffer To File</li> <li>7.5 Disaster Recovery – Added Panel Status &amp; Configuration Menus</li> <li>7.5.1 Disaster Recovery Receive – Status Information Menu</li> </ul>	90 92 93
	7.5.2 Disaster Recovery Buffer – Configuration Menu	94
	7.5.3 Disaster Recovery Transmit – Status & Configuration Information Menu	95
~	7.6 Understanding Disaster Recovery	96
8	<b>FIIE VIEWING Panel</b>	98
	8.2 File Transfer Management - User Name and Password	101
	8.3 Managing Play Files & Folders	102
g	8.4 File viewing Panel – Filter by Type	. 102 103
J	9.1 Record Panel Overview	104
	9.2 Record – Information Fields	105
	9.3.1 Record Add Transport Stream IP Configuration - IP	106
	9.3.2 Record Add Transport Stream - IP Configuration	109
	9.4 Recording Input PCAP - Configuration 9.4.1 Record Add PCAP - Stream Configuration	111 111
	9.4.2 Record Add PCAP - PCAP Configuration	113
	9.5 Recording Input ASI - Configuration	114



9.5.1 Record Enable ASI Record – Add ASI Configuration	14
9.5 Record Status & Configuration Information1	17
9.6 IP – PCAP - ASI Monitor Panel 12	21
10 Admin	23
10.1 Changing Unit Password12	24
10.2 Profile Manager	24
10.3 SNMP MIB Files 12	25
10.4 Diagnostics12	25
10.5 Update the Unit Software Version12	26
10.6 Reboot Unit	27
10.7 Reset Unit to Factory Defaults 12	27
10.8 General Configuration	27
10.9 Network Port Configuration	28
10.10 Licensing Configuration	29
10.11 Date/ Lime Configuration	30
10.12 SNMP Communities	32
10.13 SNMP Trap Managers	32
10.14 Sysiog Configuration	33
11 Reporting Panel	34
11.1 Active Alarms	34
11.2 Event Logs	35
11.3 Configuring the Logs	36
12 About Panel13	38



# 1 Introduction

The TSS 6220 Transport Stream Server provides robust streaming capabilities for customers looking for a simple solution to creating automated playout of media files into multiple channels. With the onboard, redundant storage and FTP & SMB file upload, the unit can take stored media files and play them out according to user-supplied schedules. Up to 16 independent schedules and corresponding MPEG over IP outputs can be maintained simultaneously.

The TSS 6220 provides an easy to use and cost-effective solution for schedule playout of media files for a variety of applications. An optional record feature provides TS stream and PCAP recording capabilities. An optional play feature is also available to provide individual TS stream and PCAP outputs.

The TSS 6220 further offers a time-delay option for a single stream or multiple output streams simultaneously. A unique disaster-recovery option provides a cost-effective backup of primary broadcast systems. These intuitive time-shifting and long-term storage /replay capabilities come with incredibly simple setup, configuration and status monitoring. Enhanced features like multiple delays from the same buffer and automated disaster-mode activation make the TSS 6220 usable in a huge variety of applications.

The intuitive web UI and full SNMP or web API remote control allows the TSS 6220 to offer users simple, reliable and powerful solution for operational and lab environments. The TSS 6220 comes in a rack mountable 1-RU chassis with multiple output ports and has numerous expansion options available. A 2-RU or 3-RU chassis is required for some configurations.

This manual provides startup and operational information. It is written for professional operators of video distribution systems and assumes a prerequisite level of technical knowledge.



# 2 Specifications

#### **INTERFACES PHYSICAL**

Included IP Ports:	2x RJ45 1Gbps (Each port can be used for streaming and/or
	management)
Optional IP Ports:	Additional 2x RJ45 1Gbps, Fiber 2x SFP 1/10Gbps
ASI I/O:	4x ASI ports (750hm BNC) each configurable as input or output

#### **IP FORMATS INPUT AND OUTPUT**

IP Input Formats:	UDP or RTP - RTP Header Extensions Supported
IP Output Formats:	UDP or RTP
IP Encapsulation:	1 to 7 TS Packets per IP Packet
IP Addressing:	Unicast or Multicast
IGMP Compatibility:	Version 1, 2 & 3
IP Bitrates:	250 Kbps to 200 Mbps
File Types:	Transport streams (.ts, .trp)
	PCAP Ethernet capture (.pcap)

#### MANAGEMENT

Protocols:	HTTP and SNMP
User Interfaces:	Full control via web GUI
Automation Interfaces:	Full status and control via SNMP, Configurable SNMP traps, Web services API available, Syslog message logging
Firmware Updates:	Via web GUI

#### POWER

Voltage:	100-240V
Frequency:	50-60Hz
Power Redundancy:	Dual, hot-swappable supplies on TSS 62221/62222

#### **CHASSIS OPTIONS**

TSS 62220: 1RU chassis suitable for streaming, scheduled playlists and simple recording TSS 62221: 1RU chassis suitable for time-delay, disaster-recovery and extended recording. TSS 62222: 2RU chassis suitable for time-delay, disaster-recovery and extended recording.

Size: Rack-mount 1 RU chassis (2 RU for the TSS 62222 option) 1RU Chassis Depth: 20 inches (507 mm)

Physical chassis dimensions and operating conditions vary depending on chassis and storage selection



#### **STORAGE OPTIONS**

SSD Hard Drives:

Intended Use: Streaming and playlists High-performance recording Cumulative Performance: 1Gbps+ for streaming or 300-400Mbps for time-delay and disaster-recovery

Redundancy Configuration: RAID-5 for time-delay and disaster-recovery SAS Hard Drives:

Intended Use: Long-term storage for time-delay disaster-recovery and recording Cumulative Performance: 200-250Mbps for time-delay and disaster-recovery Redundancy Configuration: RAID-6 for time-delay and disaster-recovery

Specifications are subject to change without notice.



# **3 Getting Started**

### 3.1 Introduction

This section provides an overview of what is included with your TSS 6220. It provides critical information on obtaining the unit's IP address in which to gain access to the unit's web interface GUI. It provides front/rear panel descriptions and information on getting the unit mounted in a standard rack.

## 3.2 Package Contents

The following is a list of the items that are included with the TSS 6220 unit:

1. Startup Guide

- 2. AC Power Cord
- 3. Rackmount Rails

#### Quick Start Guide TSS 6220 Sencore About This Quick Start Guide 2 1 This guide is for helping new customers get their Sencore TSS 6220 setup and 3 running as quickly and easily as possible The TSS 6220 Transport Stream Server is the perfect solution for simple scheduling and playout of media files for channel-in-a-box or file-to-live applications. It can also be used for creating signals to aids in the design, verification, manufacturing, and deployment of digital TV equipment and systems. With its intuitive web interface and the ability to handle multiple 4 schedules in one device, the TSS 6220 is the ideal tool for customers that need to quickly get their channels on air. Figure 1: Rear Port Connector Download the Full User's Manual 1. Power Connector: Provides AC power connection for powering the FOWEL Confection: Instructure pression of the system system IPMI Port: Can be used for remote server management, but not normally used in the TSS 6220 operation eth0 Port: Network port for management and streaming. eth1 Port: Network port for management and streaming. eth2 Port: Network port for management and streaming. eth2 Port: Network port for management and streaming. eth3 Port: Network ports PP ports WGA Monitor Cutrut The current versions of the full user's manuals can be obtained from the Download tab of individual product pages on our website: www.sencore.com or by emailing Sencore ProCare support at procare@sencore.com. Unpack the Equipment Check that there are no signs of damage to equipment due to transport. If the equipment appears to be damaged, please contact Sencore ProCare for support. In addition to the Sencore TSS 62220 server hardware itself, the box should include a power cables and various rackmount hardware Establish a Web Connection to the TSS 6220

If any of these items were omitted from the packaging of your unit, please call 1-605-978-4600 to obtain a replacement.

## 3.3 Installation

The TSS 6220 rackmount chassis is a 1 RU or 2 RU enclosure that is designed to be mounted in a standard 19" equipment rack. The unit is shipped with the mounting hardware necessary to safely secure the chassis in the rack. This includes rackmount ears and screws for the front along with rails to support the sides. The user is advised to always use the included rails for mounting.



The TSS 6220 is designed for front-to-back ventilation. Care must be taken to ensure that this ventilation is not impeded in any way.

To install the TSS 6220 into a rack use the following steps:

- Determine the desired position in the rack for the TSS 6220 making sure that the air intake on the front of the unit and the exhausts on the rear of the unit will not be obstructed.
- 2. Insert the rack mount clips into place over the mounting holes in the rack.
- 3. Slide the TSS 6220 into position in the rack.
- 4. Secure the TSS 6220 to the rack by installing the four screws through the front mounting holes and tightening.

WARNING: To prevent injury, the apparatus must be securely attached to the floor/wall in accordance with the installation instructions.



The TSS 6220 is designed for front-to-back ventilation. Care must be taken to ensure that this ventilation is not impeded in any way.



## 3.4 Rear Panel Connections

All of the external connections for the TSS 6220 are located on the rear of the unit. These connections include standard computer I/O (Monitor, USB, parallel, serial, audio, and LAN ports) and the TSS 6220 IP output connections. The power connection is also located on the rear of the unit. The following provides an overview description of the rear panel connections and features.



Descriptions:

- 1. Power Connector: Provides AC power connection for powering the system
- 2. IPMI Port: Can be used for remote server management, but not normally used in the TSS 6220 application for operation
- 3. eth0 Port: Network port for management and streaming/recording
- 4. eth1 Port: Network port for management and streaming/recording
- 5. eth2 Port: Port available only with the addition of optional card adding either 2x RJ45 or 2x SFP ports
- 6. eth3 Port: Port available only with the addition of optional card adding either 2x RJ45 or 2x SFP ports
- 7. VGA Monitor Output
- 8. USB Interface Ports
- 9. RS-232 Port

The TS6220 may require a 2RU chassis if your hardware requirements include the need for an ASI input/output interface and/or added Raid Drive configurations. Below is an example of the 2 RU TSS 6220 with added ASI input/output hardware.



Added Descriptions:

1. ASI input/output interface board/ports: Optional hardware providing input and output ASI ports. The ports are configurable as inputs or outputs in the application software.



### 3.5 Power Connections - Installation

Using the proper power connections is vital to the safe operation of the TSS 6220. Only use the supplied 3-prong power connectors or those with equal specifications. NEVER tamper with or remove the 3rd – prong grounding pin on any cord. This could cause damage to the unit, personnel, or property.

The TSS 6220 is intended for use on either 120V or 240V systems. The power supply will automatically detect the voltage to which it is connected. There is one power supply and one AC power cord on 1 RU chassis TSS 6220 models. 2RU or 3RU units may have dual AC power supplies. To connect AC power, perform the following:

- 1. Locate the proper AC power cord(s).
- 2. Plug the female end of the power cord (end with no prongs) into the back of the unit.
- 3. Plug the male end of the power cord into a proper protected AC outlet.
- 4. Repeat above for units equipped with dual AC supplies.

## 3.6 Front Panel Features

The front panel contains some pushbuttons and indicator lights. This section provides an overview of these features.



\*NOTE: Some design modifications may occur in which features shown in this image and described below may not be found on your TSS 6220.

Front Panel Descriptions:

- 1. UID: Pushbutton: This pushbutton provides an identification light at the rear of the unit. Press the UID pushbutton to turn on a blue light at the rear of the unit. This makes it easy to identify the unit when viewing from the back of the equipment rack. The "I" light on the front panel (#8) indicates when the UID is switched on.
- 2. Reset Pushbutton: This pushbutton provides a reset of the system. Press and release to initiate a reset operation of the operating system.



- 3. Power Pushbutton: Turns the unit AC power on and off. Hold down for 3 seconds and release to power the unit on or off.
- 4. Power Light: Indicates when the unit is powered on for normal operation.
- 5. Hard Disc Drive Light: Indicates normal OS unit disc drive activity.
- 6. Network Port Indicator Light: Indicates normal unit network port data activity.
- 7. Network Port Indicator Light: Indicates normal unit network port data activity.
- 8. i Indicator Light: This light provides an indication when the UID feature is turned on illuminating a blue light at the rear of the unit.



## 3.7 Obtaining the TSS 6220 IP Address

Operating the TSS 6220 depends on gaining access to the unit's network GUI as there is no front panel interface. Connecting to the unit's web GUI using a web browser requires knowledge of the IP address of the port in which to enter into the address field of a web browser. To acquire and or to change the port's IP address from the factory default requires viewing a computer monitor connected to the unit along with the use of a keypad to navigate the provided menus.

To obtain or make changes to the port's IP address:

- 1. Connect a computer monitor to the VGA connector and power it on. Reference Section 3.4 in this manual – (#7) rear panel port.
- 2. Connect a PC keyboard (USB type) to the rear panel USB connectors. Reference Section 3.4 in this manual – (#8) rear panel USB connection ports.



- 3. With power properly connected to the TSS 6220, push and release the front panel power
- button. Allow some time for boot up. The system starts up with no user or password entry required. A Main Menu screen appears on the monitor
- 4. The Ethernet port 0 (eth0) and Ethernet port 1 (eth1) IP addresses are shown on the initial Main Menu. Enter the IP address of the port you are using into your web browser. See the next sections (3.7, 3.8) of this manual for more details on changing the IP address and using the web GUI.



5. If the IP addresses are not shown, press the ENTER key on the keyboard to advance to the next menu. Press the Up or Down arrow keys on the keyboard to select the eth1 or

eth0 Adapter Status listing by Moving the cursor in front of the selection. Note: If no cursor is seen, press ENTER a second time.

Press Enter on the keyboard to advance to the configuration menu if you wish to change the IP mode (static or DHCP) or to change IP Address, Subnet, or Gateway addresses. See the next section in this manual for additional information.



In some instances, you may be able to use the factory default IP addresses to gain web access. The following default IP addresses are used.

Factory Default Settings:

eth0: DHCP eth1: 10.0.0.61 eth2: 10.0.0.62 (with added optional ports)



## 3.8 Unit Networking and DNS Configuration

The TSS 6220 network configuration is managed with setup menus visible by connecting a computer monitor and USB keypad to the system. The configuration menus provide entry of host names, gateway and DNS server addresses. Setup of the unit's Ethernet ports including if they are static or DHCP configured. If static, entry of the ports IP address, gateway and subnet may be entered. This section shows the typical setup menu features and describes how to setup the network settings.

Viewing the setup utility screens provided by the TSS 6220 requires connecting a computer monitor to the VGA video output connector on the rear of the unit. It requires the use of an USB keypad. View the monitor screen and use the keypad for navigation and entry. The following navigation and entry rules are used and included as a reminder on each setup screen.

Press [Enter] to confirm a selection – advance to next menu/selection Press [Esc] to go back to previous menu and accept entries Press [Left] and [Right] arrow keys to navigate Press [Up] and [Down] arrow keys to navigate Press number and letter keys for field input values Press [Del] and [Backspace] as needed for text entry

### **Configure Networks Settings**

Network configuration settings are available to enter a unit host name, define a gateway and enter primary & secondary DNS addresses. These selections are available in the "Configuration Networks" menu.

- 1. From the opening menu press the ENTER key.
- 2. Position the cursor in front of Configure Networks selection as in illustration below. Press the ENTER key to advance to the Network Configuration Menu.



- 3. To enter a Host Name: Position the cursor in front of the Host Name row using the up and down arrow keys. Press ENTER. The first field is selected for letter or number entry. Enter the number or letter. Press left or right arrow key to move to the next digit. Press ENTER key.
- 4. To enter a Default Gateway: Position the cursor in front of the Default Gateway row. Press Enter. Enter the desired gateway. Press ENTER.
- 5. To enter a Primary and/or Secondary DNS: Position the cursor in front of the Primary DNS and/or Secondary DNS row to be changed or entered. Press the ENTER key. Enter the address using number keys while navigating with the arrow keys. Press ENTER.



### Ethernet 0 and Ethernet 1 Port Configuration Changes

Network configuration settings are available to enter a static IP address, gateway address and subnet mask for Ethernet port 0 (eth0) and Ethernet port 1 (eth1). These selections are available in the respective "eth1 Adapter Status" menu and "eth0 Adapter Status" menus. The following describes the steps to enter or change a port's static IP address.

- 1. From the start menu, press the ENTER key to advance to the Unit Networking screen.
- 2. Press the Up or Down arrow key as needed to move the cursor position in front of the eth0 or eth1 row. Press the Enter key to advance to the Adapter Eth Status menu.
- 3. Use the up and down arrow keys to position the cursor in front of the IP Mode row. If it shows DHCP Press the ENTER key to select the setting field. Click the up or down arrow key to increment setting to "Static." Press ENTER.



- Use the up and down key to position the cursor in front of the IP Mode row. Press ENTER to enter the field. Enter number values to define the desired static IP address. Use the left and right arrow keys to move to different digits. Press ENTER when complete.
- 5. Change Mask and Gateway settings using the same technique as in step 4. Press the ENTER key when finished to accept entries and return to the menu.



## 3.9 Controlling the TSS 6220 Using the Web GUI

Controlling the TSS 6220 is done by a network connection to either the eth0 or eth1 ports and the use of the web interface or GUI. From any PC that is connected to the same network as the TSS 6220, open a web browser application and type the IP address of the unit in the address field. You must obtain the unit IP address with the procedure in the Startup section 3.7 on page 17.

When connecting to the web GUI, you are greeted with a Login screen as shown below. The default user field entry is admin and the default password field is left blank or no entry. Press the login button in order to login to the web interface. The User and Password can be changed for improved security in the Admin section of the GUI. Please see section 10.1 in this manual.

👗 Login	
User: Password:	admin 👻
	🔑 Login

When connected you are greeted with the home or start page of the web GUI as shown below. Depending on the options and licensing some variations in the screen may be seen.

To open the TSS 6220 web interface use one of the following supported browsers.

- Internet Explorer version 9 or newer
- Firefox
- Google Chrome
- Microsoft Edge

S TSS 6220	× +									
( 10.0.7.21/#productTabs		C Q Sea	rch			☆	Ó	Ø 1	ł ń	≡
<b>Sencore</b>	TSS 6220									
Logged in as: admin				Disk Usage:	3.8 GB / 3.7	тв	CPL	13%	۵ ۱	ogout
	Play Schedule Files Admin Reporting About									
	Play Control Panel									
	Physical Connector eth0				0					
	O Add Transport Stream O Add PCAP		Tx Bitrate	c 0.011 Mb;	ps 🔘					
	Physical Connector eth1				۵					
	Add Transport Stream     Add PCAP		Tx Bitrate	: 110.943 Mbp	ps 🕚					
	a 🔯 (i) (i) IP (Stream 1) balloons_hevc_1000p60_420 🗂 00:00:52 23	9.192.0.111:10000	0	34.000 Mbp	ps 🧶					

The user is capable of configuring parameters from this page by clicking on the selection tabs at the top of the page below the Sencore TSS 6220 header. User configuration changes are offered in each section by clicking on the (cog) which represents a settings configuration is available for the listed item. Each section contains a dropdown icon which is used to collapse or expand a section to see additional details. Further details of the common fields in the web GUI of the TSS 6220 are described

The top section or fields of the TSS 6220 web GUI includes login and operational information of the system. Below is a reference to the information provided by each field.





- 1. Logged in as: This field indicates the logged in user. This field may not be selected or changed. The User and Password may be set in the Admin tab. See section 10.1.
- 2. Selection Tabs: Provides operational control of the TSS 6220. Click on a tab to select. The tabs shown and available for selection depend on unit licensing.
- 3. Storage Disk Usage: Indicates actual disk size that is currently in use
- 4. Storage Disk Usage: Indicates total disk size available for use
- CPU: Indicates the percentage of CPU usage
   Logout: Provides quick logout, click to log out from the MIP GUI.

Field		<b>Button/Selections</b>	Description				
1.	Logged in as:	Admin (Default)	Shows the logged in user name. A view only field.				
2.	Selection	Play	Provides control for playing stream and PCAP files				
	labs	Files	Provides viewing of media disc play files				
		Record	Provides record captures of TS streams or PCAP				
		Schedule	Provides for schedule playout of TS streams				
		Delay	Provides buffer and delayed output of TS stream				
	Disaster Recovery Admin Reporting About Provides buffer delaye recovery playout Provides administrative Provides reporting and	Provides buffer delayed output and disaster					
		Admin	recovery playout				
		Reporting	Provides administrative tasks				
		About	Provides reporting and logging				
			Provides unit information				
3.	Disk Usage	Example shown: 63.0 GB	Indicates the disc drive memory space that is currently being used by the system				
4.	Disk Usage	Example shown: 3.7.TB	Indicates the total disc drive memory space available for TS and PCAP stream storage				
5.	CPU	Example shown: 13%	Indicates percentage of CPU processing capacity that is in use.				
6.	Logout	Logout Click on icon	Click to log out of the web GUI connection to the TSS 6220				



## 3.10 Simplified Startup - Getting a Stream Playing

This section provides a quick start process to assist you in getting an output stream playing to the Ethernet 1 port. This section provides only enough information to get you outputting a TS stream. It is not intended to completely summarize all the information contained in this manual. Please reference other portions of the manual to answer questions and become familiar with the TSS 6220 and its features.

### Get Unit Powered & Network Connected:

- 1. Locate the TSS 6220 in a convenient location or mount in an equipment rack in which AC power and a connection to the network is closely available. Connect an AC cord from the rear of the unit to the AC outlet.
- 2. Connect a network cable between the Ethernet port 1 (Port nearest the VGA video connector on rear panel) and to your network.
- Momentarily press the Power button on the unit front panel. You will observe and hear normal PC startup activities. The front panel power light should illuminate and drive light activity indicated. Wait while the unit boots up.

### Establishing a Web Connection with the Pre-Set Management IP Address

To access the TSS 6220 web user interface, it is necessary to establish an Ethernet connection to the device. There are two alternative ways to connect to the TSS 6220's management IP address: 1.) Use the pre-set management IP address or 2.) Connect a monitor and keyboard to the server to retrieve the IP address. The following steps describe how to use a pre-set management IP address on a PC via a connection to the Eth1 port.

The TSS 6220 is shipped with the following factory settings for the Eth1 network ports.

Eth1 Default IP address: **10.0.0.61** 

You can connect to the web UI of the TSS 6220 using a PC and connecting directly from the PC's network port to the Eth1 port with an Ethernet cable. Configure the PC's network port settings to permit a direct connection.

For Windows, the network parameters are set in the Control Panel — Network and Internet — Network and Sharing Center — Network Connection — Properties — Internet Protocol Version 4 Properties viewing menu.



Select the user defined address, and set the PC's IP address to 10.0.0.60 and the subnet mask to 255.255.0.0.

When the IP address of the PC has been set in the same subnet as the TSS 6220's factory setting, a web browser can be used to access the web user interface at 10.0.0.61.



#### **Retrieve and/or Configure the Unit IP address**

You can determine the TSS 6220's IP address of both Eth0 and Eth1 ports by connecting a monitor and keyboard to the server. All configuration of the TSS 6220 is done via a PC and web. The Ethernet port 1 can serve both as a management port for web browser access and as the streamer IP/PCAP network port. The following section describes how to connect a monitor and keyboard to the server to retrieve and/or change the unit's IP address on both ports. To connect to the unit's web browser:

 Connect a computer monitor to the TSS 6220. Connect a VGA video cable from the VGA connector at the rear of the TSS 6220 to the computer monitor. Power up the computer monitor. Connect a PC USB keyboard to one of the USB ports at the rear of the TSS 6220. The display screen on the monitor will indicate the current IP address for the Ethernet 1 port (eth1). It also shows the IP address for the other Ethernet 0 port (eth0)



 If you do not have a computer monitor you may try to access the web GUI as described in the previous page using the factory default IP settings shown below: Ethernet port 0 (eth0): 10.0.0.61 Ethernet port 1 (eth1): 10.0.0.62

### Use PC Web Browser to Connect to the TSS 6220's Web GUI

1. Connect a PC to the same network as the TSS 6220. Open a web browser application on the PC and in the top address field enter the IP address of the eth1 port as found in the previous step.

👗 Login	
User: Password:	admin 👻
	🔑 Login

 The Login screen appears with a successful connection. The default User name is admin and the default password is blank or no entry. Click on the Login field.

When successfully logged in the web connection to the GUI produces a Schedule Control Panel. A Play Control and/or Record Panel may be shown depending on unit licensing.

Sencore TS	S 6220								
Logged in as: admin	Schedule Files A	Imin Reporting About				Time: 15:58:	14 Disk Usage:	61.5 GB / 3.7 TB	CPU: 13%
	Schedule Control Panel								
	Physical Connector eth0								
	Add Schedule					Tx Bitrate:	0.013 Mbps	۲	
	Physical Connector eth1							$\odot$	
	O Add Schedule					Tx Bitrate:	128.510 Mbps	۲	
	IP (Schedule 1)	Current Source: SHREK.trp	3	00:00:37	239.192.0.170:10001	٢	18.500 Mbps		
	IP (Schedule 2)	Current Source: GAP(No Output)	2	00:00:00	239.192.0.200:10000	٢	0.000 Mbps	۲	

#### Use FTP or SMB to Load Stream/PCAP Files to Media Storage Drives

The TSS 6220's media drives are available to store files used for playout in a schedule. Click on the "Files" tab at the top of the GUI to view available files. If a file is available for selection, click on the Schedule tab and skip to the instructions "**Creating a Play/Event Schedule**" instruction section.



If no files are available you will need to access and use FTP or SMB to load files to the media drive. Read on in this section for details on accessing FTP and loading files/folders. Using FTP requires starting an instance of Windows Explorer on your PC and entering the address of the FTP server. Follow the instructions below.

Files Admin Reporting	About						
Files Control Panel							
To upload or manage files go to 20142101540							
← → 1 / Search root							
> 🚔 met	File Name	Туре	Size	Bitrate	Duration		
	Ballons_Nevc_1000p30_420_0004_12M0ps_L4.0_MT_3.0p	19	100.7 MB	14.00 Whom	00.01.00		
	H Balcons_seamless_0000.3p	39	610.4 MB	19.39 Https	00:04:24		
	📴 608-T	Folder					

The FTP server address is located in the File Viewing Panel (Files tab) in the TSS 6220's web GUI. Look near the top of the page under the Files Control Panel header. (See circle in illustration) Click at the end of the field and drag the mouse to highlight the ftp address. Copy the address – (Keyboard Ctrl-C key sequence).

	Sche	dule	Files	Admin	Reporting	About	
Files Co	ontrol I	Panel					
File 1	fransfer	r Manaç	gement	To upload	or manage files	s go to ftp://1	0.0.7.112
	1	1					
- v 👝 r	toot				File Name		

Open a Windows Explorer application window on your PC. In the header of this application -

paste in the ftp address listed in the TSS 6220's File menu on the second line below the Files Control Panel heading. Paste the ftp address – (Ctrl-V) as shown in the example figure. Example shown: ftp://10.0.15.60. Press the Enter key on your keyboard.

○○○ (≒ ftp://10.0.15.60)	
File Edit View Tools Help	
Organize 🔻 New library	
✓ ★ Favorites ■ Desktop	Libraries Open a library to

Upon connection to the ftp server you will be prompted for a username and password: Below is the default User and Password. *The Username/Password can be set - see section 8.2 for instructions.* 

 Log On As
 ►

 Image: Second S

User: root Password: Note: The Password field is left blank or no entry as shown.

Click on the Log On field to access the FTP server.

Upon connection to the ftp server you can view the current folders and play files in the TSS 6220.

To transfer files to the ftp server, you may use common window's based file copy and paste techniques or drag and drop techniques. For example, open a



2<sup>nd</sup> version of Windows Explorer application. (Right mouse click on the Windows Explorer icon at the bottom system tray – click on Windows Explorer) You now have both the FTP server window and the Windows Explorer application running. Position both the FTP window and the Windows Explorer windows beside each other on your PC screen. See example below. Click on a file or



folder in the Windows Explorer window and drag it into the right side of the FTP server window. The folder or file is transferred as illustrated below.

🚱 🔵 🗢 📔 🕨 Kropuenske, (	Sien 🕨 My Videos 🕨	G  ▼					
File Edit View Tools He	lp	File Edit View Tools Help					
Organize 👻 🛕 Play with	VLC media player 👻 Share with 💌 Burn New folder	Organize 🕶					
Facorites     Destop     Recert Places     Recert Places     Kopuenske, Glen     Downloads     Downloads     Downloads     Destop     Downloads     Destop     Dewnloads     Destop     D	2013-07-09 00.02 Sencero TU-228, FKYpert Stream/gett.mp4     2015-06-12 020 FF Meter Training- SLM4/96/LMmp4     CLORADO.TRP     COLORADO.ap     Generator L, Inia 7, Colof BarePattern, SNR, IMG, DLS1.MOV     Generator L, Inia 7, Colof BarePattern, SNR, IMG, DLS1.MOV     Generator L, Inia 7, Substantian Stream, Sterk, Sterk, State J, Mod, DLS3.MOV     Generator L, Inia 7, Substantian, Sterk, MCL, Sterk, Sterk, State J, Mod, DLS3.MOV     Generator L, Inia 7, Substantian, Sterk, MCL, Sterk, MCL, State, Sterk, Sterk, State J, Mod, DLS3.MOV     Generator L, Inia 9, Duekstater Sterk, Sterk, Mod, DLS3.MOV     Generator L, Inia 9, Duekstater Sterk, Sterk, Mod, DLS3.MOV     Generator L, Inia 9, Duekstater Sterk, Sterk, Mod, DLS3.MOV     Generator L, Inia 5, Sterk, MCL, State, Sterk, McL, DLS3.MOV     Generator L, Inia 5, Sterk, MCL, State, Sterk, McL, DLS3.MOV     Generator L, Inia 5, Sterk, MCL, State, Sterk, McL, DLS3.MOV     Generator L, Inia 5, Sterk, MCL, State, MCL, MLS3.MOV     Generator L, Inia 5, Sterk, MCL, Sterk, MCL, MLS3.MOV     Generator L, Inia 5, Sterk, MCL, MLS3.MOV     Generator L, Inia 5, Sterk, MCL, MLS3.MOV     Generator L, Inia 5, Sterk, MCL, MLS4.MOV     Generator L, Inia 5, Sterk, MCL, MLS4.MOV     Generator L, Inia 5, Sterk, MCL, MLS5.MOV     Music, Choice, BUR.s     A Teta J, J, SKpertRecordinade Ip     Tari-CartURE-SNLOCELs     A Teta J, SKP, Sterk, MCCRELs     A Teta J, SKP, Sterk, MCCRELs     A Teta J, SKP, Sterk, MCCRELs     A Teta J, SKP, MCL, MLS5.MOV     A Sterk, MCL, MLS5.MEL, MLS5.MEL, MLS5.MOV     A Sterk, MCL, MLS5.MEL, MLS5.MEL	SUM * SMG SMG SMG SMG SMG SMG Soun					
customers (\\fs2\ftp)							

#### **Creating a Play/Event Schedule**

If licensed for the Schedule playout capability, you can create a play schedule to produce a playout stream. To create a schedule click on the Schedule tab and then click on the O Add Schedule icon. Note that this selection is available for each of the Ethernet ports.

	Schedule	Files	Admin	Reporting	About									
Sc	hedule Control P	anel												
F	hysical Connect	or eth0												$\mathbf{S}$
$\triangleleft$	Add Schedule	>									Tx Bitrate:	0.013	Mbps	۲
F	hysical Connect	or eth1												
6	Add Schedule										Tx Bitrate:	128.510	Mbps	
	🗈 🎲 IP (Sche	edule 1)	Curr	ent Source: SH	IREK.trp		0	0:00:37	239.192.0.170:10	001	٢	18.500	Mbps	

A Configure IP menu opens to a General configuration menu. You may enter an alias name or use the default name. Select a Gap Mode using the dropdown selections in this field. The Gap Mode defines what you want to happen in the output between scheduled stream playout events. If you choose "Default File," click the Browse field and select a TS file to playout during gap periods. Define the IP Ethernet port, Destination IP address and port. Click the Apply field at the bottom to create the schedule.

The created schedule is shown as a listing or row. There can be up to 16 different schedules or rows. To define the playout streams, dates, and times in a schedule, click on the con to select the Scheduling Configuration Panel.



Physical Connector eth1					۵
O Add Schedule		Tx Bitrate:	73.797	Mbps	
🖃 🌼 IP (Schedule 1)	Current Source: GAP(starz_hevc_1080030 👮 00:00:16 239.1	92.0.170:10001	4.500	Mbps	۲



Click the Add Event icon to add a timed event or stream to the playlist. Browse and select the TS file you wish to create as a playout event. The file loads as a row into the schedule and default times and dates are generated. Click in the Start Date, Start Time, or Time Slot Duration fields to change these default values. Listings are show according to date and times in the schedule. See Section 5 in this manual for more details.

To start a file or event playing out immediately in a schedule, click on the Start Date field and enter the current date. Click on the Start Time field and enter a start time that is only a few seconds from the current time. You may need to check and/or adjust the current time of the Server. The current date/time is shown and set under the Admin tab in the Date/Time section. See the Admin section of this manual for more details. Once the start time is reached, the event begins playout on the specified Ethernet port.

C	Configure IP (Schedule 7)											
	Add Event				🔯 Clear Old Events	📕 Clear Schedule	👔 Upload Ever	nts				
Γ	Start Date	Start Time	Time Slot Duration	File Name			File Duration					
L	. 09/27/2017	15:20:22	01:50:12	SILENT_BOB.trp			01:50:12	0				
L	. 09/27/2017	15:15:23	00:04:59	NASA_HEVC4K_5min_SPTSv2.trp			00:04:59	9				
	. 09/27/2017	17:10:34	00:01:55	fransat.ts			00:01:55	9				
L												
L												

### In Play Control Panel Add a Transport Stream

The TSS 6220 web GUI may include a Play tab, if licensed for the Play out feature. To create or add a playing stream to the IP output on the Ethernet 1 port, click on the **Physical Connector** eth1 heading. An Add Transport Stream menu opens and defaults to the General tab.

I	Play Schedule Files Admin Reporting About			
	Play Control Panel			
	Physical Connector eth0			۵
<	Add Transport Stream     Add PCAP	Tx Bitrate:	0.024 Mbps	۲
	Physical Connector eth1			٥
	O Add Transport Stream O Add PCAP	Tx Bitrate:	61.874 Mbps	۲
	🖼 🌼 🕕 🔹 IP (Stream 1) SILENT_BOB.trp 📑 01:08:28 239.192.0.212:10000	٢	18.500 Mbps	۲

### **Configure the Output Stream**

The General configuration menu provides selections defining the output IP stream and IP destination address and ports. Select a source file for playout. Click on the Browse field and navigate to the file you wish to select for playout. Open the file. It will populate the Source listing with your file name.

Define the IP parameters. The Physical Connector field should indicate eth1. If not, select the dropdown and select eth1. Click on the Destination IP field and enter the destination address. Enter a Destination Port in the Destination Port field. Click on the Apply box at the bottom of the menu to add the stream to the Ethernet port 1 (eth1) output.





### Start the Output Stream Playing

After the stream is added you will see the stream listed in the Physical Connector eth1 section of the Play Control Panel. If you don't see the listing click on the Show/Hide icon . Click on the Play icon I to begin producing the stream output. The light at the far right turns green and the bit rate field beside it begins to show the output transport stream bitrate. The stream time begins to increment and the blue highlight advances from the left (start) to the right (end) in the center progress window.

Physical Connector eth1			
Add Transport Stream O Add PCAP		Tx Bitr	ate: 30.167 Mbps
a 🔅 🕕 🔹 IP (Stream 1) Subtides.trp	00:00:11	239.192.0.200:1000	😂 5.100 Mbps 🔴

Your selected stream is now outputting to the network. Click on the cog wheel icon 
 to make changes to the stream output configuration. You may also click on the 
 icon to bring up a configuration and monitoring panel. It provides a shortcut to improve viewing and playout statistics while offering some convenient control options.



# 4 Play Control Panel

The play feature is a licensed feature of the TSS 6220. When licensed, the Play tab is shown and available to select. Selecting or clicking on the Play tab advances the GUI to the Play Control Panel. The Play Control panel provides management of TS stream files and PCAP files that are manually configured for playout from the TSS 6220. This panel manages each of the available Ethernet streaming ports defining which TS streams and PCAP files are sourced to each port. This panel further defines the criteria of each of the streams and PCAPs added to the output in the Play feature. A TS 6220 chassis may be optionally configured to support the addition of an added hardware card which provides multiple ASI input/output ports. The following sections in this chapter of the manual provide an overview of the features and menus associated with the Play feature of the TSS6220.

### 4.1 Play Control Panel Overview

The Play Control Panel includes a section for each available Ethernet output port. The sections are identified by headers indicating the physical connector port. For example, the Ethernet port 0 is shown as "Physical Connector eth0." If you have added the optional Ethernet ports to the TSS 6220, then two additional sections are included for Physical Connector eth2 and eth3. If you have the optional ASI card, additional sections are shown for these physical output ports. See section 4.7 for configuration of an ASI output transport stream.

Pla A s Admin B ting A	bout							
Play Con of Panel								$\frown$
Physical Connector eth0							C	<u> </u>
Add Transport Stream  Add PCAP					Tx Bitrate:	0.019 Mbps		
Physical Connector eth1							٥	
Add Transport Stream O Add PCAP					Tx Bitrate:	77.633 Mbps -	$\sim$	
⊞ 🌼 🕕 💿 IP (Stream 1)	Subtitles.trp	<b>1</b>	00:00:07	<b>E</b> :1000	0	5.100 Mbps	۲	
⊞ 🌼 🕕 💿 IP (Stream 3)	balloons_hevc_1080p30_422	<b>1</b>	00:00:38	239.192.0.202:10000	0	50.000 Mbps	۲	
🗉 🌼 🕕 💿 PCAP (Stream 1)	239_192_0_200-10000_100M	<b>1</b>	00:00:45		٢	20.785 Mbps		
Physical Connector eth2							٥	
Add Transport Stream Add PCAP					Tx Bitrate:	0.000 Mbps		
Physical Connector eth3							٥	
Add Transport Stream Add PCAP					Tx Bitrate:	0.000 Mbps		

Each Ethernet port section includes some common control fields. The following table provides an overview of the common fields in the Play Control Panel.

Item or	Field	Button/Action	Description
Α.	Add an IP Transport Stream Output	Add Click on this icon to add a TS file to stream to the output	Provides menu to select a file and define stream output characteristics adding the output to the Ethernet port. See section 4.3 for details
В.	Add PCAP stream output	Add Click on this icon to add a PCAP to the output	Provides a menu to select a PCAP file and criteria to define it adding it to the Ethernet output port 0. See section 4.4 for details

C.	Show/Hide port info	Selectable, click on the icon	Hides all the stream and PCAP listings, click to hide or click to show all streams/PCAPs
D.	TX Bitrate	Not a selectable field	Shows the total output bitrate of all the streams and PCAP to the Ethernet port
Ε.	TS Stream	See next section	Row defining a stream output to Ethernet 1 port (eth1)
F.	PCAP stream	See next section	Row defining a PCAP output to the Ethernet 1 port

### 4.2 Stream/PCAP Information Overview

The Play Control Panel shows all the playout IP streams and PCAPs. Each playback transport stream and PCAP file output has a row of information and control functions. Depending on licensing, up to 100 play listings are currently available. There are common data fields for each stream/PCAP listing forming columns of information in the panel. This section provides a brief definition of the information provided in each column.



Overview of the informational fields in the Play Control main panel.

Item or	Field Name	<b>Button/Action</b>	Description
Α.	Status & Configuration	Click on this icon	Provides a window showing IP stream/PCAP status and configuration information
В.	Configuration Menu Select	Click on this icon	Provides a menu with configuration settings to define the output IP/PCAP stream and IP address
C.	Output Control	<ul> <li>Pause -click on icon to Pause</li> <li>Play – click on icon to start output</li> </ul>	Indicates IP/PCAP stream as playing or paused. Click on icon to pause or play. When paused, the current location of the stream or PCAP is maintained.
D.	Stop Control	Click on icon to stop IP/PCAP output	Stops a playing or looping IP/PCAP output. Click on play icon to restart – restarts at file starting point.
E.	Stream name or alias	Not selectable, No action	Shows a default output IP/PCAP stream name. See section xx for naming streams.
F.	TS/PCAP File name	Double click to browse/select different file	Indicates the current selected play file.



G.	Stream configuration	Click on icon to open configuration menu	Provides convenient overview of stream playout status, some critical settings, and provides some control features. See section 4.6 for details.
H.	Play Status	Not selectable	Indicates a stream is playing or active. Indicates play position/time within the start-to-end duration time span. Visual blue highlight indicates stream progress.
I.	IP Address/Port	Not selectable	Indicates the destination IP address and port with MPEG-IP output. Indicates Null Stuffing status as Enabled or Disabled when using an ASI output
J.	Delete icon	Click <a>to delete</a> stream or PCAP	Removes a IP/PCAP stream from the play listings and Ethernet output
K.	Total Port Bitrate Indication	Not selectable, view only	Indicates accumulative Ethernet port bitrate of the addition of all playing TS streams and PCAP files
L.	Bitrate Indication	Not selectable, view only	Indicates bitrate of the individual stream to the Ethernet port
М.	Status Indicator		Indicates status of output:
			Gray: Inactive – stopped or paused
			Green: Good playback output condition
			Red: Fault condition

### 4.3 Player - Adding a Transport Stream - IP

To create a new IP stream output requires that you select a play file and define its output parameters. To create or add a playing stream, click on the Additional icon. Note that this selection is available for each of the Physical Connector Ethernet ports of your TSS 6220. Select the Additional icon in the section corresponding to the Ethernet port in which you want the output to stream.

Play Files Admin Reporting About		_		
Play Control Panel				
Physical Connector eth0				
Add Transport Stream Add PCAP	Tx Bitrate: 0.023 Mbps			
Physical Connector eth1 O				
Add Transport Stream	Tx Bitrate: 30.167 Mbps			

The Play Control Panel is simplified for viewing with a Hide/Show streams feature for each Physical Connector eth section. To show all the streams outputting to an Ethernet port click on the Show/Hide icon



### 4.3.1 Add Transport Stream - IP - General Configuration

Upon clicking on the Add Transport Stream icon, the Add Transport Stream menu opens. This menu contains 3 selection tabs which provide menus for defining the transport IP stream. By default, the General tab is selected providing some common selections to configure the output. The General configuration menu provides selections defining the output IP stream and IP destination address and ports. This section provides an overview of the configuration fields.

The first step is to select a source file for playout. Click on the Browse field and navigate to the file you wish to select for playout. Select the file and open it. It will populate the Source listing with your file name. In the Stream section of



the menu, click on the Alias field and enter an alias name, if desired. This is not required as the application automatically assigns a name as an incrementing stream number.

If you want the playing stream upon reaching the ending point to loop back to the start and continue playing out, click on the dropdown arrow in the Looping field. Select "Enabled" if you want the looping feature active. When looping is enabled the Seamless Looping field becomes available to configure. Set this field to "Enabled" if you want the TSS 6220 to properly seam the ending point with the beginning point of the stream. With this enabled the TSS 6220 corrects PCR discontinuities and continuity count values.

Define the IP parameters in the IP section. Verify that the Physical Connector field indicates the desired Ethernet port. If not, select the Ethernet port you wish to output the stream. Enter the Destination IP and the Destination Port values. Click on the Destination IP field and enter the destination address. Click on the Destination Port field and enter a Destination Port. Click on the Apply box to add the stream to the Playout Control Panel under the respective Ethernet port.

Setting	Range	Description
Stream - Alias	Value and or Letter Entry	Provides entry to name or assign alias to identify a stream. If no name is entered, the TSS 6220 assigns an incrementing stream number
Stream - Looping	Enabled Disabled	Turns on the looping function in which the file plays to the end and then loops back and starts playing from the start again. When disabled, the stream stops when it reaches the end of the file
Seamless Looping	Enabled Disabled	Enabled: Provides corrections to PCR values and PAT/PMT table continuity counts so as to appear to the receiver as a continuous uninterrupted stream.
Source	Not selectable	Indicates "No Source" when no file has been selected. Indicates the selected file name when a file is opened.
Browse	Click on field to access stream files in library	Provides navigation to browse to available stream and PCAP files for selection.

Add Transport Stream - Summary of General Tab Settings



IP – Physical Connector	Selects eth0 or eth1, the standard Ethernet ports,	Entry selects which of the available Ethernet ports on the TSS 6220 that the stream output is assigned
	eth2 and eth3 available if TSS 6220 has added optional ports	
Destination IP	224.0.0.0 – 239.255.255.255	This address is the IP address the source device is sending to for a multi-cast. A unicast would use an IP address less than 224.0.0.0.
Destination Port	0 – 65535	This is the UDP port the source device is sending to.

### 4.3.2 Add Transport Stream - Advanced IP Configuration

The Add Transport Stream Advanced IP tab includes settings to define the TSS 6220 as a specific source device for IGMPv3. This feature allows each steam to be seen by the network as a unique source device with a unique IP address, and/or source port, and/or MAC address. This section provides descriptions of the settings in the Advanced IP menu.

Add Transport Stream		
General Advanced IP	ile	
Source IP Mode:	Auto	Ŧ
Source IP:	0.0.0.0	
Source Port:	3020	÷
Source MAC Mode:	Auto	~
Source MAC:	00:00:00:00:00:00	
TS Packets Per IP Packet:	7	÷
	Apply C	ancel

Setting	Range	Description
Source IP Mode	Settings: Select Auto or Manual	In Auto, the TSS 6220 simulates a source device and creates and communicates a unique source IP address. In Manual, a user entered source IP address can be entered.
Source IP	Available for entry when Source IP Mode is set to Manual.	Provides entry of a Source IP address for the stream that is communicated by the TSS 6220
Source Port	Click up or down arrows to increment value shown. Click in field and enter value.	Specifies a value for the source port associated with the stream.
	Range: 1030 to 65535	



Source MAC Mode	Auto, Manual	In Auto, the TSS 6220 simulates a source device and creates and communicates a unique source MAC address for the stream. In Manual, a user Source MAC address can be entered.
Source MAC	In "Manual Mode" enter MAC address,	Provides entry of a MAC address you want to specify as the Source MAC for the stream
TS Packets Per IP Packet	Enter value 1 to 7, Default is 7	This setting determines the number of TS stream packets that are inserted into IP packets. 7 being the maximum and the typical setting. Lesser packets may be selected.

### 4.3.3 Add Transport Stream - IP - File Configuration

The File tab within the Add Transport Stream menu provides selection to define starting and ending points when playing out an IP stream file. It further provides a selection to automatically or manually control the output IP stream bitrate. To access this configuration menu, click on the File header tab

You can enter starting and ending times within the play duration time of the IP play file. You can enable or disable the playout to conform to these entered start point and end point times. Set the Start/End Point field to "Enabled" and enter the Start Point and End Point.

Add Transport Stream			
General	Advanced IP	File	
Start/End I	Points:	Disabled	~
Start Point		00:00:00	
End Point:		00:00:00	
Bitrate Mo	de:	Auto	v
Bitrate (Mbps):		0	A V
		Apply	Cancel

The Bitrate Mode provides an automatic setting in which the

TSS 6220 automatically determines the ideal playout stream rate. You can set the Bitrate mode to Manual when you wish to increase the output bitrate. In the manual mode, click on the up and/or down arrows in the bitrate field to increase or decrease the bitrate in Mbps.

Setting	Range	Description
Start/End Points	Enabled Disabled	Adds control of the output defining a starting and stopping point within the play duration of the selected stream or PCAP play file
Start Point	Range of play file duration but prior to End Point	This setting establishes a point/time in the play file duration which defines the starting point of the stream or PCAP file when it is streaming to the output port
End Point	Range of play file duration but after the start point	This setting establishes a point/time in the play file duration which defines the ending point of the stream or PCAP file when it is streaming to the output port
Bitrate Mode	Auto Manual	In Auto the TSS 6220 determines the playout rate automatically. In Manual, the user may modify or enter a playout bitrate.
Bitrate (Manual)	Enter Bitrate	Provides user entry to manually enter a desired output bitrate

The following chart provides an overview of the configuration settings in the File menu.



The Bitrate Mode provides an automatic setting in which the TSS 6220 automatically determines the ideal playout stream rate. You can set the Bitrate Mode to Manual to increase the output bitrate. In the manual mode, click on the up and/or down arrows in the bitrate field to enter a bitrate in Mbps.

Add Transport Stream				
General	Advanced IP	File		
Start/End I	Points:	Enabled	-	
Start Point		00:00:00		
End Point:		00:00:00		
Bitrate Mo	de:	Manual	~	
Bitrate (Mt	ops):	0	\$	
		Apply	Cancol	
		Apply	Cancel	

# 4.4 Adding a PCAP Play File

To create a new PCAP output file test stream requires that you select a PCAP file and define its output parameters. To create or add a PCAP stream, click on the Add PCAP icon. Note that this selection is available for each of the Physical Connector Ethernet ports of your TSS 6220. Select the Add PCAP icon in the section corresponding to the Ethernet port in which you want the output to stream. This section describes how to select, add and configure a PCAP file to playout one of the TSS 6220's Ethernet port outputs.

Play Files Admin Reporting About	
Play Control Panel	
Physical Connector eth0	۵
Add Transport Streen Add PCAP	Tx Bitrate: 0.023 Mbps
Physical Connector eth1	۵
Add Transport Streen Add PCAP	Tx Bitrate: 30.167 Mbps

The Play Control Panel is simplified for viewing with a Hide/Show streams feature for each Physical Connector eth section. To show all the streams outputting to an Ethernet port click on the Show/Hide icon

Play	Record	Schedule	Files	Admin	Reporting	About								
Play (	Control Pan	el												
Phys	sical Conne	ctor eth0										$\langle$		
() A	dd Transport	Stream 🛈	Add PCA	Þ							Tx Bitrate:	0.013	Mbps	
Phys	sical Conne	ctor eth1												$\mathbf{O}$
• A	dd Transport	Stream 🔘	Add PCA	p							Tx Bitrate:	69.258	Mbps	
٠	۰ ا	IP (Stream)	am 1)	mont	y_python.trp		<b>1</b>	00:00:00	]	239.192.0.100:10000	9	10.522	Mbps	۲



### 4.4.1 Add PCAP - General PCAP Play Settings

When adding a PCAP file to play on the TSS 6220 use the configuration settings in the General, Advanced PCAP, and File tabs to configure the output.

Click on the O Add PCAP icon to add a PCAP file to the play list. Upon clicking the icon, the Add Transport Stream menu opens. This menu contains 3 selection tabs which provide menus for defining the transport PCAP stream. By default, the General tab is selected providing some common selections to configure the output. The General configuration menu provides selections defining the output PCAP stream and its destination address and ports. This section provides an overview of the configuration fields.

The first step is to select a source PCAP file for playout. Click on the Browse field and navigate to the PCAP file you wish to select for playout. Select the file and open it. It will populate the Source listing with the file name. In the Stream section of the menu click in the Alias field and enter an alias name as desired. This is not required as the application automatically assigns a name as an incrementing stream number.

If you want the playing PCAP stream upon reaching the ending point to loop back to the start and

continue playing out, click on the dropdown arrow in the Looping field. Select "Enabled" if you want the looping feature active.

Define the PCAP parameters in the PCAP section. Verify that the Physical Connector field indicates the desired Ethernet port. If not, select the Ethernet port you wish to output the stream. Enter the Destination IP and the Destination Port values.

The PCAP file contains within it the destination and port addresses so these parameters cannot be entered in the same manner as with a TS stream. When finished with the configuration, click on the Apply box to add the stream to the Playout Control Panel under the respective Ethernet port.

Add PCAP					
General Advanced PCAP	File				
Stream					
Alias:	Stream 1				
Looping:	Disabled 👻				
Source:	NO SOURCE				
	Browse				
PCAP					
Physical Connector:	eth0 👻				
	Apply Cancel				

Add PCAP menu – Summary of General Tab Settings

Setting	Range	Description
Stream - Alias	Value and or Letter Entry	Enter name you want to identify stream. If no name is entered, the TSS 6220 assigns an incrementing stream number
Looping	Enabled Disabled	Enabled: Turns on the looping function in which the PCAP file plays to the end and then loops back and starts playing from the start again.
		Disabled: After the file is started it plays to the end of the file and stops.
Source	Not selectable	Indicates "No Source" when no file has been selected. Indicates the selected file name when a file is opened.
Browse	Click on <b>Brows</b> field to access PCAP files in	Provides navigation to browse to available PCAP files for selection.



IP - Physical<br/>ConnectorSelects eth0 or eth1,<br/>the standard Ethernet<br/>ports,Entry selects which of the available Ethernet ports<br/>on the TSS 6220 that the PCAP output is assigned<br/>available if TSS 6220<br/>has added optional<br/>ports

### 4.4.2 Add PCAP - Advanced PCAP Settings

The Advanced PCAP selection tab provides a menu to define the Source MAC Mode. The MAC address indicated by the PCAP play output may be configured to match the MAC address of the NIC or be IP Source criteria. The following provides descriptions of the settings included.

### Source MAC Mode: From NIC

When the MAC address source is set to "From NIC" the MAC address for all the Ethernet frames transmitted during PCAP play output shall be the same as the MAC address of the selected output NIC.

Add PCAP	) Fit	
General Advanced PCAP	File	_
Source MAC Mode:	From NIC	-
	From Source	
	From NIC	
1		
1		
1		
	Apply Cancel	

#### Source MAC Mode: From Source

When the MAC address source is set to "From Source" the MAC address for all Ethernet frames transmitted during a PCAP play output shall be the same as captured in the selected PCAP file.

Setting	Range	Description
Source MAC	From NIC	Determines the MAC address transmitted with all
Mode	From Source	Ethernet frames during PCAP output.
		From NIC: MAC is the same as MAC of selected NIC
		From Source: MAC address is the same as MAC captured in the selected PCAP file.

### 4.4.3 Add PCAP - File Play Settings

The File tab within the Add PCAP menu provides selection to define starting and ending points when playing out a PCAP file. You can enter starting and ending times within the play duration time of the PCAP file. You can enable or disable the playout to conform to these entered start point and end point times that you specified. To access this configuration menu, click on the File header tab. The following section provides an overview of this menu.



Keep in mind some general rules. You cannot set an End Point time that is after the normal End Play time of the

selected file. You cannot set an End Point that is prior to the Start Point time. You can set a Start


Point and End Point in the File Menu, but you must set the Start/End Points to "Enabled" to playout only the part of the PCAP specified by the Start/End Point times.

Add PCAP – File	Tab Settings	Summary
-----------------	--------------	---------

Setting	Range	Description
Start/End Points	Enabled Disabled	Adds control of the output defining a starting and stopping point within the play duration of the selected stream or PCAP play file. Enabled: Directs output to follow the defined start and ending time references entered. Disabled: Output PCAP does not follow start or ending time entries. Playout is from the beginning of the file to the end of the file
Start Point	Range of play file duration but prior to End Point	This setting establishes a point/time in the play file duration which defines the starting point of the stream or PCAP file when it is streaming to the output port
End Point	Range of play file duration but after the start point	This setting establishes a point/time in the play file duration which defines the ending point of the stream or PCAP file when it is streaming to the output port

To setup starting and ending playback points for a PCAP file. Note the duration of the file in the Play Control Panel prior to specifying the starting and ending points as the times selected must be within the duration offered by the file.

Select the cog wheel for the PCAP file to be specified. Click on the File tab to select the File Menu containing the starting and end points settings. Click the dropdown arrow in the Start/End Points field and select "Enabled." The Start Point and Ending Point fields become available for entry. Click on

the appropriate hour, minute, seconds field and modify to the desired start time. Repeat to enter the desired end time. Note the total duration when setting the Start Point and End Point values.

When starting and ending points are specified and enabled, the playback windows observed in the Playback Control Panel change to indicate the start/end limits that are specified. Brackets indicate the starting and ending points.



Enabled

00:00:05

00.00.18

## 4.5 IP Stream and PCAP Status & Configuration Information

For each of the IP streams or PCAP files that are listed in the Play Control Panel a Status and

Configuration window is available. To access this window and add it to the panel, click on the <sup>III</sup> icon at the left of the row containing the IP stream or PCAP listing. The Status and Configuration boxes are added to the Play Control Panel

Phy	sical (	Conn	ector	eth1	
• A	dd Tra	inspo	rt Stre	am 💿 Add PCAP	
		١		IP (Stream 1)	COLORADO.TRP

General

Start Point:

End Point

Start/End Points

Advanced PCAP

providing status and configuration details. Click on the icon at the same location to hide the status and configuration information.



#### **IP Stream - Status and Configuration Windows**

This section summarizes the status and configuration information provided for an IP playout stream.

Physical Connector eth1							۵
O Add Transport Stream O Add PCAP					Tx Bitrate:	30.169 Mbps	
🖃 🌼 🕕 💿 IP (Stream 1)	Subtitles.trp	3	00:00:10	239.192.0.200:1000	9	5.100 Mbps	۲
Status           File Size:         8.6 MB           Duration:         00:00:14           Source PP:         192.163.1.6           Source MAC:         0C:C4:7A:CD:2D:73           Mode:         Multicast           Receiver MAC:         N/A	Configuration Playback Loop: Enabled Source IP Mode: Auto Source Port: 3020 Source MAC Mode: Auto TS Packets: 7 Seamles Looping: Enabled Bitrate Mode: Auto						
🗄 🌼 🕕 💿 IP (Stream 2)	21Paris_1.trp	-	00:01:08	239.192.0.201:10000	0	24.128 Mbps	۲

Summary of the listings and descriptions of the Status section.

Status Listing	Description
File Size:	Indicates the total memory size of the play file
Duration:	Indicates the total playtime or duration of the selected file or PCAP
Source IP:	Indicates the source IP address
Source MAC:	Indicates the source MAC address
Mode:	Indicates the current streaming Mode, Multicast, Unicast,
Receiver MAC:	Indicates the MAC address of the receiver, when available or communicated by the receiver



Configuration Listing	Description
Playback Loop:	Indicates the Playback Loop setting as Enabled or Disabled. Indicates if the Playback Loop feature is active for stream playout
Source IP Mode:	Indicates the Source IP Mode setting as Auto or Manual. Auto indicates the IP automatically selected by the TSS 6220. Manual provides user entry of the source IP address.
Source Port:	Indicates the Source Port value setting
Source MAC Mode:	Indicates the Source MAC Mode setting as Auto or Manual. Auto indicates the MAC address is automatically selected and communicated by the TSS 6220. Manual provides user entry of the MAC address communicated as the Source
TS Packets:	Indicates the number of TS packets per IP packet setting
Seamless Looping:	Indicates the Seamless Looping Setting as Enabled or Disabled
Bitrate Mode:	Indicates the Bitrate Mode setting as Auto or Manual.

Summary of the listings and descriptions of the Configuration section.

#### **PCAP File - Status and Configuration Windows**

Click on the <sup>III</sup> icon at the left of the row containing a PCAP stream adds the Status and Configuration windows to the Play Control Panel. There is less status and configuration information for a PCAP file compared to an IP stream file. The following describes the information provided in the Status and Configuration windows. Click on the <sup>III</sup> icon at the same location to hide the Status and Configuration windows.



Duration: Indicates the total playtime or duration of the sele	ted PCAP file
--	---------------

Source MAC: Indicates the source MAC address



Configuration Listing	Description
Playback Loop:	Indicates the Playback Loop setting as Enabled or Disabled. Indicates if the Playback Loop feature is active for PCAP playout. When Enabled, the PCAP output upon playing to the end of the file loops back and restarts
Source MAC Mode:	Indicates the Source MAC Mode setting as From NIC or From Source. Setting determines the Source MAC transmitted with Ethernet frames. From Source: MAC address is the same as MAC captured in the selected PCAP file.

## 4.6 IP and PCAP Monitor Panel

For each listed transport stream or PCAP row in the Play Control panel you will find a <sup>1</sup> icon. Clicking on this icon provides a monitoring panel for that transport stream or PCAP output. This panel provides a convenient shortcut to improve viewing of the playout statistics and provides several convenient control options. The following descriptions provide an overview of the features provided.



Configure IP/PCAP Panel Descriptions

Item or	Field Name	Button/Action	Description
Α.	File name	Not a selectable field	Indicates the current source file for the transport stream or PCAP file that is playing.
В.	Browse	Click to browse or view available files	Provides quick access to view files, to locate current play file or search for files. Files cannot be selected in this menu to replace the existing play file.
C.	Bitrate	Not selectable	Indicates the bitrate of IP or PCAP stream
D.	Play location time indicator	Not selectable	Indicates the current play position or time within the duration or total play time of the IP or PCAP stream
E.	Stream End Time	Not selectable	Indicates the total time duration or the ending time of the IP or PCAP stream.



F.	Play Bar Progress Indicator	Not selectable	Provides visual indicator that the stream output is active and progressing. Indicates the current play position or time relative to the start and end points.
G.	Play/Pause Control	<ul> <li>Pause,Click on icon to pause output</li> <li>Play – click on icon to start output</li> </ul>	Indicates IP/PCAP stream as playing or paused. Click on icon to pause or play. When paused, the current location of the stream or PCAP is maintained.
H.	Stop Control	Click on icon to stop IP/PCAP output	Stops a playing or looping IP/PCAP output. Click on play icon to restart – restarts at file starting point.
I.	Loop Control	<sup> </sup>	<ul> <li>Indicates the loop mode is enabled. The stream plays to the end and loops to the starting point and continues to play out.</li> <li>Indicates the loop mode is disabled. The stream plays to the end and stops.</li> </ul>
J.	Start Point Time	Value cannot be changed in this menu	Indicates a specified stream starting point time.
K.	Start/End Time Play Enable	Click box – adds check in box (Enabled), click box removes check mark (Disabled)	Provides Enable/Disable control of start/end points of the playout stream. Enabled (Box checked) stream plays from entered start to end points. Disabled (box not checked) the stream plays from beginning to end of the file
L.	End Point Time	Value cannot be changed in this menu	Indicates a specified stream End Point time.

## 4.7 Play Control Panel - ASI Output

A TS 6220 chassis may be optionally configured to support the addition of an added hardware card which provides multiple ASI output ports. The following sections in this chapter provide an overview of the features and menus associated with the Play feature of the TSS 6220 when configuring and using an ASI output port.

When the hardware supports ASI output playout, the Play Panel contains sections which indicate available ASI ports. When configured for ASI output, the sections include the configuration information and status of the playout streaming.

Play Record Schedule Delay	Disaster Recovery	Files Admin	Reporting Abo	ut			
Play Control Panel							
Physical Connector eth0							۵
Add Transport Stream     Add PCAP					Tx Bitrate:	0.011 Mbps	۲
Physical Connector eth1							۵
Add Transport Stream     Add PCAP					Tx Bitrate:	- Mbps	۲
ASI Port 1							۵
Cooble A Si Play							
ASI Port 2							۵
Enable ASI Play							
🗉 🌼 🕕 💿 ASI (Stream 2)	BALLOONS.TRP		00:00:19	Null Stuffing: Disabled	9	19.393 Mbps	۲



#### 4.7.1 Add Transport Stream - ASI - General Configuration

To create and add a playout of a transport stream to an ASI output port, click on the To create and add a playout of a transport stream to an ASI output port, click on the To create and add a playout of a transport stream to an ASI output port, click on the To create and Play icon in one of the ASI Port sections of the main Play Panel. If the Enable ASI Play icon is grayed out, it indicates that the listed ASI port is in use and not available to select and configure for an output. The ASI output ports can only be used by one application. The ASI port can be used by the Play, Record, Schedule, Delay and Disaster Recovery features.

When you click on the **C** Enable ASI Play icon, in which the port is available (not grayed out), the Add ASI menu opens. This menu contains 2 selection tabs which provide menus for defining the ASI output stream. By default, the General tab is selected providing some common selections to configure the output. The General configuration menu provides selections to create an alias name, select from available source streams in the library to playout, and control the looping of the output stream. This section provides an overview of the configuration fields.

Stream	
Alias:	Stream 1
Looping:	Enabled
Seamless Looping:	Enabled -
Source:	BALLOONS.TRP
	Browse

The first step is to select a source file for playout. Click on

the Browse field and navigate to the file you wish to select for playout. Select the file and open it. It will populate the Source listing with your file name. In the Stream section of the menu, click on the Alias field and enter an alias name, if desired. This is not required as the application automatically assigns a name as an incrementing stream number.

If you want the playing stream upon reaching the ending point to loop back to the start and continue playing out, click on the dropdown arrow in the Looping field. Select "Enabled" if you want the looping feature active. When looping is enabled the Seamless Looping field becomes available to configure. Set this field to "Enabled" if you want the TSS 6220 to properly seam the ending point with the beginning point of the stream. With this enabled the TSS 6220 corrects PCR discontinuities and continuity count values.

Setting	Range	Description
Stream - Alias	Value and or Letter Entry	Provides entry to name or assign alias to identify a stream. If no name is entered, the TSS 6220 assigns an incrementing stream number.
Stream - Looping	Enabled Disabled	Turns on the looping function in which the file plays to the end and then loops back and starts playing from the start again. When disabled, the stream stops when it reaches the end of the file.
Seamless Looping	Enabled Disabled	Enabled: Provides corrections to PCR values and PAT/PMT table continuity counts so as to appear to the receiver as a continuous uninterrupted stream.
Source	Not selectable	Indicates "No Source" when no file has been selected. Indicates the selected file name when a file is selected and opened for ASI playout.
Browse	Click on errors field to access stream files in library	Provides navigation to browse to available stream and PCAP files for selection.

Add ASI Menu – Summary of General Tab Settings



#### 4.7.2 Add Transport Stream - File - Configuration

The File tab within the Add ASI menu provides selection to define starting and ending points when playing out an ASI stream file. It further provides a selection to automatically or manually control the output stream bitrate. Also, there is a capability to add null stuffing to the output transport stream to increase the total output bit rate. To access this configuration menu, click on the File header tab.

You can enter starting and ending times within the play duration time of the ASI play file. You can enable or disable the playout to conform to these entered start point and end point times. Set the Start/End Point field to "Enabled" and enter the Start Point and End Point.

Add ASI General File		
Start/End Points:	Enabled	Ŧ
Start Point:	00:00:10	
End Point:	00:02:00	
Bitrate Mode:	Auto	v
Bitrate (Mbps):	0	i è
Null Stuffing:	Enabled	-
Total Bitrate (Mbps):	0	\$
	Apply	Cancel

The Bitrate Mode provides an automatic setting in which the TSS 6220 automatically determines the ideal playout stream rate. You can set the Bitrate mode to Manual when you wish to increase the output bitrate. In the manual mode, click on the up and/or down arrows in the bitrate field to increase or decrease the bitrate in Mbps. Keep in mind that increasing or decreasing the bit rate from its automatic or normal PCR determined rate by the TSS 6220, may cause undesired changes with TS receive devices as PCR timing is no longer maintained.

The Null Stuffing feature of the Add ASI menu, provides the ability to add Null stuffing to the output stream. Click on the dropdown arrow field and enable the feature. Enter the desired total bitrate (Mbps) or use the up and/or down arrows in the Total Bitrate (Mbps) field to select the desired total bitrate.

Setting	Range	Description
Start/End Points	Enabled	Adds control of the output defining a starting and stopping point within the play duration of the
	Disabled	selected stream or PCAP play file
Start Point	Range of play file duration but prior to End Point	This setting establishes a point/time in the play file duration which defines the starting point of the stream or PCAP file when it is streaming to the output port
End Point	Range of play file duration but after the start point	This setting establishes a point/time in the play file duration which defines the ending point of the stream or PCAP file when it is streaming to the output port
Bitrate Mode	Auto	In Auto the TSS 6220 determines the playout rate automatically. In Manual, the user may modify or
	Manual	enter a playout bitrate.
Bitrate (Manual)	Enter Bitrate	Provides user entry to manually enter a desired output bitrate

The following chart provides an overview of the configuration settings in the File menu.



# **5** Schedule Panel

The Schedule Panel provides the capability to control output streams over an IP network according to a timed schedule. You must create a schedule defining the IP output parameters, the transport streams to playout, the start and end times for each streaming event within the schedule, and what you wish to output during gaps or times between scheduled events. Up to 16 schedules can be created/listed for playout. Optional hardware is available to provide a scheduled output via an ASI output port(s). This section describes how to create, save, play, and manage schedules within the TSS 6220.

Play	S	chedule	Files	Admin	Reportin	g About								
Sche	dule C	Control Pa	anel											
Phy	sical	Connecto	or eth0											۵
٥	Add Sc	hedule									Tx Bitrate:	0.033 1	Mbps	۲
Phy	sical	Connecto	or eth1											0
	Add Sc	hedule									Tx Bitrate:	101.100	Mbps	
	ŵ	IP (Sche	dule 1)	Curre	ent Source:	GAP(balloons	_hevc_1080p6	. 对	00:00:00	239.192.0.188:10000	9	34.000	Mbps	۲
Ð	ŵ	IP (Sche	dule 2)	Curre	ent Source:	GAP(Null Pack	(ets)	7	00:00:00	239.192.0.145:10000	9	10.000	Mbps	۲
æ	ŵ	IP (Sche	dule 3)	Curre	ent Source:	GAP(starz_her	vc_1080p30_4	. 큧	00:00:28	239.192.0.169:10000	٢	4.500 1	Mbps	۲
æ	÷	IP (Sche	dule 4)	Curre	ent Source:	GAP(No Outpu	ıt)	7	00:00:00	239.150.0.29:10000	9	0.000	Mbps	۲
æ	÷	IP (Sche	dule 5)	Curre	ent Source:	GAP(starz_he	vc_1080p30_4	. 😰	00:00:00	239.192.0.45:10000	9	4.500 1	Mbps	۲
⊞	÷\$	IP (Sche	dule 6)	Curre	ent Source:	GAP(No Outpu	ıt)	7	00:00:00	239.192.0.12:10000	٢	0.000	Mbps	۲
Ð	÷\$}+	IP (Sche	dule 7)	Curre	ent Source:	GAP(No Outpu	ıt)	7	00:00:00	239.192.0.200:10000	٢	0.000 1	Mbps	۲
æ	÷ĝŀ	IP (Sche	dule 8)	Curre	ent Source:	GAP(No Outpu	ıt)	7	00:00:00	239.192.0.200:10000	٢	0.000 1	Mbps	۲
Ð	÷	IP (Sche	dule 9)	Curre	ent Source:	GAP(No Outpu	ıt)	7	00:00:00	239.192.0.109:10000	٢	0.000 1	Mbps	۲
Ð	÷\$	IP (Sche	dule 10)	Curre	ent Source:	GAP(No Outpu	ıt)	7	00:00:00	239.192.0.16:10000	0	0.000 1	Mbps	۲
Ð	÷	IP (Sche	dule 11)	Curre	ent Source:	GAP(starz_he	vc_1080p30_4	. 😰	00:00:00	239.192.0.19:10000	٢	0.000 1	Mbps	۲
Ð	÷\$	IP (Sche	dule 12)	Curre	ent Source:	GAP(No Outpu	ıt)	7	00:00:00	239.192.0.200:10000	٢	0.000 1	Mbps	۲
	÷.	IP (Sche	dule 13)	Curre	ent Source:	GAP(1MB_TS	_Rate_h.264	7	00:00:00	239.192.0.21:10000	9	1.000	Mbps	۲
۲	÷\$	IP (Sche	dule 14)	Curre	ent Source:	GAP(No Outpu	ıt)	7	00:00:00	239.192.0.22:10000	٢	0.000	Mbps	۲
Ð	÷	IP (Sche	dule 15)	Curre	ent Source:	GAP(Null Pack	(ets)	7	00:00:00	239.192.0.200:10000	٢	10.000 I	Mbps	۲

Schedule Panel showing ASI ports when optional ASI input/output hardware is included in the TSS6220 configuration.

										 10.00.17	olak obuge.	11.0 0	
Play	Record	Schedule	Delay	Disaster Recovery	Files	Admin	Reporting	About					
Scheo	lule Contre	ol Panel											
Inte	rface eth0												$\mathbf{O}$
• •	dd Schedul	e								Tx Bitrate:	0.009 Mt	ps	
Inte	rface eth1												$\mathbf{O}$
• •	dd Schedul	8								Tx Bitrate:	- Mt	ps	
ASI	Port 1												$\mathbf{O}$
() E	nable ASI S	hedule											
ASI	Port 2												$\mathbf{O}$
() E	nable ASI S	hedule											



## 5.1 Schedule Panel Overview

The Schedule Panel includes a section for each available ethernet output port. The sections are identified by headers indicating the physical connector port. For example, the ethernet port 0 is shown as "Physical Connector eth0." If you have added the optional Ethernet ports to the TSS 6220, then two additional sections are included for Physical Connector eth2 and eth3. If optional ASI hardware is included with the TSS 6220 the panel includes sections for each ASI port.

Sc.         dule Control Panel           Phylical Connector eth0         Image: Control Panel           Image: Connector eth1         Image: Connector eth1           Image: Image: Image: Connector eth1         Image: Connector eth1           Image: Image: Image: Connector eth1         Image: Connector eth1           Image: Image: Image: Image: Connector eth1         Image: Connector eth1           Image: Image: Image: Image: Connector eth1         Image: Connector eth1           Image: Image: Image: Image: Image: Connector eth1         Image: Image	A	So	chedule Files	Admin	Reporting	About							_	
Phy         cal Connector eth0         Image: Connector eth1         Image: Connector	Sc. c	lule (	Control Panel											$\overline{\mathbf{C}}$
O Add Schedule       Tx Bitrate:       0.019 Mbps       D         Physical Connector eth1       B       Image: Connector eth1       Image: Connech1	Phy.	ical	Connector eth0											9
B         Tx Bitrate:         88:271         Mbps         Tx Bitrate:         S8:271         M	() A	dd Sc	hedule							Tx Bitrate:	0.019 Mbps -	-	שי	
O Add Schedule         Tx Bitrate:         88.271         Mbps         Image: Contract Source:         Sile NT_BOB trp         O0:01:118         239.192.0170.10001         B 8.000         Mbps         Image: Contract Source:         Image: Contract Source:         Contract Source: </th <th>Phys</th> <th>sical</th> <th>Connector eth1</th> <th>B</th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th>٢</th> <th></th> <th></th>	Phys	sical	Connector eth1	B								٢		
	• A	dd Sc	chedule	в						Tx Bitrate:	88.271 Mbps		ſ	
IP (Schedule 3)             Current Source: GAP(NASA_HEVC4K_5min               00:03:50               239.192.0.201:10000               20.000 Mbps                 IP (Schedule 4)          Current Source: GAP(SILENT_BOB.trp)              00:03:50               239.192.0.205:10000               218.500 Mbps		÷	IP (Schedule 1)	Current	Source: SIL	ENT_BOB.trp	2	00:51:18	239.192.0.170:10001	٢	18.500 Mbps		<	Е
IP (Schedule 4)         Current Source:         GAP(SILENT_BOB.trp)         Image: Object of the state of the s	÷	\$	IP (Schedule 3)	Current	Source: GAF	P(NASA_HEVC4K_5min.		00:03:50	239.192.0.201:10000	0	20.000 Mbps	۲		
	٠	-ij)+	IP (Schedule 4)	Current	Source: GAF	P(SILENT_BOB.trp)	7	00:08:49	239.192.0.205:10000	٢	18.500 Mbps	۲		
⊕         iP (Schedule 5)         Current Source: GAP(No Output)         ₪         00:00:00         239.192.0.200:10000         ●         0.000         Mbps         ●	٠	÷	IP (Schedule 5)	Current	Source: GAF	P(No Output)	7	00:00:00	239.192.0.200:10000	9	0.000 Mbps	۲		
🗃 🎲 IP (Schedule 7) Current Source: GAP(Null Packets) 🔯 00:00:00 239.192.0.166:10001 🤤 10.000 Mbps 🔵	٠	\$	IP (Schedule 7)	Current	Source: GAF	P(Null Packets)	7	00:00:00	239.192.0.166:10001	٢	10.000 Mbps	۲		

Each Ethernet port section includes some common control fields. The following is a general overview of items in the Schedule Control Panel and description of some common fields. The remainder of this chapter describes definition of the fields and configuration menus.

Item or	r Field	<b>Button/Action</b>	Description
Α.	Add Schedule to eth0 port	Add Click on this icon to add a schedule to eth 0	Provides menus to select files and define a schedule to stream to the output
В.	Add Schedule to eth1 port	Add Click this icon to add a schedule to stream to eth 1	Provides a menu to select files and define a schedule to stream to the output
C.	Show/Hide port info	Selectable, click on the icon	Hides or shows all the schedule listings, click to hide or click to show all schedules
D.	TX Bitrate	Not a selectable field	Shows the total output bitrate of all the streams and PCAP to the Ethernet port
E.	Schedule	See next section	Row showing a created schedule and information regarding the schedule's current status and output to Ethernet 1 port (eth1)

ASI Port 1	0
C Enable ASI Schedule	
ASI Port 2	۵
C Enable ASI Schedule	



## 5.2 Schedule - Information Fields

The Schedule Control Panel lists all the playout schedules. There can be up to a total of 16 schedules. Each schedule has a row of information and related control functions. There are common data fields for each listed schedule forming columns of information in the panel. This section provides a brief definition of the information provided in each column.



Item or	Field Name	<b>Button/Action</b>	Description
Α.	Status & Configuration	Click on this icon	Provides a window showing IP stream/PCAP status and configuration information. See section 5.4 for details.
В.	Configuration Menu Select	Click on this icon	Provides a menu with configuration settings to define the output IP/PCAP stream and IP address
C.	Stream name or alias	Not selectable, No action	Shows a default output IP/PCAP stream name. See section xx for naming streams.
D.	Physical Connector/port	Not Selectable	Indicates the Physical Ethernet port which the streams in the listed schedule will playout
E.	Current Source	Not Selectable	Indicates the current file actively streaming out within the schedule. Or, indicate a gap or time between scheduled playout streams and how the gap is filled.
F.	Schedule configuration	Click on icon to open the configuration menu	Provides overview of schedule playout events and status. Provides indications of time conflicts, Provides scheduling additions, edits and control features. See section 5.5 for details.
G.	Play Status	Not Selectable	Indicates a stream is playing or active. Indicates play position/time within the start-to-end duration time span. Blue highlight indicates stream progress.
Н.	IP Address/Port	Not Selectable	Indicates the destination IP address and port
I.	Delete icon	Click to delete the schedule	Removes a schedule from the listed schedules
J.	Total Port Bitrate Indication	Not selectable, view only	Indicates accumulative Ethernet port bitrate of the addition of all playing TS streams and PCAP files
К.	Bitrate Indication	Not selectable, view only	Indicates bitrate of the individual stream to the Ethernet port
L.	Status Indicator	Not Selectable	Indicates status of output: Gray: Inactive – output stopped, Green: Good active playout, Red: Playout fault or time conflict



## 5.3 Adding or Creating a Schedule Configuration - General

To create a new IP schedule requires that you select files and configure the output parameters. To create a schedule click on the O Add Schedule icon. Note that this selection is available for each of the Physical connector Ethernet ports of your TSS 6220. Select the O Add Schedule icon in the section corresponding to the Ethernet port in which you want to output the created schedule.



The Schedule Panel is simplified for viewing with a Hide/Show streams feature for each Physical Connector eth section. To show all the streams outputting to an Ethernet port click on the Show/Hide icon

#### 5.3.1 Add Schedule - General Configuration

Upon clicking on the O Add Schedule icon, the Configure IP menu opens to a General configuration menu. By default, the General tab is selected providing some common selections to configure the schedule. The General configuration menu provides selections defining the output IP stream and IP destination address and ports. This section provides an overview of the configuration fields.

The Schedule section provides alias naming and definition of the Gap Mode. Click on the Alias field and enter an alias name, if desired. This is not required as the application automatically assigns a name as an incrementing schedule number.

The Gap Mode defines what the output is when there is a time delay between when

Configure IP (Schedule 3)							
General Advanced IP							
Schedule							
Alias:	Schedule 3						
Gap Mode:	Default File 👻						
Source:	starz_hevc_1080p30_4						
	Browse						
- IP							
Physical Connector:	eth1 -						
Destination IP:	239.192.0.169						
Destination Port:	10000 \$						
	Apply Cancel						

a file ends and another is scheduled to begin. The Gap Mode provides selections to provide no output, null packet output, and output of a default TS file specified. When the gap mode is set to the "Default File" mode, the Source field defines the TS transport stream that plays out during a gap. Click on the Browse tab and choose the stream to be the default file.

Define the IP parameters in the IP section. Verify that the Physical Connector field indicates the desired Ethernet port. If not, select the Ethernet port you wish to output the stream. Enter the Destination IP and the Destination Port values. Click on the Destination IP field and enter the destination address. Click on the Destination Port field and enter a Destination Port. Click on the Apply box to add the schedule to the Schedule Control Panel under the respective Ethernet port.



Setting	Range/Selections	Description
Stream - Alias	Value and or Letter Entry	Provides entry to name or assign alias to identify a schedule. If no name is entered, the TSS 6220 assigns an incrementing schedule number. There is a limit of 16 schedules
GAP Mode	Click Dropdown –	Determines the output when a gap of time exists in the scheduled playout times between streams
	select from listed	Default File: Outputs choose TS file
		No Output: Outputs no output stream
		Null Packets: Outputs TS stream with null packets
Source	Not selectable	Indicates "No Source" when no file has been selected. Indicates the selected file name when a file is selected - opened.
Browse	Click on <b>Browse</b> field to access stream files in library	Provides navigation to browse to available stream and PCAP files for selection.
IP – Physical Connector	Selects eth0 or eth1, the standard Ethernet ports,	Entry selects which of the available Ethernet ports on the TSS 6220 that the stream output is assigned
	eth2 and eth3 available if TSS 6220 has added optional ports	
Destination IP	224.0.0.0 – 239.255.255.255	This address is the IP address the source device is sending to for a multi-cast. A unicast would use an IP address less than 224.0.0.0.
Destination Port	0 – 65535	This is the UDP port the source device is sending to.

#### Add Schedule – General Tab Settings



### 5.3.2 Add Schedule - Advanced IP Configuration

When adding a schedule requiring the configuration of the IP port as a source specific device for IGMPv3 use the Advanced IP Configuration menu. Click on the Advanced IP tab.

The Advanced IP tab includes settings to define the TSS 6220 as a specific source device for IGMPv3. This feature allows each steam to be seen by the network as a unique source device with a unique IP address, and/or source port, and/or MAC address. The chart below provides descriptions of the settings provided in the Advanced IP menu.

Configure IP (Schedule 3)							
General Advanced IP							
Source IP Mode:	Auto -						
Source IP:	0.0.0.0						
Source Port:	3020 \$						
Source MAC Mode:	Auto 👻						
Source MAC:	00:00:00:00:00						
TS Packets Per IP Packet:	7 \$						
Encapsulation:	UDP 👻						
	Apply Cancel						

Setting	Range	Description
Source IP Mode	Settings: Select Auto or Manual	In Auto, the TSS 6220 simulates a source device and creates and communicates a unique source IP address. In Manual, a user entered source IP address can be entered.
Source IP	Available for entry when Source IP Mode is set to Manual.	Provides entry of a Source IP address for the stream that is communicated by the TSS 6220
Source Port	Click up or down arrows to increment value shown. Click in field and enter value. Range: 1030 to	Specifies a value for the source port associated with the stream.
	65535	
Source MAC Mode	Auto, Manual	In Auto, the TSS 6220 simulates a source device and creates and communicates a unique source MAC address for the stream. In Manual, a user Source MAC address can be entered.
Source MAC	In "Manual Mode" enter MAC address,	Provides entry of a MAC address you want to specify as the Source MAC for the stream
TS Packets Per IP Packet	Enter value 1 to 7, Default is 7	This setting determines the number of TS stream packets that are inserted into IP packets. 7 being the maximum and the typical setting. Lesser packets may be selected.



### 5.3.3 Add Schedule – ASI - General Configuration

Upon clicking on the C Enable ASI Schedule icon, the Add ASI menu opens to a General configuration menu. The General configuration menu provides selections defining the output ASI stream. This section provides an overview of the configuration fields

The Schedule section provides alias naming and definition of the Gap Mode. Click on the Alias field and enter an alias name, if desired. This is not required as the application automatically assigns an incrementing schedule name.

The Gap Mode defines what the output is when there is a time delay between when a file ends and another is scheduled to begin. The Gap Mode provides selections to provide no output, null packet output, and output of a default TS file specified. When the gap mode is set to the "Default File"

Add ASI			
General	)		
- Schedule -			
Alias:		Schedule 2	
Gap Mode:		No Output	~
Source:			
			Browse
ASI —			
Null Stuffing		Disabled	-
Total Bitrate	(Mbps):	0	-
		Apply	Cancel

mode, the Source field defines the TS transport stream that plays out during a gap. Click on the Browse tab and choose the stream to be the default file.

The bottom ASI section provides a Null Stuffing feature. Select the dropdown arrow and select Enabled to make the Null Stuffing feature active. When active the Total Bitrate (Mbps) row becomes available for changing the output bitrate. Use the up and/or down arrows to select a desired bitrate or enter the desired bitrate. Modifying the normal bit rate of the stream deletes available null packets or adds null packets. Keep in mind changing the native bit rate alters normal PCR timing of the stream. When selections are complete, clicking on the Apply box adds the schedule event to the Schedule Control Panel under the respective ASI port.

Setting	Range/Selections	Description
Stream - Alias	Value and or Letter Entry	Provides entry to name or assign alias to identify a schedule. If no name is entered, an incrementing schedule number is assigned.
GAP Mode	Click Dropdown – select from listed	Determines the output when a gap of time exists in the scheduled playout times between streams Default File: Outputs choose TS fileNo Output: Outputs no output streamNull Packets: Outputs TS stream with null packets
Source	Not selectable	Indicates "No Source" when no file has been selected. Indicates the selected file name when a file is selected - opened.
Browse	Click on <b>Brows</b> field to access stream files in library	Provides navigation to browse to available stream and PCAP files for selection.
ASI Null Stuffing	Select Enabled or Disabled with drop down field	Disabled (Default): Leaves output bitrate unchanged. Enabled: Increase or Decreased output bitrate by deleting or adding null packets
Total Bitrate (Mbps)	Up/Down arrows increment field value. Direct Enter rate value	Modifies the native bit rate of the scheduled stream on the output by adding or deleting null packets in the TS stream.

#### Add Schedule – ASI - General Tab Settings



## 5.4 Schedule Configuration & Information Window

For each of the listed schedules in the Schedule Panel, a configuration & status window is available. Click on the <sup>B</sup> icon at the left of the row containing the schedule. The Configuration box and the schedule's start/date and file information is listed. Click on the <sup>B</sup> icon at the same location to hide the status and configuration information to simplify the Schedule panel.

Physical Connector eth1						
O Add Schedule						
IP (Schedule 1)	Current Source: SILENT_BOB.trp	7	00: <mark>51:1</mark> 8			

This section summarizes the data shown in the Configuration & Information details window.

Physical Connector eth1								٥
O Add Schedule						Tx Bitrate:	73.797 Mbps	
IP (Schedule 1)     Current Sol	urce: GAP(starz_	hevc_1080p30_	🔊 🛛 0	0:00:16	239.192.0.170:10001	9	4.500 Mbps	۲
Configuration	Start Date 🕇	Start Time 🕇	Time Slot Duration	File Name				
Gap Mode: Default File Disfault File: starz hevc 1080p30 420 08bit	09/22/2017	13:47:00	00:01:00	balloons_hevc_108	0p60_420_08bit_30Mbps_L4.1_HT_	_3.trp		
	09/22/2017	13:48:00	01:50:12	SILENT_BOB.trp				
	09/22/2017	15:38:12	01:50:12	SILENT_BOB.trp				
	$\mathbb{N}$		N	N				
	<u> </u>	_\ز						
	C	D	E	F				
	$\square$	$\subseteq$		Ľ				

Item or	Field Name	<b>Button/Action</b>	Description
Α.	Gap Mode	Not Selectable	Indicates the Gap Mode defined for the stream in the schedule. No output, Null Packets, or Default file
В.	Default File	Not Selectable	Indicates the Default File to be used in the gap if the gap mode is set to Default File.
C.	Start Date	Not selectable	Indicates the Start Date as to when the file within the schedule begins to playout.
D.	Start Time	Not Selectable	Indicates the Start Time during the Start Date as to when the file within the schedule begins to playout.
E.	Time Slot Duration	Not Selectable	Indicates the time slot or duration of time that the playout of the file requires once it begins playout.
F.	File Name	Not Selectable	The name of the file that is to be played in the scheduled time.



## 5.5 Scheduling Configuration Panel

The Scheduling Configuration Panel provide configuration of the play out scheduling or events. Click on the configuration Panel. Each of the rows or schedules, up to 16 different listings, has its own unique schedule for configuration. Select the icon in the row corresponding to the schedule you wish to edit.

Physical Connector eth1						٥
O Add Schedule			Tx Bitrate:	73.797	Mbps	
🖃 🌼 IP (Schedule 1)	Current Source: GAP(starz_hevc_1080630 🐉 00:00:16	239.192.0.170:10001	٢	4.500	Mbps	۲

This panel provides configuration of the play schedule. In this panel you create timed events or scheduled playout of stream files. You can review the current schedule and make additions, deletions, or changes. You can completely clear the current schedule and build a new one or load a schedule from a CSV file. This section provides a description of the fields contained in the Scheduling Configuration Panel.



Item or	Field Name	Button/Action	Description
Α.	Add Event	Click ③ Add Event icon to access Schedule Configuration Panel	Adds a timed event or stream (row) to the play list. When selected a Browse Files menu appears for file selection. File loads as a row into the schedule showing the default start date/time and file information.
В.	Clear Old Events	Click on Clear Old Events icon to select	Clears all the events or playlist files that are prior to the system's current date/time
C.	Clear Schedule	Click on Clear Schedule icon to select	Clears all events or playlist files from the schedule.
D.	Upload Events	Click on Upload Events icon to select	Provides selection of a CSV data file configured with matching date, time, file names, file duration data to directly populate Scheduling Configuration Panel field. See section 5.5.3.
E.	Start Date	Click in field to select.Enter date or click on calender to select date.	Selects or enters a date in which the event/file is scheduled to play out
F.	Start Time	Click in field to	Selects or enters a time within the scheduled day in which the event/file is scheduled to start playing out.



	select. Enter time.	scheduled time.
G. Time S Duratio	otClick in field tonselect. Enter valuesto shorten orlengthen theduration	Defaults to the playout time of the selected file. Permits increasing or decreasing the time. If decreased, the playout ends prior to the end of the stream. If lengthened, the event or playout continues at the end of the stream
H. File Na	me Not selectable	Indicates the name of the selected file within the play list of the schedule
I. File Lei	ngth Not Selectable	Indicates the play length or time duration of the play file listed
J. Delete	Click 🤤 to delete the schedule	Deletes the schedule listing or row from the schedule

#### 5.5.1 Scheduling Configuration Panel - Colors

The Schedule Panel provides background colors to supplement conditions or activities regarding the schedule. An active stream which is playing out is highlighted in green. A pink highlighted row indicates that this event is in conflict with another playout event. It is common for two rows to be highlighted pink when a schedule conflict exists.

Co	onfigure IP (Schedule 10)									
C	O Add Event			🔯 Clear Old Events 🛛 📑 Clear Schedule	👔 Upload Eve	nts				
	Start Date	Start Time	Time Slot Duration	File Name	File Duration					
	09/29/2017	16:26:00	00:01:00	balloons_hevc_1080p60_420_08bit_30Mbps_L4.1_HT_3.trp	00:01:00	0				
	09/29/2017	16:27:00	01:50:12	SILENT_BOB.trp	01:50:12	0				
	09/29/2017	18:10:12	01:50:12	SILENT_BOB.trp	01:50:12	0				
	09/29/2017	20:00:24	00:07:00	GAP(Null Packets)		0				
	09/29/2017	20:07:24	01:32:06	monty_python.trp	01:32:06	0				
4	09/29/2017	21:39:30	01:50:12	SILENT_BOB.trp	01:50:12	0				
	09/29/2017	23:29:42	00:04:59	NASA_HEVC4K_5min_SPTSv2.trp	00:04:59	0				

Background color	Button/Action
Dark Grey	Indicates a gap in the schedule and the gap playout stream specified
Red	Indicates a fault condition. A conflict in the times of the schedule between events. A missing or corrupt file. Scroll your mouse over red background listing for information on the conflict
Green	Indicates an active event in the schedule, this event or stream is currently playing out
Blue	Row is selected or highlighted
No color (While)	Indicates a normal listing or event in the schedule
Light Blue	Currently selected with Move Cursor

The colors in the Schedule Panel are summarized in the following chart.



#### 5.5.2 Scheduling Configuration Panel – Moving Cursor

The Configuration panel provides a Moving Cursor in which to select and move a listed event within the schedule. Moving your mouse over a listed event or row highlights the event producing a cursor at the left of the listing.

	0	Add Event			🔯 Clear Old Events 🛛 📆 Clear Schedule	🁔 Upload Ever	nts
		Start Date	Start Time	Time Slot Duration	File Name	File Duration	
	-	00/28/2017	16:30:03	00:01:55	fransat.ts	00:01:55	0
$\subseteq$	₽	09/28/2017	16:29:03	00:01:00	balloons_hevc_1080p30_420_08bit_12Mbps_L4.0_MT_3.trp	00:01:00	0
		09/28/2017	14:56:57	01:32:06	monty_python.trp	01:32:06	0

The Moving Cursor is used to drag and drop a listed event to a time earlier or later in the schedule. To move a listed event, hoover your mouse over the Moving Cursor location at the left side of the listing, left mouse click on the Moving Cursor, drag your mouse moving the listing up or down in the listed schedule, and release the left mouse click to position the event at the new location in the schedule.

#### 5.5.3 Scheduling Configuration Panel - Loading Event Lists

The Configuration Panel provides the ability to upload an event schedule that is created and/or edited in a spreadsheet. To upload a schedule of events, click on the **1** Upload Events icon at the upper right of the display menu. Browse to the CSV file containing your schedule. Select and open the file to upload the listed events from the CSV file to the Schedule. Once in the schedule you can move or edit uploaded events.

6	) Add Event			🔀 Clear Old Events 🛛 🔒 Clear Schedule	Upload Eve	nts
	Start Date	Start Time	Time Slot Duration	File Name	File Duration	
	09/28/2017	16:30:03	00:01:55	fransat.ts	00:01:55	0
<b>P</b>	09/28/2017	16:29:03	00:01:00	balloons_hevc_1080p30_420_08bit_12Mbps_L4.0_MT_3.trp	00:01:00	0
	09/28/2017	14:56:57	01:32:06	monty_python.trp	01:32:06	0

When uploading events, the values in the CSV format must match the columns in the TSS 6220 schedule for date, time, duration, and file naming. File naming must accurately match file naming of play files. Below is a sample file showing CSV formatting.

F	ile Home	Insert I	Page Layout	Formul	as Data	Review	View	Acrobat	Team
	🗎 🔏 Cut	Calibr	i	* 11 *	A A ≡	= =	≫,-	📲 Wrap Text	Ge
Pa	ste 🛷 Format Pair	nter B	<u>u</u> .	💷 🔹 🍐	• <u>A</u> • 🔳		ŧ∎ ŧ∎	📲 Merge & O	Center * \$
	Clipboard	- Fai	For	nt	Fai		Alignme	nt	Gi -
	11	• (*	$f_{x}$						
	А	В	С	D	E	F	G	Н	1
1	9/21/2017	18:25:00	0:01:00	/balloons	_hevc_108	0p60_420_	08bit_30M	Mbps_L4.1_H	T_3.trp
2	9/21/2017	18:26:00	1:50:12	/SILENT_E	BOB.trp				
3	9/21/2017	20:16:12	1:50:12	/SILENT_E	BOB.trp				
4	9/21/2017	22:06:24	1:32:06	/monty_p	ython.trp				
5	9/21/2017	23:38:30	1:50:12	/SILENT_E	BOB.trp				
6	9/22/2017	1:28:42	0:04:59	/NASA_H	EVC4K_5m	in_SPTSv2.	trp		
7									



# 6 Delay Viewing Panel

The Delay Viewing Panel provides configuration and monitoring of the time shifted output(s) of the TSS 6220. The Time Delay feature's primary application is for time-zone shifting. An input TS stream is configured for receive and sent to a record buffer. The buffered TS stream data is time stamped and after the specified delay is output to a specified physical port to a defined destination address/port.

The Delay feature of the TSS 6220 is an optional licensed feature. When licensed the Delay tab is present in the main menu and available for selection. Each delayed in-to-out line (Single Delay Line) is a licensed feature. Multiple single delay line licenses may be purchased as needed.

If a particular buffered input (Defined Delay) must have two or more different outputs with different delays, a Multi Transmit Delay License is needed. This enables multiple delay outputs with different defined delays and/or output ports and addresses. An instant of the Multi Transmit Delay License is needed for each of the single delay lines in which multiple output delays are needed.

The following sections in this chapter describe configuration and monitoring menus for receive, buffering and outputting of the TS streams in the Delay function.

Play	Rec	ord Schedule	Delay	Disaster Recovery	Files	Admin	Reporting	About				
Delay	Contro	l Panel										
🛈 Add	Delay											
Dela	y 1											
्रिः Co	onfigur	e Delay 💿 Add 🛛	)elay Transr	nit						C	Remo	ve
Ŧ	0 I	Receive		Physical Connector	eth1		239.192.1.30:1	030		19.387 N	lbps	
٠	0 1	Buffer		State: Normal Ope	ration	6	00	:16:00	Buffer Size: 2.4 GB			
٠	ا 🕲	Transmit 1(16 min)		Physical Connector	eth1		239,192.5.200	10001	Original Time: 2018-06-21 15:10:37 🤤	19.393 N	lbps	
Ŧ	ا 🕸	Transmit 2(2 min)		Physical Connector	eth1		239.192.0.220	10009	Original Time: 2018-06-21 15:24:38 🤤	19.393 N	lbps	
Dela	y 2											٥
୍ଦିକ Co	onfigur	e Delay 🔘 Add 🛛	elay Transr)	nit						C	Remo	ve
Ð	0 I	Receive		Physical Connector	eth1		239.192.1.80:1	080		0.000 N	lbps	
Ŧ	• I	Buffer		State: Buffering			00	:00:00	Buffer Size: 0 Bytes			۲
Ŧ	ا 😵	Transmit 1(10 min)		Physical Connector	eth1		239.192.0.220	1005	Original Time: GAP(BALLOONS.TRP	19.393 N	lbps	



## 6.1 Delay Panel Overview

The Delay Panel is used to add or create a delay line or path routing an input stream to a memory buffer and after the specified delay time routing it to an output. The TSS 6220 provides flexibility in setting up the receiver, buffer, delay, and output criteria. When a delay is created and defined, the information is shown in the Delay Panel. For each created delay, there is a row in the panel indicating the receive criteria, a row showing the buffer criteria, and a row indicating transmit or output criteria.

Depending on licensing, multiple delays may be created with unique inputs, delays and outputs. Also, multiple outputs are possible from a particular buffered input. The following diagram and descriptions provide an overview of the layout and fields common in the Delay Panel. The remainder of this chapter provides additional details of configuration menus, informational fields in the panel, and dropdown information menus.

Play Record Schedule Delay	Disaster Recovery Files	Admin Repo	orting About			
Delay Control Panel Add Delay Delay 1 Configure Delay Add Delay Tran	smit D				E Remove	F
a o Receive	Physical Connector: eth1	239.19	2.1.30:1030	(	Mbps 🚱	G
o Buffer	State: Normal Operation		00:05:00	Buffer Size: 774.5 MB	H	
C Transmit 1(5 min)	Physical Connector. eth1	239.19	2.5.200:10001	Original Time: 2018-06-21 08:43:02 🥥	19.393 Mbps 🕘	<u> </u>

Item or	r Field	<b>Button/Action</b>	Description
Α.	Select Delay	Click on this field to access the Delay feature	Provides access to the Delay feature and all configuration and monitoring menus. Delay license required.
В.	Add Delay	Add Click this icon to add a delay to an output ethernet port	Provides access to menus to add and configure receive, buffer, and transmit criteria for the delay
C.	Configure Delay	Click this icon to access control menus	Provides access to menus to setup/control receive, buffer and transmit criteria for the delay
D.	Add Delay Transmit	Add Click this icon to add a delay	Adds a second transmit output for the existing delay, Requires Multi Transmit Delay license. This selection only available with feature license.
E.	Delete Delay	Remove Click on this ito delete this delay	Removes the delay from the panel, stops input receive, buffering and transmit of this delay.
F.	Show/Hide Delay fields	Selectable, click on the icon	Collapses the receive, buffer, transmit rows to not be visible in the panel. Expands to show receive, buffer, and transmit rows when collapsed
G.	Receive	See section: 6.2	This row shows the receiver information for the Delay
н.	<b>Buffer Section</b>	See section 6.3	This row shows the buffer information for the Delay
I.	Transmit section	See next section 6.4 for field descriptions	This row shows the transmit output information of the defined Delay



## 6.2 Delay Panel – Receive Fields

The Delay Panel contains a row of information that is identified by a "Receive" indication. This row of information summarizes the configuration of the receiver for the created delay. The information in the fields indicate user settings that were specified when the Delay was added or created. See section 6.5.2 in this manual for setup of the Delay receive. The fields also indicate current status on the input stream and bitrate. The following information provides a description of each field in the Receive row.



Receive row field descriptions:

Item or	Field	<b>Button/Action</b>	Description
Α.	Adds Status Menu to Delay Panel	Click icon to add dropdown menu	Adds a dropdown section to the Receive row showing additional Configuration & Status information. See section 6.7.
В.	Receive	Not a selectable field	Identifies row as information relative to receive and input stream of the Delay
C.	Physical Input Ethernet Port	Not a selectable field	Indicates the user defined receive or destination input IP address and port. Or, indicates ASI port when selected.
D.	Input Destination Address/Port	Not a selectable field	Indicates the user defined receive or destination input IP address and port. Indicates ASI input sync status when ASI input port is selected.
E.	Input Bitrate	Not a selectable field	Indicates the incoming TS stream bitrate
F.	Status Indicator	Status Light, not a selectable field	Red: No input stream, abnormal input or condition
			Green: Input active/enabled – input normal
			Gray: input inactive/disabled – input not setup



## 6.3 Delay – Buffer Information Fields

The Delay Panel contains a row of information that is identified by a "Buffer" indication. This row of information summarizes the configuration of the Buffer for the created delay. The information in the fields indicate user settings that were specified when the Delay was added or created. See section 6.5.3 in this manual for related setup settings for the Delay buffer. The row further indicates status information in regard to buffering the input stream. The following information provides a description of each field in the Buffer row.



Buffer row field descriptions:

Item or	Field	<b>Button/Action</b>	Description
Α.	Adds Configuration Status Menu to Delay Panel	Click icon to add dropdown menu	Adds a dropdown section to the Buffer row showing additional buffer configuration information. See section 6.8 for details
В.	Buffer	Not a selectable field	Identifies row as information relative to receive buffer
C.	Buffer state	Not a selectable field	Indicates the current state of the buffer
			Buffering: Building a buffer to specified time delay
			Normal Operation: - actively buffering delayed output
D.	Extract File Menu	Click on icon, provides an Extract Buffer to File menu	Provides a menu in which the time referenced stream data in the buffer memory can be named and extracted to a file
E.	Buffer Time Delay	Not a selectable field	Indicates when buffer is filling as moving blue filler, indicates time of data capture or delay time of buffer
F.	Buffer Size	Not a selectable field	Indicates the buffer size as drive storage space being utilized
G.	Status	Status Light, not	Red: No output stream, abnormal output or condition
	marcator	a selectable field	Green: Input active/enabled – output normal
			Gray: Input inactive/disabled – output not setup



## 6.4 Delay – Transmit Information Fields

The Delay Panel contains a row of information that is identified with a "Transmit" indicator. This row of information summarizes the configuration of the delayed transmit or output. The information in the fields indicate user settings that were specified when the Delay was added or created. See section 6.5.3 in this manual for setup of the Delay transmit criteria. The fields also indicate current status information of the delayed output. The following information provides a description of each field in the Transmit row.



Transmit row field descriptions:

Item or	Field Name	<b>Button/Action</b>	Description
Α.	Status & Configuration	Click on this icon	Provides an added window showing Delay output status and configuration information. See section 6.9 for details.
В.	Configuration Menu Select	Click on this icon	Provides configuration menus to define/modify the delay's input receive, buffer and transmit criteria
C.	Identifies Transmit section	Not selectable, No action	Identifies the row as related to the delay's output or transmit criteria
D.	Physical Connector/port	Not Selectable field, view only	Indicates the Physical Ethernet port specified inwhich the delay stream in the listed schedule will playout. Or, indicates the ASI Output port when used
E.	IP Address/Port	Not Selectable field, view only	Indicates the destination IP address and port of delay output. Indicates the ASI output Null Stuffing status when ASI output port is in use
F.	Original Time	Not Selectable field, view only	Provides reference of the original date and time that the delay is referenced. Time stamps of buffered data
G.	Delete icon	Click <a> to delete</a> the schedule	Removes a schedule from the listed schedules
Н.	Total Port Bitrate Indication	Not selectable, view only	Indicates Ethernet port bitrate of the delayed output TS streams
I.	Status Indicator	Not Selectable	Indicates status of output: Gray: Inactive – output stopped, Green: Good active playout, Red: Playout fault or no output

## 6.5 Delay – Adding or Creating a Delay Line

To create a new delay input to output line or path requires that you define receive, buffer and ouput parameters. To create a Delay click on the Add Delay icon. This icon is located just below the Delay Control Panel heading. If you do not see this section, click on the Expand Panel icon located at the right. Note that this selection is available for each of the Delays that are available.

To add a delay requires that you configure general, receive, and output parameters. These parameters are found in General, Receive, and TxDelay menus that are available upon clicking the <sup>O</sup> Add Delay</sup> icon. The next sections describe these menus and their configuration.

Play Record	Schedule	Delay	Disaster Recovery	Files	Admin	Reporting	About
Delay Control P	anel						
O Add Delay	$\mathbf{)}$						

#### 6.5.1 Add Delay - General Configuration - IP

When clicking on the <sup>O</sup> Add Delay icon, the Add Delay menu is shown. There is a General, Receive, and Transmit (TxDelay) menus which are selected by clicking on the available tabs. The General configuration menu is shown by default. This menu provides alias naming and definition of the Gap Mode. Click on the Alias field and enter an alias name, if desired. This is not required as the application automatically assigns a name as an incrementing delay number.

The Gap Mode importantly defines what the output is when there is no output transport stream to output from the delay buffer. This occurs when the buffer is buffering and has not yet reached the specified delay time. It also occurs when the input TS stream is lost



and the buffer streams out until it is depleted. It can also occur when bad data is contained in the buffer because of an input stream issue.

The Gap Mode user selections include: No Output, Null packets, Default File, and Live . The No Output setting would provide no output TS stream when a gap occurs. The Null Packets setting would fill the output ts stream with Null Packets when a gap occurs. A Live setting would route the input ts stream directly to the output bypassing the delay buffer when a gap occurs. When set to Default File, the Source field defines the TS transport stream that plays out during a gap. Click on the Browse tab and choose the stream to be the default file.

Setting	Range/Selections	Description
Stream – Alias	Value and or Letter Entry	Provides entry of alias to identify a delay. If no alias is entered, the TSS 6220 assigns a delay number.
GAP Mode	Click Dropdown – select from listed	Determines the output when a gap exists and there is no TS stream from the buffer. Selections include: Default File: Outputs choose TS file No Output: Outputs no output stream Null Packets: Outputs TS stream with null packets Live: Routes the input TS stream directly to output
Source	Not selectable	Indicates "No Source" when no file has been selected. Indicates the selected file name when a file is selected.



#### 6.5.2 Add Delay - Receive Configuration - IP

The Add Delay Receive menu provides configuration of the IP input used to receive the MPEG-IP unicast or multicast. The selected input TS stream is routed to the delay buffer. This section provides descriptions of the settings in this menu.

Configure the IP input to be active by setting the Receive field to "Enabled." Select the Physical Connector or Ethernet port on the TSS 6220 to use as the receive port. Eth 0 and Eth 1 are available. If the added ethernet port hardware option is added to your TSS 6220, then eth 2 and eth 3 will be available in the dropdown.

Select the Mode of the receiver to be Unicast or Multicast. For unicast, specify the destination port in the Destination IP field. For Multicast, specify the Destination IP address, and Destination IP Port.

Add Delay		
General Receive Tx Dela	iy 1	
Receive:	Enabled	-
Physical Connector:	eth1	~
Mode:	Multicast	-
Destination IP:	239.192.0.200	
Destination Port:	10000	\$
IGMP Filter Mode:	Exclude	-
Add IGMP Address	🥥 Re	nove All
IGMP Address		Remove
	Apply	Cancel

The IP configuration section further includes settings to provide IGMPv3 features. An IGMP filter may be implemented for use to specify the inclusion or exclusion of source addresses. The TSS 6220 is IGMPv3 compliant. IGMPv3 allows each steam to be seen by the network as relating to a unique source device with a unique IP address, port, and/or MAC address. IGMPv2 is used to join/leave multicast streams by default if no IGMP Filter addresses are entered in the Add IGMP Address section of the menu. If IGMP Filter Mode addresses are specified then IGMPv3 is automatically used.

Settings	Range	Description
IP Receive	Enable	This setting allows the user to enable or disable
	Disabled	these input stream settings.
IP – Physical Connector	Selects eth0 or eth1, the standard Ethernet ports,	Entry selects which of the available Ethernet ports on the TSS 6220 that the stream record input is assigned
	eth2 and eth3 available if TSS 6220 has added optional ports	
IP Mode	Multicast Unicast	<i>Multicast</i> setting allows the unit to receive multicast streams. Multicast streams originate from the IP range 224.0.0.0 – 239.255.255.255. <i>Unicast</i> allows the unit to receive unicast streams. Unicast streams originate directly from a source device.
Destination IP	Enter Value: 224.0.0.0 - 239.255.255.255	This address is the IP address the source device is sending to for a multi-cast. A unicast would use an IP address less than 224.0.0.0. This setting is only available when receiving a multicast stream. This address is the IP address the source device is sending to.
Destination Port	Enter Value: 0 – 65535	This is the UDP port the source device is sending to. This is the only setting required to receive a



		unicast stream.
IGMP Filter Mode	Settings: Select Include or Exclude	Used on networks supporting IGMPv3. If this setting is set to <i>Exclude</i> any streams originating from the user defined IP addresses will be rejected. If this setting is set to <i>Include</i> any streams originating from the user defined IP addresses will be received.
Add IGMP Address	Click in field - Enter IP address to include or exclude as per filter mode: Values: 0.0.0.0 – 255.255.255.255	Enter and list IP address of IGMPv3 to include or exclude as a filter setting.
Remove All	Click on icon	Removes or clears all the listed IGMPv3 address

#### 6.5.3 Add Delay – Transmit Configuration - IP

The Add Delay Transmit (Tx Delay) menu provides configuration of the Delay output. Click on the Tx Delay tab (Example: Treevel). If licensed for multiple transmit outputs, more than one tab is shown with incrementing tab numbers. Select and configure each Transmit (Tx Delay) output.

Configure the delayed transmit output to be active by setting the Transmit field to "Enabled." Select the Physical Connector or Ethernet port on the TSS 6220 to use as the receive port. Eth 0 and Eth 1 are available. If the added ethernet port hardware option is added to your TSS 6220, then eth 2 and eth 3 will be available in the dropdown. Note that the receive and transmit ports for the delay may be the same or different ports.



Select and enter the Destination IP and port values. For unicast, specify the destination port in the Destination IP

field. For Multicast, specify the Destination IP address, and Destination IP Port.

The Tx Delay tab includes settings to define the TSS 6220 as a specific source device for IGMPv3. This allows each steam to be seen by the network as a unique source device with a unique IP address, and/or source port, and/or MAC address. This section provides descriptions of the settings.

When finished with all the selections in the General, Receive, and TxDelay menus, click on the Apply box at the bottom to apply changes and create the delay. The delay is added to the Delay Panel. You can access the configuration menus for changes by clicking on the cog wheel in the Transmit row of the created Delay.

Setting	Range	Description
Transmit	Selects Enabled or Disabled	Enable or disables the output stream and related menu settings.
		Enabled: Source & Destination settings and output streaming is active
		Disabled: Source & Destination settings and output



Delay (minutes)	Select – enter delay in minutes	Entry of the number of minutes the output is delayed in respect to the input. The time delay of the stream buffer
IP – Physical Connector	Selects eth0 or eth1, the standard Ethernet ports, eth2 and eth3 available if TSS 6220 has added optional ports	Entry selects which of the available Ethernet ports on the TSS 6220 that the transmit delay stream is assigned to output
Destination IP	224.0.0.0 – 239.255.255.255	This address is the IP address the source device is sending to for a multi-cast. A unicast would use an IP address less than 224.0.0.0.
Destination Port	0 – 65535	This is the UDP port the source device is sending to.
Source IP Mode	Select Auto or Manual	Provides entry of how the TSS 6220 communicates a source IP address
		Auto: Allows TSS 6220 to automatically select and communicate the source IP address
		Manual: Provides user entry of the source address
Source IP	Available for entry when Source IP Mode is set to Manual.	Provides entry of a Source IP address for the stream that is communicated by the TSS 6220
Source Port	Click up or down arrows to increment value shown. Click in field and enter value. Range: 1030 to 65535	Specifies a value for the source port associated with the stream.
Source MAC Mode	Select Auto or Manual	In Auto, the TSS 6220 simulates a source device and creates and communicates a unique source MAC address for the stream. In Manual, a user Source MAC address can be entered.
Source MAC	Available when Source Mac Mode Manual - enter MAC address	Provides entry of a MAC address you want to specify as the Source MAC for the stream communicated by the TSS 6220
TS Packets Per IP Packet	Enter value 1 to 7, Default is 7	This setting determines the number of TS stream packets that are inserted into IP packets. 7 being the maximum and the typical setting. Lesser packets may be selected.
Encapsulation	Select UDP or RTP	Selects IP encapsulation to UDP or RTP

#### streaming is inactive (off)



### 6.5.4 Add Delay – General Configuration – ASI In/Out

The following sections describe the Delay menus and configuration when the TSS 6220 is equipped with the optional ASI input/output hardware. Click on the Add Delay icon to create a Delay input to output path. This opens the Delay configuration menus. There is a General, Receive, and Transmit menu which is selected by clicking on the available tabs. The General configuration menu is shown by default. This menu provides alias naming and definition of the Gap Mode.

Click on the Alias field and enter an alias name, if desired. This is not required as the application automatically assigns a name as an incrementing delay number.



The Gap Mode importantly defines what the output is when there is no output transport stream to output from the delay

buffer. This occurs when the buffer is buffering and has not yet reached the specified delay time. It also occurs when the input TS stream is lost when the buffer streams out until it is depleted. It can also occur when bad data is contained in the buffer because of an input stream issue.

The Gap Mode user selections include: No Output, Null packets, Default File, and Live. The No Output setting would provide no output TS stream when a gap occurs. The Null Packets setting would fill the output ts stream with Null Packets when a gap occurs. A Live setting would route the input ts stream directly to the output bypassing the delay buffer when a gap occurs. When set to Default File, the Source field defines the TS transport stream that plays out during a gap. Click on the Browse tab and choose the stream to be the default file.

Setting	Range/Selections	Description
Stream – Alias	Value and or Letter Entry	Provides entry of alias to identify a delay. If no alias is entered, the TSS 6220 assigns a delay number.
GAP Mode	Click Dropdown – select from listed	Determines the output when a gap exists and there is no TS stream from the buffer. Selections include: Default File: Outputs choose TS file No Output: Outputs no output stream Null Packets: Outputs TS stream with null packets Live: Routes the input TS stream directly to output
Source	Not selectable	Indicates "No Source" when no file has been selected. Indicates the selected file name when a file is selected.



#### 6.5.5 Add Delay – Receive Configuration – ASI In/Out

The Add Delay Receive menu provides selection and configuration of the ASI input used to receive the incoming stream. When the TSS 6220 contains the optional ASI input/output hardware, a Receive Type field is included in the Receive Menu. This field configures the Delay input as ASI or MPEG-IP. In the Receive Type field click on the drop down arrow and select ASI. This configures the TSS 6220 to receive and rout an ASI input TS stream for the created Delay. It further modifies the Receive menu for ASI input configuration.

When the Receive Type is ASI, the Receive menu provides selection for enabling or disabling the input port. It further provides selection of one of the available ASI hardware ports. Set the Receive field to "Enabled" to start receiving and buffering the incoming stream. If the



Receive field is set to "Disabled" the incoming TS stream on the selected port is not received or routed to the delay buffer.

The optional ASI hardware contains 4 ASI ports. These ports can be configured by the TSS 6220 as an input or as an output port. The ASI ports are available for use by licensed features of the TSS 6220 including the Play, Record, Delay, and Disaster Recovery features. An ASI port cannot be shared by these features. Once a port is assigned or in use by a feature, the port becomes unavailable (greyed out) for the other features. Click the drop down arrow in the Port field to view available input ports when adding a Disaster Recovery.

Setting	Range/Selections	Description	
Receive Type	ASI or MPEG-IP select, click field dropdown arrow and select	Provides selection of an input port for the Disaster Recover function. ASI: selects ASI option card MPEG-IP: select MPEG-IP port	
Receive	Enable or Disabled select, click dropdown arrow and select	Enables or Disables the receive ASI input and selected port. Enabled: Incoming TS stream on the Port specified is routed to the delay buffer. Disabled: The select ASI input Port is disabled.	
Port	Selects ASI input Port 1, 2, 3, or 4, click dropdown and select from list of available ports	Selects from available ASI ports. Ports listed are available for use and not used by other TSS 6220 features (Play, Record, Delay, Disaster Recovery). The ASI hardware contains 4 ASI ports. These ports can be either used as inputs or output ports.	

ASI Receive Menu Descriptions



#### 6.5.6 Add Delay – Transmit Configuration – ASI In/Out

The Add Delay menu provides configuration of the delay's output. Click on the Transmit tab to configure the output criteria of the Delay output. This section describes the selections in the Transmit configuration menu when an ASI output port is selected.

The Add Delay Transmit (Transmit 1) menu provides configuration of the Delay output. If licensed for multiple transmit outputs, more than one tab is shown with incrementing tab numbers. Select and configure each Transmit (Transmit x) output independently.

When the Transmit Type is ASI, the menu provides selection for the physical ASI port. The optional ASI hardware contains 4 ASI ports. These ports can be configured by the TSS 6220 as an input or as an output port. The ASI ports are available for use by licensed

Add Delay		
General Receive Tr	ansmit 1	
Transmit Type:	ASI	Ŧ
Transmit State:	Enabled	-
Delay (minutes):	60	\$
Port:	ASI Port 2	-
Null Stuffing:	Disabled	~
Total Bitrate (Mbps):	0	
	Apply	Cancel

features of the TSS 6220 including the Play, Record, Delay, and Disaster Recovery features. An ASI port cannot be shared by these features. Once a port is assigned or in use by a feature, the port becomes unavailable (greyed out) for the other features. Click the drop down arrow in the Port field to view available output ASI ports. When configuring an ASI output, select from the list of available ports.

When considering applications and licensing for multiple delays, be sure to consider the hardware configuration regarding ASI ports. Each delay output via ASI would require an independent ASI port. If the port is used as an input by any of the TSS 6220 features, it would not be available for a Delay output application.

The Null Stuffing field provides an option to add null stuffing bytes to the Disaster Recovery output to increase the output bitrate. Click on the field and enable the features. The Total Bitrate field below becomes available to enter a desired bit rate. Enter the desired bit rate, in Mbps, in the Total Bitrate field.

Setting	Range/Selections	Description			
Transmit Type	ASI or MPEG-IP, click field dropdown arrow and select	Provides selection of an output port for the Disaster Recover function. ASI: selects ASI option card MPEG-IP: select MPEG-IP port			
Transmit State	Selects ASI output as	Enabled:			
active "Enabled" or inactive "Disabled. Click dropdown arrow and select.		Disabled			
Delay (minutes)	Select – enter delay in minutes	Entry of the number of minutes the output is delayed in respect to the input. The time delay of the stream buffer			
Port	Selects ASI input Port 1, 2, 3, or 4, click dropdown and select from list of available ports	Selects from available ASI ports when Transmit Type is ASI. ASI Ports listed are available for use and not in use by other features (Play, Record, Delay). The ASI hardware contains 4 ASI ports. These ports can be either used as inputs or output ports. Once in use they are unavailable and are			
Null Stuffing	Select and choose either disabled	Disabled: The ASI output bitrate is determined automatically by the TSS 6220. The Total Bitrate			



	(default) or enabled.	(Mbps) field below in the menu is grayed out.
		Enabled: The TSS 6220 adds null stuffing to increase the playout bitrate of the ASI output stream. The user enters the desired playout bitrate in the Total Bitrate (Mbps) field in the menu.
Total bitrate	Available when Null Stuffing is enabled. Click on the field and enter a value in Mbps	Enter a value in Mbps of the desired ASI output bit rate. The TSS 6220 adds null bytes to the stream to increase the total bit rate to the entered value in Mbps.



## 6.6 Delay Buffer – Extract Buffer To File

The TSS 6220's Delay feature has an active buffer receiving TS input data stream and outputting delayed ts stream data. The buffer contains an accumulation of TS captured data in a proprietary format that extends in duration slightly beyond the specified delay time(s). The incoming buffered TS data is time stamped according to its arrival time by the system clock. The TSS 6220 offers the ability to specify a time relative to the incoming time stamps and extract a duration of the ts captured data from the buffer to a memory file. The extracted transport stream data can be written to a file in a common ts or trp format.

In the Delay's buffer section of the web GUI exists an Extract Buffer to File icon  $\blacksquare$ . It is located near the middle of the page in the Buffer section/row. Click on this icon to access the Extract Buffer to File menu. This section describes the Extract Buffer to File menu and how to extract the time stamped buffer data.

De	elay	/ 1							
<u>್</u> ರಿ	Co	onfigu	ure Delay	Add Delay Transmit					
6	Ð	0	Receive		Physical Connector: eth1	_	239.19	2.1.30:1030	
6	Ð	0	Buffer		State: Normal Operation		)	00:16:00	Buffer Size: 2.5 GB

The Extract Delay Buffer To File Menu includes selection of a Start Extract Date and Start Extract Time. This date & time selection permits a user to specify a day and time of the original arrival

time of the TS input data to the buffer. The selections must be a time which matches time stamps of data currently in the buffer. Click on the Start Extract Date field and enter a date. Or, click on the calendar icon and select a day. Click in the Start Extract Time field and edit or enter the extract time.

PLANUU IN	1012-102-101	
Extract Delay 1 Buffer To File		
Start Extract Date:	2018-06-25	
Start Extract Time:	13:13	
Extract Duration(minutes):	1	-
Extract File:	Extract_File_6-25-2018	i.trp
	Brow	se
Extract To File		

The data extraction requires a duration of time entry in the Extract Duration (minutes) field. Click in the field and enter a value, in minutes. Or click on the up and/or down arrows at the right of the field. The

duration must not exceed time stamps currently in the buffer. Reference the time indicated by the Original Time field in the Transmit section and the Delay duration setting to determine applicable start and duration entries for extraction compatibility.

Extracted data is written to a file selected by clicking on the Browse icon and selecting a current file. Or, click on Browse and enter a new file name in the bottom field of the Browse Files menu.

Finally, click on the Extract To File icon at the bottom center of the menu to start the data extraction. A progress bar will indicate the process is being implemented. If time and duration entries are not found in the time



stamped buffer data, a message will indicated. Upon conclusion of the extraction, a Buffer Extraction Success message appears. Click the OK field to acknowledge the extraction as completed.

Settings	Range	Description
Start Extract Date	Click on calendar IIII icon select date, or	This setting (date) marks the day in which the buffer data is extracted from memory and written to a file. Note: The selected date is typically



	enter date infield YYYY,MM,DD	today's date unless the delay is > 24 Hrs. The day must be in current buffer data time stamps.
Start Extract Time	Click in field, Enter time	Entry references time stamp (hours, seconds) of data in the buffer in which the data extraction starts. Note: The time reference must be within current buffer.
Extract Duration (minutes)	Time	Field indicating the time duration of the data following the start extract time that is to be extracted and written to the file.
Extract File	Click Browse field to go to Browse File menu. Select an existing file to overwrite. Or enter a file name at the bottom of the menu.	This is file name that the buffer data is extracted to. If "No Source" is shown you need to select an existing file to replace or create a new file.
Browse	Click on Browse field	Takes you to the Browse Files menu. Enter a file for selection or entry of a file name, once a file is named or selected the file name is indicated in the Extract File field
Extract to File	Click on Extract To File field to apply entries	Extracts the buffer data starting at the date/time referenced for the duration specified to the selected file name. Convert data to the file type specified.



## 6.7 Delay Receive Status & Configuration - Information Menu

Each of the Delays in the Delay Panel includes in the Receive row an added Status and Configuration window. To access this window and add it to the panel, click on the <sup>III</sup> icon at the left of the Receive row. The Status and Configuration boxes are added to the Play Control Panel providing status and configuration details. Click on the <sup>III</sup> icon at the same location to hide the status and configuration information. Following is a description of the fields and information found in this Status and Configuration window.

onfigure Delay 🔘 Add Delay	Transmit	
Receive  Status	Physical Connector: eth1 Configuration Mode: Multicast	239.192.1.30:1030 IGMP Filter List
Packets Per Frame: 7 Encapsulation: RTP	IGMP Mode: Exclude	Unsolicited IGMP Report

The Sync Status, Packets Per Frame and Encapsulation fields provide information regarding the incoming receive transport stream. Should the Sync Status indicate "Unlocked" check the IP configuration settings and/or availability of the MPEG-IP stream.

Status Listing	Description
Sync Status	Indicates the source TS stream is being received and TS sync is established. Locked: Indicates receiving and locked to TS stream Sync
Packets Per Frame:	Indicates the TS packets per TS frame in the incoming TS stream
Encapsulation:	Indicates receive IP stream encapsulation, RTP, UDP

The Configuration part of the window provides information regarding the settings for the Delay input. The Configuration window also indicates the IGMP receive mode, unicast or multicast, along with IGMPv3 settings and filter addresses.

Configuration Listing	Description
Mode:	Indicates if input receive is configured as Unicast or Multicast
IGMP Mode:	Indicates Include or Exclude mode for IGMPv3 address entry
Add IGMP Filter	Indicates listed IGMP filter addresses to Exclude or Include for Source Specific IGMPv3



## 6.8 Delay Buffer Configuration – Information Menu

In the Buffer section of each of the delays there is an added Configuration information window available. To access this window and add it to the panel, click on the <sup>III</sup> icon at the left of the Buffer section in the panel. The Configuration window is added to the Delay panel. Click on the <sup>III</sup> icon at the same location to hide the status and configuration information. Below is a description of the information you will find in this information window.

p Buffer	State: Buffering	<b>a</b>	00:01:24
onfiguration			1
p Mode:	No Output		
ap File:	NO SOURCE		
aximum Buffer Duration(minut	es): 5		

The Configuration window includes information regarding the Gap Mode and Gap File. It further includes the Maximum Buffer Duration listed in minutes. Recall that the Gap Mode permits selection of what you want to be outputting when there is no TS stream available to output from the buffer. A transport stream file may be selected to output during a gap condition by setting the Gap Mode to Default File. A selected file for this is indicated in the Gap File field of the Configuration window.

Configuration Listing	Description
Gap Mode:	Indicates the Gap mode:
	Default File: Outputs selected TS file No Output: Outputs no output stream Null Packets: Outputs TS stream with null packets Live: Routes the input TS stream directly to the Delay output
Gap File:	Indicates the ts file in which to play out when the Gap Mode is set to Default File and there is no TS data in the Buffer to output.
Maximum Buffer Duration (minutes)	Indicates the maximum Buffer time in minutes



## 6.9 Delay Transmit Status – Information Menu

An added Status and Configuration window is available in the Transmit section of each of the delays listed in the Delay Panel. To access this window and add it to the panel for viewing, click on the <sup>III</sup> icon at the left of the Transmit row. The Status and Configuration boxes are added to the Play Control Panel providing status and configuration details. Click on the <sup>III</sup> icon at the same location to hide the status and configuration information. This section of the manual provides a description of the fields and information found in this Status and Configuration window.

anansmit (15		Filysical Connector.	euri	238.182.5.200.1000
Status		- Configuration		
Source IP: 10.0.0.	66	Source IP Mode:	Auto	
Source MAC: 0C:C4:	7A:C9:AB:41	Source Port:	3020	
Mode: Multicas	st	Source MAC Mode:	Auto	
Receiver MAC: N/A		TS Packets:	7	

The Status part of the window provides information regarding the Delay's output status. It includes an indication of its source IP address and its MAC address. It indicates if the output is in a unicast or multicast configuration status. It indicates a MAC address from a destination receiver if applicable.

Status Listing	Description
Source IP	Indicates the Source IP address
Source MAC	Indicates the Source MAC address,
Mode	Indicates Transmit IP mode, either Unicast or Multicast
Receiver MAC	Indicates a MAC address as indicated by a destination receiver

The Configuration part of the window provides information regarding the Delay's transmit user settings selected in the TX Delay configuration menu. See section 6.5.3 in this manual.

Configuration Listing	Description
Source IP Mode:	Auto or Manual
Source Port	Indicates the port value selected
Source MAC Mode	Indicates Automatic or Manual as selected in the TX Delay setup menu.
TS Packets	Indicates the number of TS packets per IP packet as selected by the user in the TX Delay setup menu


# 6.10 Multi Transmit Delay – Tx Delay 2 Configuration

The TSS 6220 can provide multiple transmit outputs from a Delay buffer. The added transmit output can specify a different delay time, a different physical output Ethernet port, and unique output address/port configurations. A Multi Transmit Delay License is needed to add this capability to the Delay feature. For each Delay that requires more than one output, an additional instance of a Multi Transmit Delay License is needed.

When the TSS 6220 contains Multi Transmit Delay licensing, after a Delay is added and shown in the Delay Panel, you will see an Add Delay Transmit <sup>O</sup> Add Delay Transmit icon. This icon is located in the Delay Panel heading as illustrated below. Click on this icon to add a second transmit output to the existing Delay. This manual section describes the related settings and menus.

Delay 1									
()):	Config	ure Del y 💿 Add Delay Transmit	)				🤤 Rei	move	
٠	0	Receive	Physical Connector: eth1 239.192.1.30:1030				19.387 Mbps	۲	
H	0	Buffer	State: Normal Operation		00:16:00	Buffer Size: 2.4 GB		۲	
Ð	٨	Transmit 1(16 min)	Physical Connector: eth1		239.192.5.200:10001	Original Time: 2018-06-21 15:10:37 🤤	19.393 Mbps	۲	
٠	-	Transmit 2(2 min)	Physical Connector: eth1		239.192.0.220:10009	Original Time: 2018-06-21 15:24:38 🤤	19.393 Mbps	۲	

Upon clicking the Add Delay Transmit icon

G Add Delay Transmit, the Configure Delay menu appears. The menu now contains an additional tab in which to configure a 2<sup>nd</sup> delayed output (Example: Tx Delay 2). The TS 6220 adds tabs with incrementing Tx Delay numbers. The new Tx Delay tab is selected by default and provides configuration of the added delay's output. The table below summarizes the available configuration settings.

When finished with all the selections in the added TxDelay menu, click on the Apply box at the bottom to apply changes. The delay is added to the Delay Panel. You can access the configuration menus for changes by clicking on the cog wheel in the Transmit row of the created Delay.



It should be noted that the Delay time for the added Delay can be longer or shorter in duration compared to the existing delay(s). If the delay is longer, upon adding the added delay, the TSS 6220 responds by adding more buffered data to extend the existing buffer time. The newly created delay will not output original delayed input TS stream unit the buffer has built up the needed buffered delay time. The Gap Mode determines the output of the TSS 6220 during this buffer build up delay.

Setting	Range	Description
Transmit	Selects Enabled or Disabled	Enable or disables the output stream and related menu settings.
		Enabled: Source & Destination settings and output streaming is active
		Disabled: Source & Destination settings and output streaming is inactive (off)



Delay (minutes)	Select – enter delay in minutes	Entry of the number of minutes the output is delayed in respect to the input. The time delay of the stream buffer			
IP – Physical Connector	Selects eth0 or eth1, the standard Ethernet ports, eth2 and eth3 available if TSS 6220 has added optional ports	Entry selects which of the available Ethernet ports on the TSS 6220 that the transmit delay stream is assigned to output			
Destination IP	224.0.0.0 – 239.255.255.255	This address is the IP address the source device is sending to for a multi-cast. A unicast would use an IP address less than 224.0.0.0.			
Destination Port	0 – 65535	This is the UDP port the source device is sending to.			
Source IP Mode	Select Auto or Manual	Provides entry of how the TSS 6220 communicates a source IP address			
		Auto: Allows TSS 6220 to automatically select and communicate the source IP address			
		Manual: Provides user entry of the source address			
Source IP	Available for entry when Source IP Mode is set to Manual.	Provides entry of a Source IP address for the stream that is communicated by the TSS 6220			
Source Port	Click up or down arrows to increment value shown. Click in field and enter value. Range: 1030 to	Specifies a value for the source port associated with the stream.			
Sauraa MAC	Coloct Auto or	In Auto, the TCC C220 simulates a source device			
Mode	Manual	and creates and communicates a unique source MAC address for the stream. In Manual, a user Source MAC address can be entered.			
Source MAC	Available when Source Mac Mode Manual - enter MAC address	Provides entry of a MAC address you want to specify as the Source MAC for the stream communicated by the TSS 6220			
TS Packets Per IP Packet	Enter value 1 to 7, Default is 7	This setting determines the number of TS stream packets that are inserted into IP packets. 7 being the maximum and the typical setting. Lesser packets may be selected.			
Encapsulation	Select UDP or RTP	Selects IP encapsulation to UDP or RTP			

# 7 Disaster Recovery Viewing Panel

The Disaster Recovery Viewing Panel provides the capability to setup, control and monitor the Disaster Recovery capability offered by the TSS 6220. The Disaster Recovery feature receives and monitors a defined incoming transport stream. The input stream is buffered to memory storing a defined period of programming segments, typically a day or week. The Disaster Recovery feature serves as a programming backup. Upon the loss of the incoming stream, the stored memory is available to stream to the output.

The Disaster Recovery feature of the TSS 6220 is an optional licensed feature. The Disaster Recovery tab is only available when the feature is licensed. More than one license is available for purchase if multiple disaster recovery output capabilities are needed. Options for increasing the hardware, memory storage drive, and redundancy RAID configuration capabilities are available to accommodate different requirements.

The following sections in this chapter describe configuration and monitoring menus for receive, buffering and outputting of the TS streams in the Delay function.





# 7.1 Disaster Recovery Panel Overview

The TSS 6220 provides flexibility in setting up the receiver, buffer, and disaster recovery criteria Depending on licensing, more than one Disaster Recovery can be added and shown in the Disaster Recovery Panel. Each disaster recovery added has its unique user defined input, buffering, transmit, and disaster recovery management settings. Each disaster recovery added has its own section in the panel and is identified by a heading with incrementing identifying numbers Disaster Recovery 1, Disaster Recovery 2, etc.

For each Disaster Recovery in the panel, there is a row in the panel indicating the receive criteria, a row indicating buffer information and a row indicating Transmit or output information. The following diagram and descriptions provide an overview of the layout and fields common in the Disaster Recovery Panel. The remainder of this chapter provides additional details of configuration menus, informational fields in the panel, and dropdown information menus.

Play Record Schedule Delay	A Disacter Recovery Files	Admin Reporting	About		_
Disaster Recovery Control Panel C Add Disaster Recovery Disaster Recovery 1	3			D	E
🔅 Configure Disaster Recovery				Remove	
. ● eceive	Physical Connector: eth1	239.192.1.50:10	50	19 ps •	<u> </u>
😠 o 🙈 🚕 B ter	State: Buffering	➡ 00:0	9:13 Buffer Size: 1.3 GB		
■ • C t	Physical Connector: eth1	239.192.10.222	Originally Aired: GAP(No Output	ıt) 0.000 Mbps 🍥	<u></u> н

Item or	Field	<b>Button/Action</b>	Description
Α.	Select Disaster Recovery	Delay Click on this field to access the Disaster Recovery feature	Provides access to the Disaster Recovery feature and all configuration and monitoring menus. Disaster Recovery license required.
В.	Add Disaster Recovery	Add Click this icon to add a recovery to an output ethernet port	Provides access to menus to add and configure receive, buffer, and transmit criteria for the disaster recovery
C.	Configure Disaster Recovery	Click this icon to access control menus	Provides access to menus to setup/control receive, buffer and transmit criteria for the Disaster Recovery
D.	Delete Delay	Remove Click on this ito delete this delay	Removes the recovery from the panel, stops input receive, buffering and transmit of this disaster recovery.
E.	Show/Hide Disaster Recovery fields	Selectable, click on the icon	Collapses the receive, buffer, transmit rows to not be visible in the panel. When collapsed, it expands to show receive, buffer, and transmit rows
F.	Receive	See section: 7.2.1 for descriptions	This row shows the receiver information for the Disaster Recovery
G.	Buffer Section	See section 7.2.2 for descriptions	This row shows the buffer information for the Disaster Recovery
Н.	Transmit section	See next section 7.2.3 for field descriptions	This row shows the transmit output information of the Disaster Recovery



# 7.2 Disaster Recovery Panel Descriptions

The Disaster Recovery Panel contains three rows of information associated with an added Disaster Recovery. A top row is identified as receive information. A middle row identified as buffer information. A bottom row is identified as transmit information. This section provides descriptions of the information contained in these sections of the Disaster Recovery Panel. The next section in this chapter, Section 7.3, describes configuration settings which determine or impact the listed information described in the panel.

#### 7.2.1 Disaster Recovery Panel – Receive Descriptions

The Disaster Recovery Panel contains a row of information that is identified by a "Receive" indication. This row of information summarizes the configuration of the receiver for the added disaster recovery. The information in the fields indicate user settings that were specified when the disaster recovery was added or created. See section 7.3.1 and 7.3.2 in this manual for setup of the disaster recovery receive. The fields also indicate current status on the input stream and bitrate. The following information provides a description of each field in the Receive row.

Disaster Recover	y 1			0
🔅 Configure Disas	ster Recovery			Remove
	Receive	Physical Connector: eth 1	239.192.1.50.1050	19.393 Mbps

Receive row field descriptions:

Item or Field		<b>Button/Action</b>	Description			
A. Adds Status Menu to Delay Panel		Click icon to add dropdown menu	Adds a dropdown section to the Receive row showing additional Status information. See section 7.5.1.			
В.	Receive row	Not a selectable field	Identifies row as information relative to receive and input stream of the Delay			
C.	Physical Input Ethernet Port	Not a selectable field	Indicates the user defined receive or destination input IP address and port. Or, indicates if an ASI input port is selected.			
D.	Input Destination Address/Port	Not a selectable field	Indicates the user defined receive or destination input IP address and port When ASI input is selected, this field indicates Sync Status (Locked, Unlocked)			
E.	Input Bitrate	Not a selectable field	Indicates the incoming TS stream bitrate			
F.	Status	Status Light, not	Red: No input stream, abnormal input or condition			
	Indicator	a selectable field	Green: Input active/enabled - input normal			
			Gray: Input inactive/disabled – input not setup			



#### 7.2.2 Disaster Recovery Panel – Buffer Descriptions

The Disaster Recovery Panel contains a row of information that is identified by a "Buffer" indication. This row of information summarizes the configuration of the Buffer for the created delay and disaster recovery mode. The information in the fields indicate user settings that were specified when the Disaster Recovery was added or created. See section 7.3.1 for setup of the Disaster Recovery buffer. The row further indicates status information in regard to buffering the input stream. The following information provides a description of each field in the Buffer row.

Disaster Recovery 1									
🎲 Configure Disaster Recovery					🤤 Rem	ove			
Receive	Physical Connector: eth1	239.192	.1.50:1050		19.393 Mbps	۲			
🗉 o 🛕 👍 Buffer	State: Buffering	<b></b>	00:09:13	Buffer Size: 1.3 GB		۲			
ABCD	E	F	G	Н					

Buffer row field descriptions:

tem or	Field	<b>Button/Action</b>	Description			
Α.	Adds Configuration Menu to Delay Panel	Click icon to add dropdown menu	Adds a dropdown section to the Buffer row showing additional buffer configuration information. See section 7.5.2 for details			
В.	Start Disaster Recovery Output	Click on 🍐 icon to Start Disaster Recovery mode	Starts Disaster Recovery output mode, outputs buffered delayed stream, stops recording,			
C.	Stop Disaster Recovery Output	Click on 🍐 icon to exit the Disaster Recovery mode	Stops Disaster Recovery output mode, resumes input recording to buffer			
D.	Buffer Label	Not a selectable field	Identifies row as information relative to receive buffer			
E.	Buffer state	Not a selectable field	Indicates the current state of the buffer			
			Buffering: Building a buffer to specified duration			
			Normal Operation: - actively buffering delayed output			
			Recovery: buffering disaster recovery output			
F.	Extract File Menu	Glick on icon, provides an Extract Buffer to File menu	Provides a menu in which the time referenced stream data in the buffer memory can be named and extracted to a file			
G.	Buffer Time Delay	Not a selectable field	Indicates when buffer is filling as moving blue filler, indicates time of data capture or delay time of buffer			
Н.	Buffer Size	Not a selectable field	Indicates the buffer size as drive storage space being utilized			
I.	Status	Status Light, not	Red: No output stream, abnormal output or condition			
	mulcator	a selectable field	Green: Input active/enabled – output normal			
			Gray: Input inactive/disabled – output not setup			



### 7.2.3 Disaster Recovery Panel – Transmit Descriptions

The Disaster Recovery Panel contains a row of information that is identified by a "Transmit" or "ASI Transmit" indication. This row of information summarizes the configuration of the delayed transmit or output. The information in the fields indicate user settings that were specified when the Disaster Recovery was created. See section 7.3.3 in this manual for setup of the Disaster Recovery transmit criteria. The fields also indicate current status of the delayed output or transmit. The following information provides a description of each field in the Transmit row.

Disaster Recovery 1										
🔅 Configure Dis	Configure Disaster Recovery									
H 0	Receive	Physical Connector: eth1	239.1	92.1.50:1050		19.393 Mbp	is 🔘			
H • 🔺	👍 Buffer	State: Buffering		00:09:13	Buffer Size: 1.3 GB		۲			
	Transmit	Physical Connector: eth1	239.1	92.10.222:10009	Originally Aired: GAP(No Output)	0.000 Mbp	)s 🔘			
A	В	C		D	E	F	G			

Transmit row field descriptions:

Item or	r Field Name	<b>Button/Action</b>	Description
Α.	Status & Configuration	Click on this icon	Provides an added window showing Delay output status and configuration information. See section 7.5.3 for details.
В.	Identifies Transmit section	Not selectable, No action	Identifies the row as related to the delay's output
C.	Physical Connector/port	Not Selectable	Indicates the Physical Ethernet port or ASI port which the delay stream in the listed schedule will playout
D.	IP Address/Port	Not Selectable	Indicates the destination IP address and port of delay output. If ASI output port is used this field indicates Null Stuffing status (Enabled or Disabled)
E.	Original Time	Not Selectable	Provides reference of the original date and time that the delay is referenced.
F.	Total Port Bitrate Indication	Not selectable, view only	Indicates Ethernet port bitrate of the delayed output TS streams
G.	Status Indicator	Not Selectable	Indicates status of output: Gray: Inactive – output stopped, Green: Good active playout, Red: Playout fault or no output



# 7.3 Adding & Configuring a Disaster Recovery

To add a Disaster Recovery which provides a new delay input to output line or path and disaster recovery mode in the event of an incoming stream failure/disaster, requires that you define receive, buffer and ouput parameters. To create a Disaster Recovery listing, click on the Add Disaster Recovery icon. This icon is located just below the Disaster Recovery Panel heading. If you do not see this section, click on the Expand Panel icon located at the right. Note that this selection is available for each of the Disaster Recovery Control Panels that are available by licensing.

To add a delay requires that you configure general, receive, and output parameters. These parameters are found in General, Receive, and Transmit menus that are available upon clicking the Add Disaster Recovery icon. The following sections describe these menus and their configuration. See sections 7.3.4, 7.3.5, and 7.3.6 for information on menus and configuration when adding a Disaster Recovery using the optional ASI input/output hardware.

F	Play	Record	Schedule	Delay	Disaster Recovery	Files	Admin	Reporting	About			
D	Disaster Recovery Control Panel											
	Add 🕻	Disaster Re	covery									

#### 7.3.1 Adding a Disaster Recovery - General Menu

When clicking on the Add Disaster Recovery icon, the Add Disaster Recovery menu is shown. There is a General, Receive, and Transmit menu which is selected by clicking on the available tabs. The General configuration menu is shown by default. This menu provides alias naming, recovery buffer and triggering criteria, and definition of the Gap Mode.

Click on the Alias field and enter an alias name, if desired. This is not required as the application automatically assigns a name as an incrementing Disaster Recovery number

The Recovery Buffer (days) field permits entry, or selection with the up and down arrow increments, of the days in the buffer. This is the duration of the stored or buffered input TS stream data. It also is the delay, in days specified, between the incoming TS stream data and the delayed output.

The Disaster Trigger Mode is set to either manual or automatic. The manual setting requires a user selection to start a Disaster Recovery mode or output. An auto setting directs the TSS 6220 to automatically start a

Add Disastery Recovery			
General Receive T	ransmit		
Alias:	Di	isaster Recov	ery 2
Recovery Buffer(days):	14	l .	÷
Disaster Trigger Mode:	М	anual	~
Trigger Delay(minutes):	0		\$
Gap Mode:	N	o Output	~
Source:	N		Browse
		Apply	Cancel

Disaster Recovery mode output after the input TS stream is lost and after the specified user Trigger Delay (minutes) has elapsed. When the Disaster Trigger Mode is set to Auto, the Trigger Delay (minutes) field becomes available for entry or for selection of minutes by using the up and down arrow fields. The entry determines the time (minutes) in which the Disaster Trigger output mode waits for the incoming TS stream to return prior to implementing a Disaster Recovery mode output. If the incoming TS stream, remains missing for the duration of the minutes specified in the Trigger Delay (minutes) field, the Disaster Recovery mode is implemented.

The Gap Mode importantly defines what the output is when there is no output transport stream to output from the delay buffer. This occurs when the buffer is buffering and has not yet reached the



specified delay time. This condition also occurs upon ending an active Disaster Recovery mode or output, as the buffer would need to catch up to the specified Recovery Buffer Days. A gap mode also occurs if buffer TS stream data is bad because of an input stream issue when recorded into the buffer.

The Gap Mode user selections include: No Output, Null packets, Default File, and Live. The No Output setting would provide no output TS stream when a gap occurs. The Null Packets setting would fill the output ts stream with Null Packets when a gap occurs. A Live setting would route the input ts stream directly to the output bypassing the delay buffer when a gap occurs. When set to Default File, the Source field defines the TS transport stream that plays out during a gap. Click on the Browse tab and choose the stream to be the default file.

Setting	Range/Selections	Description
Stream – Alias	Value and or Letter Entry	Provides entry of alias to identify a delay. If no alias is entered, the TSS 6220 assigns a delay number.
Recovery Buffer (days)	Enter value or click on up and down arrow fields to increment value	Provides entry of days. Defines the duration, in days, of the buffered memory size/duration of input TS stream data. Duration also is the delay between incoming TS stream and output when not in a Disaster Recovery mode.
Disaster Trigger Select to Manual or		Manual:
Mode	Auto	Auto:
Trigger Delay (minutes)	Enter value or click on up and down arrow fields to increment value	Provides entry of minutes. In Auto Disaster Trigger Mode - Defines the wait time (minutes) between when the input TS stream is lost (start of disaster) and when the output is replaced by the buffered TS stream data (Disaster Recovery Output started)
GAP Mode	Click Dropdown – select from listed	Determines the output when a gap exists and there is no TS stream from the buffer. Selections include: Default File: Outputs choose TS file No Output: Outputs no output stream Null Packets: Outputs TS stream with null packets Live: Routes the input TS stream directly to output
Source	Not selectable	Indicates "No Source" when no file has been selected. Indicates the selected file name when a file is selected.



#### 7.3.2 Adding a Disaster Recovery – Receive Menu

The Add Disaster Recovery Receive menu provides configuration of the IP input used to receive the MPEG-IP unicast or multicast. The selected input TS stream is routed to the delay buffer. This section provides descriptions of the settings in this menu

Configure the IP input to be active by setting the Receive field to "Enabled." Select the Physical Connector or Ethernet port on the TSS 6220 to use as the receive port. Eth 0 and Eth 1 are available. If the added ethernet port hardware option is added to your TSS 6220, then eth 2 and eth 3 will be available in the dropdown.

Select the Mode of the receiver to be Unicast or Multicast. For unicast, specify the destination port in the Destination IP field. For Multicast, specify the Destination IP address, and Destination IP Port.

The IP configuration section further includes settings to provide IGMPv3 features. An IGMP filter may be implemented for use to specify the inclusion or exclusion

Add Disastery Recovery		
General Receive Transm	it	
Receive:	Enabled	v
Physical Connector:	eth1	~
Mode:	Multicast	~
Destination IP:	239.192.0.200	
Destination Port:	10000	* *
IGMP Filter Mode:	Exclude	-
Add IGMP Address	🥥 Rer	nove All
IGMP Address		Remove
	Apply	Cancel

of source addresses. The TSS 6220 is IGMPv3 compliant. IGMPv3 allows each steam to be seen by the network as relating to a unique source device with a unique IP address, port, and/or MAC address. IGMPv2 is used to join/leave multicast streams by default if no IGMP Filter addresses are entered in the Add IGMP Address section of the menu. If IGMP Filter Mode addresses are specified then IGMPv3 is automatically used.

Settings	Range	Description
IP Receive	Enable Disabled	This setting allows the user to enable or disable these input stream settings.
IP – Physical Connector	Selects eth0 or eth1, the standard Ethernet ports, eth2 and eth3 available if TSS 6220 has added optional ports	Entry selects which of the available Ethernet ports on the TSS 6220 that the stream input is assigned
IP Mode	Multicast Unicast	<i>Multicast</i> setting allows the unit to receive multicast streams. Multicast streams originate from the IP range 224.0.0.0 – 239.255.255.255. <i>Unicast</i> allows the unit to receive unicast streams. Unicast streams originate directly from a source device.
Destination IP	Enter Value: 224.0.0.0 - 239.255.255.255	This address is the IP address the source device is sending to for a multi-cast. A unicast would use an IP address less than 224.0.0.0. This setting is only available when receiving a multicast stream. This address is the IP address the source device is sending to.



Destination Port	Enter Value: 0 – 65535	This is the UDP port the source device is sending to. This is the only setting required to receive a unicast stream.
IGMP Filter Mode	Settings: Select Include or Exclude	Used on networks supporting IGMPv3. If this setting is set to <i>Exclude</i> any streams originating from the user defined IP addresses will be rejected. If this setting is set to <i>Include</i> any streams originating from the user defined IP addresses will be received.
Add IGMP Address	Click in field - Enter IP address to include or exclude as per filter mode: Values: 0.0.0.0 - 255.255.255.255	Enter and list IP address of IGMPv3 to include or exclude as a filter setting.
Remove All	Click on icon	Removes or clears all the listed IGMPv3 address



### 7.3.3 Adding a Disaster Recovery – Transmit Menu

The Add Disaster Recovery menu provides configuration of the Disaster Recovery delay output. Click on the Transmit tab to configure the output criteria of the Disaster Recovery output. This section describes the selections in the Transmit configuration menu.

Select the Physical Connector or Ethernet port on the TSS 6220 to use as the receive port. Eth 0 and Eth 1 are available. If the added ethernet port hardware option is added to your TSS 6220, then eth 2 and eth 3 will be available in the dropdown. Note that the receive and transmit ports for the delay may be the same or different ports.

Select and enter the Destination IP and port values. For unicast, specify the destination port in the Destination IP field. For Multicast, specify the Destination IP address, and Destination IP Port.

The Transmit tab includes settings to define the TSS 6220 as a specific source device for IGMPv3. This allows each steam to be seen by the network as a unique source device with a unique IP address, and/or source port, and/or MAC address. This section provides descriptions of the settings.

Add Disastery Recovery		
General Receive Transm	it	
Physical Connector:	eth1	*
Destination IP:	239.192.0.200	
Destination Port:	10000	\$
Source IP Mode:	Auto	-
Source IP:	0.0.0.0	
Source Port:	3020	-
Source MAC Mode:	Auto	*
Source MAC:	00:00:00:00:00:00	
TS Packets Per IP Packet:	7	*
Encapsulation:	UDP	*
	Apply Cance	el

Setting	Range	Description
IP – Physical Connector	Selects eth0 or eth1, the standard Ethernet ports,	Entry selects which of the available Ethernet ports on the TSS 6220 that the transmit delay stream is assigned to output
	eth2 and eth3 available if TSS 6220 has added optional ports	
Destination IP	224.0.0.0 – 239.255.255.255	This address is the IP address the source device is sending to for a multi-cast. A unicast would use an IP address less than 224.0.0.0.
Destination Port	0 – 65535	This is the UDP port the source device is sending to.
Source IP Mode	Select Auto or Manual	Provides entry of how the TSS 6220 communicates a source IP address
		Auto: Allows TSS 6220 to automatically select and communicate the source IP address
		Manual: Provides user entry of the source address
Source IP	Available for entry when Source IP Mode is set to Manual.	Provides entry of a Source IP address for the stream that is communicated by the TSS 6220



Source Port	Click up or down arrows to increment value shown. Click in field and enter value.	Specifies a value for the source port associated with the stream.
	Range: 1030 to 65535	
Source MAC Mode	Select Auto or Manual	In Auto, the TSS 6220 simulates a source device and creates and communicates a unique source MAC address for the stream. In Manual, a user Source MAC address can be entered.
Source MAC	Available when Source Mac Mode Manual - enter MAC address	Provides entry of a MAC address you want to specify as the Source MAC for the stream communicated by the TSS 6220
TS Packets Per IP Packet	Enter value 1 to 7, Default is 7	This setting determines the number of TS stream packets that are inserted into IP packets. 7 being the maximum and the typical setting. Lesser packets may be selected.
Encapsulation	Select UDP or RTP	Selects IP encapsulation to UDP or RTP

When finished with all the selections in the General, Receive, and Transmit menus, click on the Apply box at the bottom to apply changes and create the Disaster Recovery. The Disaster Recovery is added to the Disaster Recovery Panel. You can access the configuration menus for changes by clicking on the Configure Disaster Recovery





### 7.3.4 Adding a Disaster Recovery – General Menu – ASI In/Out

The following sections describe the Disaster Recovery menus and configuration when the TSS 6220 is equipped with the optional ASI input/output hardware. Click on the Add Disaster Recovery icon to create a Disaster Recovery. This opens the Disaster Recovery configuration menus. There is a General, Receive, and Transmit menu which is selected by clicking on the available tabs. The General configuration menu is shown by default. This menu provides alias naming, recovery buffer and triggering criteria, and definition of the Gap Mode.

Click on the Alias field and enter an alias name, if desired. This is not required as the application automatically assigns a name as an incrementing Disaster Recovery number

The Recovery Buffer (days) field permits entry, or selection with the up and down arrow increments, of the days in the buffer. This is the duration of the stored or buffered input TS stream data. It also is the delay, in days specified, between the incoming TS stream data and the delayed output.

The Disaster Trigger Mode is set to either manual or automatic. The manual setting requires a user selection to start a Disaster Recovery mode or output. An auto setting directs the TSS 6220 to automatically start a Disaster

Add Disastery Recovery			
General Receive	Transmit		
Alias:	Γ	Disaster Recov	very 2
Recovery Buffer(days):		14	\$
Disaster Trigger Mode:		Auto	
Trigger Delay(minutes):		0	\$
Gap Mode:		No Output	-
Source:			
			Browse
		Apply	Cancel

Recovery mode output after the input TS stream is lost and after the specified user Trigger Delay (minutes) has elapsed. When the Disaster Trigger Mode is set to Auto, the Trigger Delay (minutes) field becomes available for entry or for selection of minutes by using the up and down arrow fields. The entry determines the time (minutes) in which the Disaster Trigger output mode waits for the incoming TS stream to return prior to implementing a Disaster Recovery mode output. If the incoming TS stream, remains missing for the duration of the minutes specified in the Trigger Delay (minutes) field, the Disaster Recovery mode is implemented.

The Gap Mode importantly defines what the output is when there is no output transport stream to output from the delay buffer. This occurs when the buffer is buffering and has not yet reached the specified delay time. This condition also occurs upon ending an active Disaster Recovery mode or output, as the buffer would need to catch up to the specified Recovery Buffer Days. A gap mode also occurs if buffer TS stream data is bad because of an input stream issue when recorded into the buffer.

The Gap Mode user selections include: No Output, Null packets, Default File, and Live. The No Output setting would provide no output TS stream when a gap occurs. The Null Packets setting would fill the output ts stream with Null Packets when a gap occurs. A Live setting would route the input ts stream directly to the output bypassing the delay buffer when a gap occurs. When set to Default File, the Source field defines the TS transport stream that plays out during a gap. Click on the Browse tab and choose the stream to be the default file.

Setting	Range/Selections	Description
Stream – Alias	Value and or Letter Entry	Provides entry of alias to identify a delay. If no alias is entered, the TSS 6220 assigns a delay number.
Recovery Buffer (days)	Enter value or click on up and down	Provides entry of days. Defines the duration, in days, of the buffered memory size/duration of input



	arrow fields to increment value	TS stream data. Duration also is the delay between incoming TS stream and output when not in a Disaster Recovery mode.
Disaster Trigger	Select to Manual or	Manual:
wode	Auto	Auto:
Trigger Delay (minutes)	Enter value or click on up and down arrow fields to increment value	Provides entry of minutes. In Auto Disaster Trigger Mode - Defines the wait time (minutes) between when the input TS stream is lost (start of disaster) and when the output is replaced by the buffered TS stream data (Disaster Recovery Output started)
GAP Mode	Click Dropdown – select from listed	Determines the output when a gap exists and there is no TS stream from the buffer. Selections include: Default File: Outputs choose TS file No Output: Outputs no output stream Null Packets: Outputs TS stream with null packets Live: Routes the input TS stream directly to output
Source	Not selectable	Indicates "No Source" when no file has been selected. Indicates the selected file name when a file is selected.



### 7.3.5 Adding a Disaster Recovery – Receive Menu – ASI In/Out

The Add Disaster Recovery Receive menu provides selection and configuration of the ASI input used to receive the incoming stream. When the TSS 6220 contains the optional ASI input/output hardware, a Receive Type field is included in the Receive Menu. This field configures the Disaster Recovery input as ASI or MPEG-IP. In the Receive Type field click on the drop down arrow and select ASI. This configures the TSS 6220 to receive and rout an ASI input TS stream for the created Disaster Recovery. It further modifies the Receive menu for ASI input configuration.

When the Receive Type is ASI, the Receive menu provides selection for enabling or disabling the input port. It further provides selection of one of the available ASI hardware ports. Set the Receive field to "Enabled" to start receiving and



buffering the incoming stream. Click on the dropdown in the Receive field and select "Enabled" to make the selected ASI port active.

The optional ASI hardware contains 4 ASI ports. These ports can be configured by the TSS 6220 as an input or as an output port. The ASI ports are available for use by licensed features of the TSS 6220 including the Play, Record, Delay, and Disaster Recovery features. An ASI port cannot be shared by these features. Once a port is assigned or in use by a feature, the port becomes unavailable (greyed out) for the other features. Click the drop down arrow in the Port field to view available input ports when adding a Disaster Recovery.

Setting	Range/Selections	Description
Receive Type	ASI or MPEG-IP select, click field dropdown arrow and select	Provides selection of an input port for the Disaster Recover function. ASI: selects ASI option card MPEG-IP: select MPEG-IP port
Receive	Enable or Disabled select, click dropdown arrow and select	Enables or Disables the receive ASI input and selected port
Port	Selects ASI input Port 1, 2, 3, or 4, click dropdown and select from list of available ports	Selects from available ASI ports. Ports listed are available for use and not used by other TSS 6220 features (Play, Record, Delay). The ASI hardware contains 4 ASI ports. These ports can be either used as inputs or output ports.

ASI Receive Menu Descriptions



### 7.3.6 Adding a Disaster Recovery – Transmit Menu – ASI In/Out

The Add Disaster Recovery menu provides configuration of the Disaster Recovery delay output. Click on the Transmit tab to configure the output criteria of the Disaster Recovery output. This section describes the selections in the Transmit configuration menu when an ASI type is used.

A Transmit Type field is included in the Transmit Menu. This field configures the Disaster Recovery output as ASI or MPEG-IP. In the Transmit Type field click on the drop down arrow and select ASI. This configures the TSS 6220 to output the Disaster Recovery ts stream to a selected ASI port. When selected, it modifies the Transmit menu for ASI output configuration.

When the Transmit Type is ASI, the menu provides selection for the physical ASI port. The optional ASI hardware contains 4 ASI ports. These ports can be configured by the TSS 6220 as an input or as an output port. The ASI ports are available for use by licensed features of the TSS 6220 including the Play, Record, Delay, and Disaster Recovery features. An ASI port cannot be shared by these features. Once a port is

Add Disastery Recovery			
General	Receive Tr	ransmit	
Transmit T	ype:	ASI	+
Port:		ASI Port 1	
Null Stuffin	g:	Disabled	Ŧ
Total Bitrate (Mbps):		0	\$
		Apply	Cancel

assigned or in use by a feature, the port becomes unavailable (greyed out) for the other features. Click the drop down arrow in the Port field to view available output ASI ports. When configuring an ASI out, select from the list of available ports.

The Null Stuffing field provides an option to add null stuffing bytes to the Disaster Recovery output to increase the output bitrate. Click on the field and enable the features. The Total Bitrate field below becomes available to enter a desired bit rate. Enter the desired bit rate, in Mbps, in the Total Bitrate field.

Setting	Range/Selections	Description
Transmit Type	ASI or MPEG-IP, click field dropdown arrow and select	Provides selection of an output port for the Disaster Recover function. ASI: selects ASI option card MPEG-IP: select MPEG-IP port
Port	Selects ASI input Port 1, 2, 3, or 4, click dropdown and select from list of available ports	Selects from available ASI ports when Transmit Type is ASI. ASI Ports listed are available for use and not in use by other features (Play, Record, Delay). The ASI hardware contains 4 ASI ports. These ports can be either used as inputs or output ports. Once in use they are unavailable and are
Null Stuffing	Select and choose either disabled (default) or enabled.	Disabled: The ASI output bitrate is determined automatically by the TSS 6220. The Total Bitrate (Mbps) field below in the menu is grayed out.
		Enabled: The TSS 6220 adds null stuffing to increase the playout bitrate of the ASI output stream. The user enters the desired playout bitrate in the Total Bitrate (Mbps) field in the menu.
Total bitrate	Available when Null Stuffing is enabled. Click on the field and enter a value in Mbps	Enter a value in Mbps of the desired ASI output bit rate. The TSS 6220 adds null bytes to the stream to increase the total bit rate to the entered value in Mbps.



# 7.4 Delay Buffer – Extract Buffer To File

The TSS 6220's Disaster Recovery feature has an active buffer receiving TS input data stream and outputting delayed ts stream data. The buffer contains an accumulation of TS captured data in a proprietary format that extends in duration slightly beyond the specified recovery/delay time(s). The incoming buffered TS data is time stamped according to its arrival time by the system clock. The TSS 6220 offers the ability to specify a time relative to the incoming time stamps and extract a duration of the ts captured data from the buffer to a memory file. The extracted transport stream data can be written to a file in a common ts or trp format.

In the Delay's buffer section of the web GUI exists an Extract Buffer to File icon  $\blacksquare$ . It is located near the middle of the page in the Buffer section/row. Click on this icon to access the Extract Buffer to File menu. This section describes the Extract Buffer to File menu and how to extract the time stamped buffer data.



The Extract Disaster Recovery Buffer To File Menu includes selection of a Start Extract Date and Start Extract Time. This date & time selection permits a user to specify a day and time of the original arrival time of the TS input data to the buffer. The selections must be a time which matches time stamps of data currently in the buffer. Click on the Start Extract Date field and enter a date. Or, click on the calendar icon and select a day. Click in the Start Extract Time field and edit or enter the extract time.



The data extraction requires a defined duration which is determined by an entry in the Extract Duration (minutes) field. Click in the field and enter a value, in minutes. Or click on the up and/or down arrows at the right of the field. The duration must not exceed time stamps currently in the buffer. Reference the time indicated by the Original Time field in the Transmit section and the Delay duration setting to determine applicable start and duration entries for extraction compatibility.

Extracted data is written to a file selected by clicking on the Browse icon and selecting a current file. Or, click on Browse and enter a new file name in the bottom field of the Browse Files menu.

Finally, click on the Extract To File icon at the bottom center of the menu to start the data extraction. A progress bar will indicate the process is being implemented. If time and duration entries are not found in the time

Buffer Extraction Success	⊗
The buffer segment was successfully extracted to Extract_File_6-25-2018	.trp
ок	

stamped buffer data, a message will indicated. Upon conclusion of the extraction, a Buffer Extraction Success message appears. Click the OK field to acknowledge the extraction as completed.

Extract Disaster Recovery Buffer To File descriptions:



Settings	Range	Description
Start Extract Date	Click on calendar icon select date, or enter date infield YYYY,MM,DD	This setting (date) marks the day in which the buffer data is extracted from memory and written to a file. Note: The selected date is typically today's date unless the delay is > 24 Hrs. The day must be in current buffer data time stamps.
Start Extract Time	Click in field, Enter time	Entry references time stamp (hours, seconds) of data in the buffer in which the data extraction starts. Note: The time reference must be within current buffer.
Extract Duration (minutes)	Time	Field indicating the time duration of the data following the start extract time that is to be extracted and written to the file.
Extract File	Click Browse field to go to Browse File menu. Select an existing file to overwrite. Or enter a file name at the bottom of the menu.	This is file name that the buffer data is extracted to. If "No Source" is shown, you need to select an existing file to replace or create a new file.
Browse	Click on Browse field	Takes you to the Browse Files menu. Enter a file for selection or entry of a file name, once a file is named or selected the file name is indicated in the Extract File field
Extract to File	Click on Extract To File field to apply entries	Extracts the buffer data starting at the date/time referenced for the duration specified to the selected file name. Convert data to the file type specified.



# 7.5 Disaster Recovery – Added Panel Status & Configuration Menus

Disa	ster Recovery 1	1					
ും c	onfigure Disaste	r Recovery				🤤 Ren	nove
•		Receive	Physical Connector: eth1	239.192.1.50:1050		19.393 Mbps	۲
	• 🔺 🔺	Buffer	State: Buffering	00:09:13	Buffer Size: 1.3 GB		۲
•	þ	Transmit	Physical Connector: eth1	239.192.10.222:10009	Originally Aired: GAP(No Output)	0.000 Mbps	۲

When each of the windows are expanded complete status and configuration information is shown for receive, buffer, and transmit sections. Click on the  $\exists$  icon at the left of each row to collapse the added status and configuration panel windows.

Disaster Recovery 1					٢
🔅 Configure Disaster Recovery				🤤 Remo	ove
Status Sync Status: Locked Packets Per Frame: 7 Encapsulation: RTP	Physical Connector: eth1 Configuration Mode: Multicast IGMP Mode: Exclude	239.192.1.50.1050	ort	19.392 Mbps	۲
Configuration Recovery Buffer Recovery Buffer(days): 1 Disaster Trigger Mode: Manual Trigger Delay(minutes): 15 Gap Mode: No Output Gap File: NO SOURCE	State: Normal Operation	a 1 Days 00:00:00	Buffer Size: 398.5 GB		۲
Source IP: 10.0.0.66 Source MAC: 0C:C4:7A:C9:AB:41 Mode: Multicast Receiver MAC: N/A	Physical Connector: eth1 Configuration Source IP Mode: Auto Source Port: 3020 Source MAC Mode: Auto TS Packets: 7	239.192.10.222.10009	Originally Aired: 2018-07-12 11:06:11	19.373 Mbps	۲



#### 7.5.1 Disaster Recovery Receive – Status Information Menu

Each of the Disaster Recovery listings in the Panel includes in the Receive row an added Status and Configuration window. To access this window and add it to the panel, click on the <sup>III</sup> icon at the left of the Receive row. The Status and Configuration boxes are added to the Play Control Panel providing status and configuration details. Click on the <sup>III</sup> icon at the same location to hide the status and configuration information. Following is a description of the fields and information found in this Status and Configuration window.

Disaster Recovery 1				
🌼 Configure Disaster Recove	Configure Disaster Recovery			
Status	ive Pi	hysical Connector: eth1 - Configuration	239.192.1.50:1050	
Sync Status: Locked Packets Per Frame: 7 Encapsulation: RTP		Mode: Multicast IGMP Mode: Exclude	IGMP Filter List	port

The Sync Status, Packets Per Frame and Encapsulation fields provide information regarding the incoming receive transport stream. Should the Sync Status indicate "Unlocked" check the IP configuration settings and/or availability of the MPEG-IP stream.

Status Listing	Description
Sync Status	Indicates the source TS stream is being received and TS sync is established. Locked: Indicates receiving and locked to TS stream Sync
Packets Per Frame:	Indicates the TS packets per TS frame in the incoming TS stream
Encapsulation:	Indicates receive IP stream encapsulation, RTP, UDP

The Configuration part of the window provides information regarding the settings for the Disaster Recovery input. The Configuration window also indicates the IGMP receive mode, unicast or multicast, along with IGMPv3 settings and filter addresses.

Configuration Listing	Description
Mode:	Indicates if input receive is configured as Unicast or Multicast
IGMP Mode:	Indicates Include or Exclude mode for IGMPv3 address entry
Add IGMP Filter	Indicates listed IGMP filter addresses to Exclude or Include for Source Specific IGMPv3



(

### 7.5.2 Disaster Recovery Buffer – Configuration Menu

In the Buffer section of each of Disaster Recovery Panel listings there is an added Configuration information window available. To access this window and add it to the panel, click on the <sup>III</sup> icon at the left of the Buffer section in the panel. The Configuration window is added to the Delay panel. Click on the <sup>III</sup> icon at the same location to hide the status and configuration information. Below is a description of the information you will find in this information window.



The Configuration window includes the selected Recovery Buffer delay. This is the buffer size (delay) in days. It window indicates the selected Disaster Trigger Mode, either Manual or Automatic. If set to Auto mode, the selected delay time before the Disaster Recovery Mode is triggered is shown. The Configuration window also includes information regarding the Gap Mode and Gap File. It further includes the Maximum Buffer Duration listed in minutes. Recall that the Gap Mode permits selection of what you want to be outputting when there is no TS stream available to output from the buffer. A transport stream file may be selected to output during a gap condition by setting the Gap Mode to Default File. A selected file for this is indicated in the Gap File field of the Configuration window.

Configuration Listing	Description
Recovery Buffer (days)	Length of the buffer recording, in days
Disaster Trigger Mode	Manual: Disaster Recovery mode or playout is set to be activated manually by a user clicking on the $\clubsuit$ icon
	Automatic: Disaster Recovery mode or playout is set to be activated automatically when the input TS stream is lost for the time specified by theTrigger Delay (minutes) entry
Trigger Delay (minutes)	In Auto Disaster Trigger Mode - The delay in minutes after the incoming TS stream is lost before the Disaster Recovery mode or playout is activated
Gap Mode:	Indicates the Gap mode:
	Default File: Outputs selected TS file No Output: Outputs no output stream Null Packets: Outputs TS stream with null packets Live: Routes the input TS stream directly to the Delay output
Gap File:	Indicates the ts file in which to play out when the Gap Mode is set to Default File and there is no TS data in the Buffer to output.



#### 7.5.3 Disaster Recovery Transmit – Status & Configuration Information Menu

An added Status and Configuration window is available in the Transmit section of each of the Disaster Recovery listings in the Panel. To access this window and add it to the panel for viewing, click on the <sup>III</sup> icon at the left of the Transmit row. The Status and Configuration boxes are added to the Panel providing status and configuration details. Click on the <sup>III</sup> icon at the same location to hide the status and configuration information. This section of the manual provides a description of the fields and information found in this Status and Configuration window.

Source IP Mode: Auto	
Source Port: 3020	
Source MAC Mode: Auto	
TS Packets: 7	
	Source IP Mode:AutoSource Port:3020Source MAC Mode:AutoTS Packets:7

The Status part of the window provides information regarding the Disaster Recovery output status. It includes an indication of its source IP address and its MAC address. It indicates if the output is in a unicast or multicast configuration status. It indicates a MAC address from a destination receiver if applicable.

Status Listing	Description
Source IP	Indicates the Source IP address
Source MAC	Indicates the Source MAC address,
Mode	Indicates Transmit IP mode, either Unicast or Multicast
Receiver MAC	Indicates a MAC address as indicated by a destination receiver

The Configuration part of the window provides information regarding the Disaster Recovery's transmit user settings selected in the TX Disaster Recovery configuration menu. See section 7.3.3 in this manual.

Configuration Listing	Description
Source IP Mode:	Auto or Manual
Source Port	Indicates the port value selected
Source MAC Mode	Indicates Automatic or Manual as selected in the TX Disaster Recovery setup menu.
TS Packets	Indicates the number of TS packets per IP packet as selected by the user in the TX Disaster Recovery setup menu



# 7.6 Understanding Disaster Recovery

Disaster Recovery provides a time delayed output in the event that a disaster occurs and the live input content is lost. The following provides more detail overviewing the Disaster Recovery feature to aid in understanding of how it works and how the panel indicators provide status information.

#### **Disaster Recovery Explained: Buffering**

The Disaster Recovery feature receives an input TS that is user specified. The input TS is routed to a record memory buffer. The size of the memory buffer is determined by a user entry of the buffer delay time in days. This is the Recovery Buffer (days) setting in the General menu. The recorded TS data is timed stamped with date/time information when recorded. After the buffer is initially filled with the specified day(s) of input TS stream data, it is maintained at this size. The buffer is continuously maintained in small segments. For example, after a new 5 minute segment is added to the buffer, the oldest 5 minute segment is deleted, keeping the buffer continuously updated and at the specified delay size (days).

In the process or adding a Disaster Recovery, after the Disaster Recovery is initially added and configured, the incoming TS stream begins to fill the record buffer, when this is occurring you see the time bar in the Buffer row filling and time incrementing toward the specified buffer time. While the buffer is filling there is no buffer data with the correct time stamps to correctly delay to the output. During this time the gap mode determines what if anything is output. The Transmit row shows the output criteria determined by the gap mode. For information on the gap mode see Section 7.3.1.

Disa	as	ter I	Recovery 1	l .					۵
<u>ې</u> ه و	Co	nfig	ure Disaste	Recovery				😂 Re	nove
Ŧ		•		Receive	Physical Connector: eth1	239.192.1.50:1050		19.393 Mbps	۲
Ŧ		•	🔺 🔺	Buffer	State: Buffering	00:09:13	Buffer Size: 1.3 GB		۲
Ŧ		0		Transmit	Physical Connector: eth1	239.192.10.222:10009	Originally Aired: GAP(No Output)	0.000 Mbps	

A full buffer is indicated by the blue gauge fully filled and the indicated time matching the buffer time delay setting. The Buffer size field indicates the memory size required determined by the incoming TS bit rate and buffer delay time.

±	•	🙈 👍 Buffer	State: Normal Operation	1 Days 00:00:00	Buffer Size: 398.6 GB		۲
ŧ	0	Transmit	Physical Connector: eth1	239.192.10.222:10009	Originally Aired: 2018-07-12 14:53:20	19.414 Mbps	

Once the buffer is filled with input TS data per the specified day(s), the output becomes active by design norm. The output or transmit criteria is determined by entries in the Transmit setup menu. The output is continuously feed with the TS data that is delayed by the amount of time specified for the buffer, the Recovery Buffer (days) setting. The Disaster Recovery function serves as a delay TS signal path from a specified input to a specified output. This occurs normally while the Disaster Recovery feature is in a non-disaster mode constantly feeding a delayed output to downstream equipment. This condition is indicated by a "Normal Operation" indication in the state field of the Buffer.

#### **Disaster Recovery Explained: In Disaster Mode**

Once the buffer is filled and the delayed output is active in the Normal Operation state, the Disaster Recovery feature is ready to replace lost programming in the event the input TS stream



is lost. The TSS 6220 provides input TS stream monitoring of the normal programming. In the event the incoming TS stream programming is lost, (Disaster!) the buffered data is ready and available to continue or maintain the output. This requires that the output be switched to a Disaster Recovery mode or output. This can be done manually, or be setup for the Disaster Recovery feature to do automatically after a specified time elapses in which the input remains lost. The manual or auto mode is set in the Disaster Trigger Mode field

Configure Disaster Recovery 2						
General	Receive	Transm	it			
Alias:			Disaster Recov	very 2		
Recovery Buffer(days):			1	\$		
Disaster Trigger Mode:			Auto	-		
Trigger Del	ay(minutes):		5	÷		
Gap Mode:			No Output	*		
Source:						
				Browse		

of the General setup menu. The delay time waiting for the input TS stream to recover prior to automatically switching to the Disaster Recovery output is set in the Trigger Delay (minutes) field of the General setup menu. See section 7.3.1 in this manual for more details.

When the output is switched to a Disaster Recovery mode, either manually or when the auto criteria (minutes) elapses, the output is in the Disaster Recovery mode. This mode is indicated by an "In Disaster" indication in the Buffer's State field. In this state, the incoming buffer recording is suspended. The TS stream data in the buffer is routed to the output. The timestamps of the buffered stream data is shown in the "Originally Aired" Transmit field. The output bit rate and green status further indicate a normal disaster mode output.

Disa	Disaster Recovery 1							•
्रिः C	onfigure Disast	er Recovery					🤤 Re	move
٠	0	Receive	Physical Connector: eth1		239.192.1.50:1050		19.397 Mbps	۲
۲	• 🔺 🔺	Buffer	State: In Disaster		1 Days 00:00:00	Buffer Size: 398.5 GB		۲
۲	0	Transmit	Physical Connector: eth1		239.192.10.222:10009	Originally Aired: 2018-06-26 08:27:34	19.400 Mbps	۲

You can also launch into the Disaster Recovery mode manually by clicking on the Start Disaster icon A located in the Buffer row of the recovery panel. When a Disaster Recovery mode is active, the Transmit row status light at the right remains green and the output bitrate indicates the output bitrate. The Originally Aired field in the Transmit row of information for the Disaster Recovery indicates the timestamped date/time of the buffered TS stream data which is being output. You can reference this date/time to determine at what time the original programming was buffered. Should the Disaster Recovery reach the end of the buffered memory, it loops to the beginning of the buffer and continues to output

Ħ	•	🛕 👍 Buffer	State: Normal Operation	1 Days 00:00:00	Buffer Size: 398.6 GB		۲
Ħ	0	Transmit	Physical Connector: eth1	239.192.10.222:10009	Originally Aired: 2018-07-12 14:53:20	19.414 Mbps	۲

When the Disaster Recovery becomes active it stops recording the input to the buffer. This is indicated by the grey (inactive) status indicator light in the Receive information row. The buffer status indicator light turns red and the "State" field in the Buffer information row shows "In Disaster."

To manually end the Disaster Recovery output click on the Stop Disaster Recovery icon located next to the Start icon. This icon manually stops the Disaster Recovery mode playout from the buffer. You cannot end the Disaster Recovery automatically. Upon manually ending the Disaster Recovery mode, the buffer begins recording to recover the specified buffer time and programming delay. During this recovery time, the gap mode, as user defined, is active to the defined output.



# 8 File Viewing Panel

The File Viewing Panel provides a convenient reference to view the media drive folders and files. It provides navigation to all directory levels. It provides information of each file including the naming, type, size, bitrate and duration information. This section describes the navigation tools and information provided.

The File Viewing Panel does not provide file or folder management. All folder and file management functions are performed using FTP or SMB. FTP or SMB provides file loading to the media drive, folder creation and naming. It provides file management including renaming, moving, and deletion. The next sections in this chapter describe how to access and use FTP and SMB.

B C 15 D in Reporting	About		Е	F		
Fil. + Con. ol Panel To upload or manage filas go to ftp://10.0.53	46					
$\Rightarrow$ $\uparrow$ /				💈 Search ro	ot	
r 🚔 root	File Name	Туре	Size	Bitrate	Duration	
demo	10_HD_VBR2_a.trp	.trp	178.8 MB	25.00 Mbps	00:01:00	-
	140mbps_4pgms_BIG.trp	.trp	2.4 GB	140.00 Mbps	00:02:29	
sdi2x	160_5min.trp	.trp	5.6 GB	160.00 Mbps	00:05:00	
iai small ts	1_out_100_dropped_packets_227_10_20_30_8192.pcap	.pcap	59.3 MB			E
	20ElectronicsPrograms0x21and0x31_1secMissingEach.trp	.trp	2.8 GB	139.93 Mbps	00:02:51	
	21Paris_1.trp	.trp	200.0 MB	24.13 Mbps	00:01:09	
	239_150_003_1-10000PBS480i-B.cap	.cap	1021.1 B	N	N	<b>۱</b>
G	Н	I		1 //	K (	)

#### **File Viewing Panel Descriptions**

The File Viewing Panel is structured much like that of a PC based file viewing application. It differs in the information provided regarding the file information providing information important to managing the TSS 6220's output. The following chart provides details of the fields and features of this panel.

To view folders or files, click on the folder in the root directory. Or, you may click on the navigational icons to move back or up in the folders directory.

Item or Field Name	<b>Button/Action</b>	Description
A. Back	Click back icon 🚝	Moves back one level in the directory
B. Forward	Click forward icon ⇒	Moves forward in the directory level. When grayed out – the action is not available file
C. Up	Click Up Icon 👔	Moves up in the directory. When grayed out the action is not available
D. Directory	Not selectable	Shows the selected file/folder hierarchy with left to right order
E. Refresh Screen	Click on 🤔 icon	Refreshes the page and contents
F. Search Folder/File hierarchy	Enter text in the Search root field	Click on field – enter folder or file names to search the directory



G. Folder/File Directory	Not a Selectable field 💌 🖻 root	This field shows the selected file/folder hierarchy
H. File Name	Not a Selectable field	Indicates a list of all the files by name of in the selected folder
I. File Type	Not a selectable field	Indicates the type of file by suffix for each of the files
J. File Size	Not a selectable field	Indicates the memory size of the named file
K. File Bitrate	Not a selectable field	Indicates the playout bitrate of the file when added to the playlist and output from the TSS 6220
L. File Duration	Not a selectable field	Indicates total duration of the play file

# 8.1 FTP - SMB Loading Play Files to the TSS 6220

The TSS 6220 requires stream files and PCAP files to play out as scheduled events. The files must be loaded onto the internal storage drives of the unit. This is done using FTP (File Transfer Protocol), a common protocol used for the transfer of computer files between a client and server on a network. Or, you may use Samba (SMB), a network file sharing protocol implemented in windows.

S TSS 6220 × +					
( 10.0.7.112/#productTabs:filesTab			<ul> <li>The Internet &gt; 10.0.7.112</li> </ul>		• + Search 10.0.7.112 P
		File Edit Vie	sw Tools Help		
Sencore TSS 6220		Organize -		/	8: • 0
Logged in as: admin			1MB_TS_Rate_h.264_Hawaii.trp	auto_latency_test.ts	balloons_hevc_1080p30_420_08bit_ 12Mbps_L4.0_MT_3.trp
Play Schedule Files Admin	Reporting About	8	balloons_hevc_1080p30_422_08bit_ 16Mbps_L4.0_HT_1.trp	balloons_hevc_1080p60_420_08bit_ 30Mbps_14.1_HT_3.trp	bbb_1080p_c.ts
Files Control Panel		1 <b>*</b> 1			
💮 File Transfer Management To uploa	ad or manage files go to f <mark>o:#10.0.7.112</mark>	1	fransat.ts	HEVC rainfadev2.ts	amonty_python.trp
I /			•		•
v 📄 root	File Name		NASA_HEVC4K_5min_SPTSv2.trp	ScheduledPlayout.1	SILENT_BOB.trp
	1MB_TS_Rate_h.264_Hawaii.trp		starz hevc 1080p30 420 08bit 3.5M		
	auto_latency_test.ts	i i	bps_MP_L4.0_MT_9.trp		
	balloons_hevc_1080p30_420_08bit_12Mbps_L4.0_MT_3.trp	1			
	balloons_hevc_1080p30_422_08bt_16Mbps_L4.0_HT_1.trp	8			
	balloons_hevc_1080p60_420_08bt_30Mbps_L4.1_HT_3.trp	1			

The FTP server address is located in the File Viewing Panel (Files tab) in the TSS 6220's web GUI. Look near the top of the page under the Files Control Panel header. (See circle in illustration) Click at the end of the field and drag the mouse to highlight the ftp address. Copy the address – (Keyboard Ctrl-C key sequence). When using SMB (Samba), copy only the IP address.





Open a Windows Explorer or File Explorer application window on your PC. In the header of this application – paste in the ftp address listed in the TSS 6220's File menu on the second line below the Files Control Panel heading. Paste the ftp address – (Ctrl V). For SMB protocol, enter backslash key entries and paste only the IP address listed into the Control Panel address line of Windows Explorer. See example below.

Example ftp shown: ftp://10.0.15.60. Example SMB (Samba) shown: \\10.0.7.48

Press the Enter key on your keyboard.

Upon connection to the ftp server you will be prompted for a username and password: The default username and password is shown below. Please see the next section in this chapter on how to create a Username and Password to provide protection against unwanted file transfers or deletions.

Default Settings:

User Name: root Password: Note: The Password field is left blank or no entry as shown.

File Edit View Tools Help	
Organize 🔻 New library	
✓ ★ Favorites ■ Desktop	Cibraries
lindowe	



00.00 A		×
og on A		
<b>%</b>	Either the server accepted.	does not allow anonymous logins or the e-mail address was not
	FTP server:	10.0.15.60
	User name:	root
	Password:	
	After you log on	, you can add this server to your Favorites and return to it easily.
Δ	FTP does not en server. To prote	crypt or encode passwords or data before sending them to the cct the security of your passwords and data, use WebDAV instead.
	Log on anony	mously Save password
		Log On Cancel
-		

#### The User Name and Password can be changed and managed to provide protection against unwanted file changes by unauthorized users. Please see the next section of this manual.

Once you have entered the User name and Password, click on the Log On field to access the server.

Upon connection to the server you can view the current folders and play files in the TSS 6220.



To transfer files to the server, you may use common window's based file copy and paste techniques or drag and drop techniques. For example, open a 2<sup>nd</sup> version of Windows Explorer application. (Right mouse click on the Windows Explorer icon at the bottom system tray – click on Windows Explorer) You now have both the server window and the Windows Explorer application running. Position both the server window and the Windows Explorer windows beside each other on your PC screen. See example below. Click on a file or folder in the Windows Explorer window and drag it into the right side of the server window. The folder or file is transferred.



Kropuenske,	ulen 🖡 My Videos 🖡	C3 () ♥ 🔮 ト The Internet ト 10.0.15.60 ト					
File Edit View Tools He	elp	File Edit View Tools Help					
Organize 👻 🛕 Play with	VLC media player 👻 Share with 👻 Burn New folder	Organize 🕶					
★ Favorites     ■ Decistop     Decistop     Decistop     Downloads     Downloads     Downloads     Downloads     Documents     Muric     Muric     Mev Litrary     Fectures     Muric     Dowreinn	IDES-507-69 09.02 Sencere DTU-236_RFXpert_StreamXpert.mp4     IDES-58-12.09.03 FM Meter Training-SLML476CM.mp4     ACOLORADOLTRP     COLORADOLTRP     COLORADOLTRP     Constant, Line17_CollStanPattern_SNR_UMG_DISEMOV     ID Generator, Line17_CollStanPattern_SNR_UMG_DISEMOV     ID Generator, Line17_CollSt_Net#CellStanPattern_SNR_UMG_DISEMOV     ID Generator, Line17_SAR_UMG_DISEMOV     ID GeneratoR_DI	I SUM - SMAF					
Videos	Glentest10_20131123_175529_00001.ts	STA2					
Kropuenske, Glen     Computer     Computer     Computer     DVD RW Drive (D:)     genk (MsZ)thomeFc     pub (MsZ)thpold) (L:     Cal-Test (MsZ) (Ms)     gend (MsZ)(Co)     Product Files (Ms1)     centers (MsZ)(D)	Bi Input, reboot, during 14 Di Uver, Gabot, Lind S, SNR, MG, GJSS, MOV MMusic, Choice, SIRIS Test J, BK/perfRecord/une8.tp Bi TST-CATURE-SINCORE.ts A zelf.rp	The Copying COLORADO.TRP' From 'CLUbers' glenk (Video: to '/' Cancel					

## 8.2 File Transfer Management - User Name and Password

The ability to access the internal media files of the TSS 6220 can be limited to authorized users by setting up a User name and associated password. Click on the File Transfer Management cog wheel located at the left of the page under the Files Control Panel header.

The Configure File Transfer Management menu appears providing entry of enabling/disabling the Username/Password protection. Enabling the Allow Anonymous setting provides anonymous user entry and access. Disabling the Allow Anonymous setting enables the Username/Password user requirement.



Username/Password: These fields provide entry for a Username and associated Password. Click in the field and enter up to 32 characters. Click the Apply field to

enter changes. Click on the Show Password checkbox to check it if you wish to made the characters of the entered Password visible.

Field	Action/Range	Description
Allow Anonymous	Click dropdown arrow, select Enabled or Disabled	Enabled: Username/Password entry not required – allows anonymous user entry. Disabled: Username/Password entry required
Username		Select which network protocol used to transmit to the Syslog server
Password	Click in the box, enter up to 32 characters	IP of the Syslog server. 0.0.0.0 and 255.255.255.255 are not permitted



# 8.3 Managing Play Files & Folders

All file and folder management in the TSS 6220's media drive is done using the FTP or server. For details on getting connected and viewing the files and folders in FTP see the previous section of this manual, Section 8.1. Once connected you can use conventional window's based file management techniques. The management tasks are summarized below:

In FTP you can do the following file/folder management tasks:

Delete a file: Right click on the file you want to delete. Select Delete. Click Yes to confirm.

Rename a file: Right click on the file you want to rename. Select Rename. Click in the name field and edit or rename as desired. Click outside the field to apply

Create a new folder: Click on the File tab at the top. Click on "New". Click on the Folder. Click in naming field and name as desired

Delete a folder: Right click on the folder you wish to delete. Select Delete. Click Yes to confirm.



#### Move a folder: Select the

Rename a folder: Right

Click in the naming field

and edit or enter the

the field to apply.

folder. Click on the Edit tab at the top. Click on Move to folder selection. Browse and select the new folder location. Click Move.

## 8.4 File Viewing Panel – Filter by Type

The files listed in the File Viewing Panel may be filtered by type so you can see only the TS files or the PCAP files. The filter selection is located at the bottom right of the File Viewing Panel. Click on the dropdown to see the filter categories.

By default, the "All Files" type is selected. This shows all file types in the panel. You may select the "PCAP" or "TS Files" selections to filter and view only the chosen file types. In a library that contains many files it can speed up finding a file of interest.

To search for a known file name, use the file search utility that is located at the top right side of the panel. Click on the folder or root index listing at the left, click in the field and enter the file name.





# 9 Record Panel

The Record Panel provides a recording feature to record an incoming transport stream or PCAP. The record feature is a licensed feature of the TSS 6220. When licensed, the Record tab is shown and available to select. To access the Record Panel, click on the Record tab in the header. This chapter provides descriptions and overviews of the features provided by the Record feature.

Play	Record	Schedule	Files	Admin	Reporting	About							
Record	Record Control Panel												
Phys	Physical Connector eth0							٥					
Ac	O Add Transport Stream O Add PCAP Rx Bitrate: 0.023 Mbps 🔵						۲						
Phys	Physical Connector eth1						٥						
Ac	dd Transport	Stream 📀	Add PCAF	•						Rx Bitrate:	20.012	Mbps	۲
æ	۰ 🔅	IP (Streat	m 1 Reco	Glen_	Test_Record_	1test.ts	7	00:10:00	225.136.40.1:10016	٢	19.387	Mbps	۲
±	، چې	IP (Strea	m 2)	Glen_	Rec_Timed.trp	)	7	00:02:00	225.136.40.1:10016	٢	19.387	Mbps	۲

The record feature provides manual recording in which a user may control the start and ending of the record segment. A user may also perform a scheduled record event in which the duration of the timed recording may be specified along with the start date and time.

The TSS 6220 can record a transport stream via an MPEG-IP unicast or multicast as available from the specified Ethernet port. It also can perform a PCAP capture from the specified Ethernet port. Filters are available to specify PCAP IP address and/or port. Multiple recording events can be created and scheduled for each available Ethernet port. Each record license includes the ability to display and configure 10 record events.

To perform a recording requires that you first create and configure a recording event. A recording event can be specified as an immediate event or a scheduled (Date/Time) event. Each created record event is shown as a listing or row of information in the Record Panel. Each listing, or row of information, provides details of the event and user control of the event.

The TSS 6220 offers a hardware configuration that accommodates the addition of an ASI input/output card. With this optional hardware configuration, the Record feature may be used to record a TS stream input via an ASI port. When so equipped, the Record Panel includes sections which list the available ASI input ports. You can configure and record.

Play	Record	Schedule	Delay	Disaster Recovery	Files	Admin	Reporting	About		
Record	ecord Control Panel									
Phys	Physical Connector eth0									
	id Transport	Stream	Add PCAP						Rx Bitrate: 0.014 Mbps	۲
Phys	ical Conne	ctor eth1								0
() Ac	id Transport	Stream 🔘	Add PCAP						Rx Bitrate: Mbps	۲
ASIF	Port 1									0
O En	able ASI Rec	ord								
ASIF	Port 2									0
O En	able ASI Rec	ord								



# 9.1 Record Panel Overview

The Record Panel includes a section for each available Ethernet port. The sections are identified by headers indicating the physical connector port. For example, the Ethernet port 0 is shown as "Physical Connector eth0." If you have added the optional Ethernet ports to the TSS 6220, then two additional sections are included for Physical Connector eth2 and eth3. The shown ports are available for input transport stream or PCAP recording.



Each Ethernet port section includes some common control fields. The following is a general overview of items in the Record Control Panel and description of some common fields. The remainder of this chapter describes definition of the fields and configuration menus.

Item o	r Field	Button/Action	Description
Α.	Add Record event to eth0 port	<sup>O Add</sup> Click on this icon to add a recording event to eth 0	Provides menus to select record file, define input criteria, define record criteria and schedule,
В.	Add Record event to eth1 port	Add Click this icon to add a recording event to eth 1	Provides a menu to select record file, define input criteria, define record criteria and schedule.
C.	Show/Hide port record event info	Selectable, click on the icon	Hides or shows all the record event listings, click to hide or click to show all listed record events
D.	TX Bitrate	Not a selectable field	Shows the receive input bitrate of stream or PCAP to the Ethernet port
E.	A record event	See next section	Row showing a created record event and information regarding the event's current status and input to Ethernet 1 port (eth1)



# 9.2 Record – Information Fields

The Record Panel creates and shows all the record events and scheduled record events. Each event has a row of information and related control functions. There are common data fields for each listed record event forming columns of information in the panel. This section provides a brief definition of the information and/or features provided in each column.



Item or Field Name		Button/Action	Description
Α.	Status & Configuration	Click on this icon	Provides a window showing IP record stream/PCAP status and configuration information
В.	Configuration Menu Select	Click on this icon 🌼	Provides a menu with configuration settings to define the record IP/PCAP stream and IP address
C.	Output Control	Record - click on icon to start recording	Indicates IP/PCAP stream as record active or paused. Click on icon to record. When paused, the current location of the record event of stream or PCAP is maintained.
D.	Stop Control	Click on icon to stop IP/PCAP recording	Stops a recording or timed recording event. Click on record icon to restart – restarts at file recording point.
E.	Stream name or alias	Not selectable, No action	Shows a default output IP/PCAP stream name. See section 9.3 for naming streams. (Alias field)
F.	TS/PCAP File name	Double click to browse to name/select record file	Indicates the current selected/named record file.
G.	Stream configuration	Click on icon to open configuration menu	Provides convenient overview of stream input and record status, some critical settings, and provides some control features. See section 9.6 for details.



H.	Record Status	Not selectable	Indicates a stream recording is active. Indicates record position/time within the start-to- end duration time span. Visual blue highlight indicates stream progress.
I.	IP Address/Port ASI Sync status	Not selectable	Indicates the IP receive address and port. Or, ASI input sync status
J.	Delete icon	Click to delete stream or PCAP	Removes a IP/PCAP stream record event from the record listings
K.	Total Port Bitrate Indication	Not selectable, view only	Indicates Ethernet port receive bitrate of the addition of all recording TS streams and PCAP files. Indicates ASI input bit rate.
L.	Bitrate Indication	Not selectable, view only	Indicates bitrate of the individual record stream/PCAP to the Ethernet port
М.	M. Status Indicator	Not selectable, view only	Indicates status of record event.
			Gray: Inactive – stopped or paused
			Green: Good recording active condition
			Red: Fault record condition

# 9.3 Recording Input TS Stream Configuration - IP

To record a new TS stream requires that you select or name a record file and configure the input receive MPEG-IP parameters. To create a record event click on the Add Transport Stream icon. Note that this selection is available for each of the Physical connector Ethernet ports of your TSS 6220. Select the Add Transport Stream icon in the section corresponding to the Ethernet port in which you want to create a record event.



The Record Panel is simplified for viewing with a Hide/Show feature for each Physical Connector eth section. To show all the record events listed for use with a certain Ethernet port, click on the Show/Hide icon . To hide all the record events listed for a certain Ethernet port, click on the Show/Hide icon .



## 9.3.1 Record Add Transport Stream IP Configuration

To define a recording event, start by clicking on the Add Transport Stream icon. The Add Transport Stream menu opens. The Add Transport Stream record menu provides configuration of the record file, receive MPEG-IP configuration, and how the recording event will start and end. This menu contains 2 sections for configuring the record stream event, the top "Stream" section and the lower "IP" section. This part of the manual covers the settings within the top "Stream" section of the menu. The next part of this manual (section 9.3.2) covers the settings within the IP section of this menu.

The Stream section of the Add Transport Stream menu provides entry of an alias name to provide a convenient reference. Click on the Alias field and enter an alias name. This is not required as the application automatically assigns a name as an incrementing stream number.

The Start Mode field defines how the recording event starts. The event can be configured to start immediately when the record icon is selected in the main menu. If a manual recording start is desired, click in the Start Mode field and select "Immediate." This setting is the default setting.

Add Transport Stream				
Stream				
Alias:	Stream 3			
Start Mode:	Immediate	-		
Start Date:	2018-01-08			
Start Time:	13:16:29			
Duration:	00:10:00			
Record File:	Test_Record_Gl	en10.trp		
		Browse		
- IP				
Receive:	Enabled	~		
Physical Connector:	eth0	~		
Mode:	Multicast	~		
Destination IP:	239.192.0.200			
Destination Port:	10000	-		
IGMP Filter Mode:	Exclude	-		
Add IGMP Address	🤤 Rer	nove All		
IGMP Address		Remove		
	Apply	Cancel		

A timed recording event may be configured by clicking the dropdown in the Start Mode field and selecting Date/Time. When the Date/Time mode is selected, the menu's Start Date and Start Time entry fields become available. Select or enter the start date and the start time in these fields. It is important to note that a timed recording event once defined as an event by clicking the Apply field must be launched or be activated by clicking on the record icon in the main menu. Once the record icon is clicked, the recording event becomes active counting down to the date/time specified in the Start Date and Start time field and recording as specified by the time and duration entries.

The Duration field provides entry of the length of the recording. In the duration field enter the hours, minutes, and seconds of the record duration. Keep in mind that the record duration directly impacts the memory of the system using available storage memory space. The memory used is dependent on the incoming stream bit rate and duration of the recording.

The Output field, and associated Browser button provide definition of the record file name in memory in which to route the incoming transport stream. You can create a new file or select an existing file. To create a new file, click the Browse box Browse . Enter the new file name in the directory or file type field. Click the Apply icon Advise. To select an existing file, click on the Browse icon and select/click the existing file to re-record, replacing the existing record file with the new record event. Click on the Apply icon Advise to select the file.

	bbb_1080p_c.ts	.ts	159.0 MB	-
Glen_Test_Record_1test.ts			TS Files (*.ts, *.trp, *.mpg)	Ŧ
			Apply Cano	cel



The following chart provides a summary of the fields within the Stream section of the Add Transport Stream menu.

Setting	Range	Description
Stream – Alias	Value and or Letter Entry	Provides entry to name or assign alias to identify a record event or stream. If no name is entered, the TSS 6220 assigns an incrementing stream number
Stream – Start Mode	Immediate Date/Time	Selects recording event to start immediately upon manually clicking on the record icon in the main menu. Selecting Date/Time configures for a timed record event in which the record start date and time is defined and the Record event activated by clicking on the record/start icon
Start Date	Enter date in date fields, or click on calendar and click on month/day to select	Selects a day – month/day in which the desired recording event is to be scheduled.
Start Time	Enter time in hours, minutes, seconds	Specifies a time within the day selected in which the recording event will start
Output (Record File)	Creates a new record file name in which the input TS stream is recorded, or, selects an existing recording file to be re-recorded	Creates a new record file name in which the input TS stream is recorded, or, selects an existing recording file to be re-recorded
Browse	Click on field to access stream files in library.	Provides navigation to browse to available stream and PCAP files for selection. Provides name entry field in which to specify a new file name in which to direct the recording event to a memory file

Add Transport Stream – Stream Section Settings Overview


#### 9.3.2 Record Add Transport Stream - IP Configuration

The Add Transport Stream record configuration menu includes an IP section. The IP section provides configuration of the IP input used to receive the MPEG-IP unicast or multicast and route the TS stream to the record file. This section provides descriptions of the settings in this IP section of the menu.

To configure the IP input set the Receive field to "Enabled." Select the Physical Connector or Ethernet port on the TSS 6220 to use as the receive port. Eth 0 and Eth 1 are available. If the added ethernet port option is added to your TSS 6220 then eth 2 and eth 3 will be available in the dropdown.

Select the Mode of the receiver to be Unicast or Multicast. For unicast, specify the destination port in the Destination IP field. For Multicast, specify the Destination IP address, and Destination IP Port.

The IP configuration section further includes settings to provide IGMPv3 features. An IGMP filter may be implemented for use to specify the inclusion or exclusion of source addresses. The TSS 6220 is IGMPv3 compliant. IGMPv3 allows each steam to be seen by the network as relating to a unique source device with a unique IP address, port, and/or MAC address. IGMPv2 is



used to join/leave multicast streams by default if no IGMP Filter addresses are entered in the Add IGMP Address section of the menu. If IGMP Filter Mode addresses are specified then IGMPv3 is automatically used.

Setting	Range	Description
IP Receive	Enable Disabled	This setting allows the user to enable or disable these input stream settings.
IP – Physical Connector	Selects eth0 or eth1, the standard Ethernet ports,	Entry selects which of the available Ethernet ports on the TSS 6220 that the stream record input is assigned
	eth2 and eth3 available if TSS 6220 has added optional ports	
IP Mode	Multicast Unicast	<i>Multicast</i> setting allows the unit to receive multicast streams. Multicast streams originate from the IP range 224.0.0.0 – 239.255.255.255. <i>Unicast</i> allows the unit to receive unicast

The chart below summarizes the selections within the IP section of the Add Transport Stream record menu.



		streams. Unicast streams originate directly from a source device.
Destination IP	Enter Value: 224.0.0.0 - 239.255.255.255	This address is the IP address the source device is sending to for a multi-cast. A unicast would use an IP address less than 224.0.0.0. This setting is only available when receiving a multicast stream. This address is the IP address the source device is sending to.
Destination Port	Enter Value: 0 – 65535	This is the UDP port the source device is sending to. This is the only setting required to receive a unicast stream.
IGMP Filter Mode	Settings: Select Include or Exclude	Used on networks supporting IGMPv3. If this setting is set to <i>Exclude</i> any streams originating from the user defined IP addresses will be rejected. If this setting is set to <i>Include</i> any streams originating from the user defined IP addresses will be received.
Add IGMP Address	Click in field - Enter IP address to include or exclude as per filter mode: Values: 0.0.0.0 - 255.255.255.255	Enter and list IP address of IGMPv3 to include or exclude as a filter setting.
Remove All	Click on icon	Removes or clears all the listed IGMPv3 address



# 9.4 Recording Input PCAP - Configuration

To create a new PCAP record event requires that you select a PCAP file and define its output parameters. To create or add a PCAP record event, click on the add PCAP icon. Note that this selection is available for each of the Physical Connector Ethernet ports of your TSS 6220. Select the add PCAP icon in the section corresponding to the Ethernet port in which you want to use for the IP input. This section describes how to select, add and configure a PCAP record event for recording from one of the available Ethernet ports.



The Record Panel is simplified for viewing with a Hide/Show feature for each Physical Connector eth section. To show all the record events listed for use with a certain Ethernet port, click on the Show/Hide icon . To hide all the record events listed for a certain Ethernet port, click on the Show/Hide icon .

#### 9.4.1 Record Add PCAP - Stream Configuration

To define a recording event, start by clicking on the Add PCAP icon. The Add PCAP menu opens. This record menu provides configuration of the record file, receive MPEG-IP configuration, and how the recording event will start and end. This menu contains 2 sections for configuring the record event, the top "Stream" section and the lower "PCAP" section. This part of the manual covers the settings within the Stream section of the menu. The next part of this manual covers the settings within the PCAP section of this menu.

The Stream section of the Add PCAP menu provides entry of an alias name to provide a convenient reference. Click on the Alias field and enter an alias name. This is not required as the application automatically assigns a name as an incrementing stream number.

The Start Mode field defines how the recording event starts. The event can be configured to start immediately when the record icon is selected in the main menu. If a manual recording start is desired,



click in the Start Mode field and select "Immediate." This setting is the default setting.

A timed recording event may be configured by clicking the dropdown in the Start Mode field and selecting Date/Time. When the Date/Time mode is selected, the menu's Start Date and Start



Time entry fields become available. Select or enter the start date and the start time in these fields. It is important to note that a timed recording event once defined as an event by clicking the Apply field must be launched by clicking on the record icon in the main menu. Once the record icon is clicked, the recording event becomes active. When active, you see it counting down to the date/time specified in the Start Date and Start time fields and recording indications when the specified date/time is reached.

The Duration field provides entry of the length of the recording. In the duration field enter the hours, minutes, and seconds of the record duration. Keep in mind that the record duration directly impacts the memory of the system using available storage memory space. The memory used is dependent on the incoming PCAP data bit rate and duration of the recording.

The Output field, and associated Browser button provide definition of the record file name in memory in which to route the incoming PCAP recording. You can create a new file or select an existing PCAP file. To create a new file, click the Browse box Browse . Enter the new file name in the directory or file type field. Click the Apply icon and select/click the existing file to re-record, replacing the existing record file with the new record event. Click on the Apply icon to select the file.

Setting	Range	Description
Stream - Alias	Value and or Letter Entry	Provides entry to name or assign alias to identify a record event or stream. If no name is entered, the TSS 6220 assigns an incrementing stream number
Stream – Start Mode	Immediate Date/Time	Selects recording event to start immediately upon manually clicking on the record icon in the main menu. Selecting Date/Time configures for a timed record event in which the record start date and time is defined and the Record event activated by clicking on the record/start icon
Start Date	Enter date in date fields, or click on calendar and click on month/day to select	Selects a day – month/day in which the desired recording event is to be scheduled.
Start Time	Enter time in hours, minutes, seconds	Specifies a time within the day selected in which the recording event will start
Duration	Selectable, enter the desired duration of the record event	Specifies and/or Indicates the duration of the record event
Output (Record File)	Creates a new record file name in which the input TS stream is recorded, or, selects an existing recording file to be re-recorded	Creates a new record file name in which the input TS stream is recorded, or, selects an existing recording file to be re-recorded
Browse	Click on field to access stream files in library.	Provides navigation to browse to available stream and PCAP files for selection. Provides name entry field in which to specify a new file name in which to direct the recording event to a memory file

#### Add PCAP - Stream Section Settings Overview



#### 9.4.2 Record Add PCAP - PCAP Configuration

The Add PCAP record configuration menu includes an PCAP section. This section provides configuration of the IP PCAP input parameters used to receive and filter the MPEG-IP input and route the PCAP capture date to the specified record file. This section provides descriptions of the settings in this PCAP section of the Add PCAP menu

To configure the IP input set the Receive field to "Enabled." Select the Physical Connector or Ethernet port on the TSS 6220 to use as the receive port. Eth 0 and Eth 1 are available. If the added ethernet port option is added to your TSS 6220 then eth 2 and eth 3 will be available in the dropdown.

The PCAP record input can be filtered by enabling an incoming Destination IP address and/or a Destination IP port. Set the Destination IP Filter State field and/or the Destination Port Filter State fields to Enabled. Enter the filter IP address and/or the Destination Port Filter value in



their respective entry fields. Click on the Apply field to create the record PCAP event with the specified filter values.

Setting	Range	Description
Receive	Enable	This setting allows the user to enable or disable these input stream settings
	Disabled	inose input stream settings.
IP – Physical Connector	Selects eth0 or eth1, the standard Ethernet ports,	Entry selects which of the available Ethernet ports on the TSS 6220 that the stream record input is assigned
	eth2 and eth3 available if TSS 6220 has added optional ports	
Destination IP Filter State	Enabled Disabled	Enables or Disables an IP Address filter for the incoming PCAP recording
Destination IP	Enter $\frac{1}{2}$	This IP address is used to filter the incoming
Filter Address	255.255.255.255	PCAP recording
Destination Port	Enabled	Enables or Disables an IP Port filter for the
Filter State	Disabled	incoming PCAP recording
Destination Port Filter	Enter Port Value: 0 – 65535	This port value is used to filter the incoming PCAP recording

The chart below summarizes the selections within the IP section of the PCAP section of the Add PCAP record menu.



# 9.5 Recording Input ASI - Configuration

To record a new TS stream requires that you select or name a record file and configure the input receive ASI parameters. To create a record event click on the **C** Enable ASI Record icon. Note that this selection is available for each of the ASI ports. Select the **C** Enable ASI Record icon in the section corresponding to the ASI input port in which you want to create a record event.

If the **Canable ASI Record** icon is grayed out and you cannot select it, the ASI Port is not available for use as an input. It is being used by another application of the TSS 6220. If you hover your mouse over the field it creates a popup message window indicating which application is using the port.

Play		Schedule	Delay	Disaster Recovery	Files	Admin	Reporting	About	
Record	Record Control Panel								
Phys	ical Conne	ctor eth0							٥
() A	dd Transport	Stream	Add PCAP						Rx Bitrate: 0.014 Mbps 🔵
Phys	ical Conne	ctor eth1							0
() A	dd Transport	Stream	Add PCAP						Rx Bitrate: Mbps 🔘
ASI	Port 1								
CO EI	able ASI Rec	ord							
ASI	Port 2	_							•
CO EI	nable ASI Rec	ord							

The Record Panel is simplified for viewing with a Hide/Show feature for each Physical Connector eth section. To show all the record events listed for use with a certain Ethernet port, click on the Show/Hide icon . To hide all the record events listed for a certain Ethernet port, click on the Show/Hide icon .

#### 9.5.1 Record Enable ASI Record – Add ASI Configuration

To define a recording event using an ASI input port, start by clicking on the G Enable ASI Record icon. The Add ASI menu opens providing selections to configure the recording event. The Add ASI record menu provides configuration of the record file, receive ASI port configuration, and how the recording event will start and end.

The Add ASI menu provides entry of an alias name to provide a convenient reference for the record event. Click on the Alias field and enter an alias name. This is not required as the application automatically assigns a name as an incrementing stream number.

The Start Mode field defines how the recording event starts. The event can be configured to start immediately when the record icon is selected in the main menu. If a manual recording start is desired, click in the Start Mode field and select "Immediate." This setting is the default setting. *Note: After you apply the settings within the Add ASI menu by clicking on the "Apply" box at the bottom of the menu, to begin the immediate recording you need to click on the start icon In the Record Panel.* 



Apply Cancel



A timed recording event may be configured by clicking the dropdown in the Start Mode field and selecting Date/Time. When the Date/Time mode is selected, the menu's Start Date and Start Time entry fields become available. Select or enter the start date and the start time in these fields. It is important to note that a timed recording event once defined as an event by clicking the Apply field must be launched or be activated by clicking on the record icon in the main menu. Once the record icon is clicked, the recording event becomes active counting down to the date/time specified in the Start Date and Start time field and recording as specified by the time and duration entries.

The Duration field provides entry of the length of the recording. In the duration field enter the hours, minutes, and seconds of the record duration. Keep in mind that the record duration directly impacts the memory of the system using available storage memory space. The memory used is dependent on the incoming stream bit rate and duration of the recording.

The Output field, and associated Browser button provide definition of the record file name in memory in which to route the incoming transport stream. You can create a new file or select an existing file. To create a new file, click the Browse box Browse icon. Enter the new file name in the directory or file type field. Click the Apply icon and to create the file name. To select an existing file, click on the Browse icon and select/click the existing file to re-record, replacing the existing record file with the new record event. Click on the Apply icon to select the existing file. The file name should now be shown in the Add ASI menu in the Record File field.

$\frown$	📕 bbb_1080p_c.ts	.ts 159.0 N	1B 🔻
Glen_Test_Record_1test.ts			TS Files (*.ts, *.trp, *.mpg)
			Apply Cancel

The ASI Receive field provides an enable/disable selection which routes the incoming ASI input stream to the application or opens the path. Click on the dropdown arrow and select "Enabled" to enable the ASI connection.

The following chart provides a summary of the fields found in the Add ASI menu.

Setting	Range	Description
Stream – Alias	Value and or Letter Entry	Provides entry to name or assign alias to identify a record event or stream. If no name is entered, the TSS 6220 assigns an incrementing stream number
Stream – Start Mode	Immediate Date/Time	Selects recording event to start immediately upon manually clicking on the record icon in the main menu. Selecting Date/Time configures for a timed record event in which the record start date and time is defined and the Record event activated by clicking on the record/start icon
Start Date	Enter date in date fields, or click on calendar and click on month/day to select	Selects a day – month/day in which the desired recording event is to be scheduled.
Start Time	Enter time in hours, minutes, seconds	Specifies a time within the day selected in which the recording event will start
Duration	Selectable, enter the desired duration of the record event	Specifies and/or Indicates the duration of the record event

Add ASI – Settings Overview



Output (Record) File	Creates a new record file name in which the input TS stream is recorded, or, selects an existing recording file to be re-recorded	Creates a new record file name in which the input TS stream is recorded, or, selects an existing recording file to be re-recorded
Browse	Click on <b>Brows</b> field to access stream files in library.	Provides navigation to browse to available stream and PCAP files for selection. Provides name entry field in which to specify a new file name in which to direct the recording event to a memory file
ASI Receive	Selects Enabled or Disabled, click on	Enabled: Routes incoming TS stream on ASI port to record application
	dropdown arrow and click on Enabled or Disabled	Disabled: Opens paths so TS stream on ASI port cannot reach the record application

# 9.5 Record Status & Configuration Information

containing the IP stream or PCAP record listing. The Status and Configuration boxes are added to the Play Control Panel providing status and configuration details. Click on the <sup>a</sup> icon at the same location to hide the status and configuration information.

	Physical Connector eth1	
	O Add Transport Stream O Add PCA	Р
1	🗉 🎲 💿 📧 IP (Stream 1)	COLORADO.TRP

#### **IP Stream - Status and Configuration Windows**

The Status and Configuration windows provide information relative to the recording event. This section summarizes the information provided for a record event of an input transport stream.

The Status window indicates the size or memory space required for the recording. The Countdown to Record listing indicates or counts down the time remaining before the recording event starts. *Keep in mind that once the scheduled recording event is created, you need to click on the Record icon to activate or launch the event. Once activated, the countdown indicates the countdown or time remaining.* The Sync Status, Packets Per Frame and Encapsulation fields provide information regarding the incoming receive transport stream. Should the Sync Status indicate "Unlocked" check the IP configuration settings and/or availability of the MPEG-IP stream.

Physical Connector eth1								٥
O Add Transport Stream O Add PCAP					Rx Bitrate:	20.002	Mbps	
🗉 🌼 💿 📧 IP (Stream 1 Recor	Glen_Test_Record_1test.ts	2	00:10:00	225.136.40.1:10016	٢	19.387	Mbps	۲
🖃 🌼 💿 📧 IP (Stream 2)	Glen_Rec_Timed.trp	2	00:02:00	225.136.40.1:10016	0	19.387	Mbps	
Status           File Size:         1.5 MB           Countdown to Record:         00:00:00           Sync Status:         Locked           Packets Per Frame:         7           Encapsulation:         UDP	Configuration Start Mode: Date/Time Start Date: 2017-12-28 Start Time: 15:30:00 Duration: 00:02:00 Mode: Multicast IGMP Mode: Exclude		IGMP Filter List	ort				

Status Listing	Description
File Size:	Indicates the memory size of the record file
Countdown to Record:	Indicates the time remaining from the current time to the scheduled record time of the record event. This field indicates 00.00.00 when the event is an immediate event or when the scheduled recording time is reached or surpassed. This field shows decrementing time values or time remaining until a scheduled recording event.
Sync Status:	Indicates the source TS stream is being received and TS sync is established.
Packets Per Frame:	Indicates the TS packets per TS frame in the incoming TS stream
Encapsulation:	Indicates receive IP stream encapsulation, RTP, UDP

Summary of the informational fields in the Status window:



The Configuration part of the window provides information regarding the settings for the record event. It indicates if the record event is to be immediate or scheduled for a later specified date/time. If scheduled, it indicates the user selected start date and time along with the duration. The Configuration window also indicates the IGMP receive mode, unicast or multicast, along with IGMPv3 settings and filter addresses.

Summary of the informational fields in the Configuration window:

Configuration Listing	Description
Start Mode: Immediate or Date/Time	Indicates recording event mode: Immediate: Starts recording immediately upon manually clicking on the record icon in the main menu. Date/Time: Records when specified date/time is reached.
Start Date:	Indicates the year – month – date when the recording event is scheduled to start
Start Time:	Indicates the hour – minute – second when the recording event is scheduled to start
Duration:	Indicates the duration of the record capture once it is started.
Mode:	Indicates if input receive is configured as Unicast or Multicast
IGMP Mode:	Indicates Include or Exclude mode for IGMPv3 address entry
Add IGMP Filter	Indicates listed IGMP filter addresses to Exclude or Include for Source Specific IGMPv3

#### PCAP File - Status and Configuration Windows

Click on the <sup>III</sup> icon at the left of the row containing a PCAP stream adds the Status and Configuration windows to the Play Control Panel. There is less status and configuration information for a PCAP file compared to an IP stream file. The following describes the information provided in the Status and Configuration windows. Click on the <sup>III</sup> icon at the same location to hide the Status and Configuration windows.

Physical Connector eth1								٢
Add Transport Stream					Rx Bitrate:	20.001	Mbps	۲
ਭ 🌼 💿 💿 IP (Stream 1 Recor Glen_Tes	st_Record_1test.ts	2	00:10:00	225.136.40.1:10016	9	19.382	Mbps	۲
🖃 🌼 💿 IP (Stream 2) Glen_Re	c_Timed.trp	2	00:02:00	225.136.40.1:10016	٢	19.397	Mbps	۲
🖃 🔅 💿 PCAP (Stream 1) Glen_Ca	pture_Pcap1.pcap	7	-00:58:23		0	89.291	Mbps	۲
- Status Config	guration							
File Size: 0 Bytes Start 1	Mode: Dat	te/Time						
Countdown to Record: 00:58:23 Start I	Date: 201	17-12-31						
Packets Dropped: 0 Start 1	Time: 18:	:01:34						
Durati	ion: 00:	: 10:00						
Destin	nation IP Filter State: Disa	abled						
Destin	nation IP Filter: 239	9.192.0.200						
Destin	nation Port Filter State: Disa	abled						
Destin	nation Port Filter: 100	000						



Status Listing	Description
File Size:	Indicates the total memory size of the recorded PCAP file
Countdown to Record:	Indicates the time remaining from the current time to the scheduled record time of the record event. This field indicates 00.00.00 when the event is an immediate event or when the scheduled recording time is reached or surpassed. This field shows decrementing time values or time remaining until a scheduled recording event.
Packets Dropped:	Indicates the dropped packet count of the incoming p-cap recording

Configuration Listing	Description
Start Mode:	Indicates recording event mode: Immediate: Starts recording immediately upon manually clicking on the record icon in the main menu. Date/Time: Records when specified date/time is reached.
Start Date:	Indicates the year – month – date when the recording event is scheduled to start e
Start Time:	Indicates the hour – minute – second when the recording event is scheduled to start
Duration:	Indicates the duration of the record capture once it is started.
Destination IP Filter State:	Enables or Disables an IP Address filter for the incoming PCAP recording
Enabled or Disabled	
Destination IP Filter:	This address is used to filter the incoming PCAP recording
Destination Port Filter State:	PCAP recording Enables or Disables an IP Port filter for the incoming
Destination Port Filter:	This port value is used to filter the incoming PCAP recording



#### ASI In Record File - Status and Configuration Windows

The Status and Configuration windows provide information relative to the recording event. This section summarizes the information provided for a record event when an ASI port input transport stream is configured.

The Status window indicates the size or memory space required for the recording. The Countdown to Record listing indicates or counts down the time remaining before the recording event starts. Keep in mind that once the scheduled recording event is created, you need to click on the Record icon to activate or launch the event. Once activated, the countdown indicates the countdown or time remaining.

nable ASI Record			
🎲 💿 💿 ASI (Stream 1)	TestRecording1_Glen.trp	2	00:00:07
Status	- Configuration	1	
File Size: 69.6 MB	Start Mode: Immediate		
Countdown to Record: 00:00:00	Start Date: 2018-11-15		
	Start Time: 15:55:14		
	Duration: 00:10:00		

Summary of the informational fields in the Status window:

Status Listing	Description
File Size:	Indicates the memory size of the record file
Countdown to Record:	Indicates the time remaining from the current time to the scheduled record time of the record event. This field indicates 00.00.00 when the event is an immediate event or when the scheduled recording time is reached or surpassed. This field shows decrementing time values or time remaining until a scheduled recording event.

The Configuration part of the window provides information regarding the settings for the record event. It indicates if the record event is to be immediate or scheduled for a later specified date/time. If scheduled, it indicates the user selected start date and time along with the duration.

Summary of the informational fields in the Configuration window:

Configuration Listing	Description
Start Mode: Immediate or Date/Time	Indicates recording event mode: Immediate: Starts recording immediately upon manually clicking on the record icon in the main menu. Date/Time: Records when specified date/time is reached.
Start Date:	Indicates the year – month – date when the recording event is scheduled to start
Start Time:	Indicates the hour – minute – second when the recording event is scheduled to start
Duration:	Indicates the duration of the record event



# 9.6 IP – PCAP - ASI Monitor Panel

For each listed transport stream or PCAP record event listed in the Record Control panel you will find a record control panel you will find a record event and its corresponding input. This panel provides a convenient shortcut to improve viewing of the record event and provides several convenient control options. The following descriptions provide an overview of the features provided.



Descriptions of fields in the Record Event Monitor Panel

Item or Fie	eld Name	<b>Button/Action</b>	Description
А.	File name	Not a selectable field	Indicates the current record file for the transport stream or PCAP record event
В.	Browse	Click to browse or view available files	Provides quick access to view files, to locate current file or search for files. Files cannot be selected in this menu to replace the existing play file. (A file is created in memory after a record event is started.)
C.	Bitrate	Not selectable	Indicates the bitrate of input IP or PCAP stream
D.	Record time indicator	Not selectable	Indicates the current record time position or time within the duration. When record scheduled – indicates negative time



			countdown to record event time
E.	Record Time Duration	Not selectable	Indicates the total time duration or the ending time of the TS or PCAP recording.
F.	Record Bar Progress Indicator	Not selectable	Provides visual indicator that the stream record is active and progressing, Indicates the current record position or time relative to the start time and record duration.
G.	Start Control	Record – click on icon to start output	Indicates IP/PCAP stream as playing or paused. Click on icon to pause or play. When paused, the current location of the stream or PCAP is maintained.
Н.	Stop Control	<ul> <li>Click on icon to stop IP/PCAP recording</li> </ul>	Stops a recording of IP/PCAP input. Click on record icon to restart – restarts at file starting point.
I.	Record Mode	Indicates Immediate or Date/Time mode	Indicates the record mode. If Date/Time is selected the day and time information indicates the record start date and time
J.	Record Date	Value cannot be changed in this menu	Indicates a specified stream record starting - year, month, day
K.	Record Time	Value cannot be changed in this menu	Indicates a specified stream record starting time – hour, minute, second



# 10 Admin

The Admin Panel provides administrative tasks and configuration settings. To access the Admin Control Panel, click on the Admin tab. This section provides descriptions and overviews of the features provided by the Admin panel.

Play Record Schedule Disaster Recovery	Files Admin Repo	orting About					
Admin Control Panel							
🔑 Change Password 📄 Profiles 🛛 🛓 SNMP MIBs	Diagnostics			📑 Upda	te Unit 📑 R	eboot 🛛 🧐 Rese	t to Defaults
🗊 General Settings							0
Dentigure General Settings							
Unit Alias: (No Alias)							
<b>A</b> Notwork							•
Configure Networks Hostnama: (nano) Dafe	ault Catoway: otb0 Brims	Namacanar 1	72 16 0 96				•
Nome A	Mode ID Address	Subset Meek	Cotoway	MAC	Link Statua	Ty Data	Dy Date
Name T	DHCP 10.0.7.38	255 255 0 0	10.0.1.3	0C:C4:7A:C9:AB:40	1Ghos (IIn)	0.003	0.020
🔅 eth1	Static 10.0.0.66	255.255.0.0	10.0.1.4	0C:C4:7A:C9:AB:41	1Gbps (Up)	69.273	20.013
							•
				Ceffuere	Cupped Agrees	ant Evolution:	0010.01.12
Apply License Key				Software	Support Agreen	ient Expiration:	2019-01-13
Option					Supported	State	Instances
TSS 6220 - Plavlist License					Yes	Licensed	16
TSS 62202 - File Record License					Yes	Licensed	100
TSS 62203 - File Play License					Yes	Licensed	1
TSS 62204 - Disaster Recovery License					Yes	Licensed	16
👮 Date / Time							0
🌼 Configure Date / Time							
Update Mode: NTP							
Current Date: 2018-01-08							
Current Time: 15:32:28							
Time Zone: US/Central							
3ª CNMD Communities							•
Configure SNMP Communities							
Read-Only Community: public Read-Write Community: private							
· · ·							

Located directly under the admin control panel are the options for saving/loading profiles, changing the network password, downloading the SNMP MIBs, downloading the diagnostic file, updating the unit software, and resetting to factory defaults. Below is a short description of each feature shown. The sections that follow provide more operational details for each of the features.

Admin Control Panel					
Change Password	Profiles	Diagnostics	📑 Update Unit	Reboot	🤹 Reset to Defaults

**Change Password**: Select to change the unit's network login credentials (User name/password) **Profiles:** Creates, saves, recalls, applies unit configuration files or profiles

**Diagnostics**: Generates a diagnostic file to be used by Engineers when analyzing unit operation. **Update Unit**: Provides updates to the unit's operational software version.

**Reboot:** Provides a reboot of the unit.

Reset to Defaults: Resets the unit to factory preset configuration settings.



# 10.1 Changing Unit Password

The TSS 6220 can be assigned an access password and the current access password can be changed. In order to make changes to passwords, click the

Change Password button. A window will appear to enter the current password and new password. Enter the new password and click the Apply field.

# 10.2 Profile Manager

The TSS 6220 has the ability to save all configuration settings to a file (profile). Multiple profiles can be saved for recall. Profiles can be saved locally, renamed and saved to external storage to be used on other TSS 6220s. Profiles can be used to quickly and easily change the configuration of an TSS 6220 to suit different applications.

New Password:		
Confirm Password:		
	Apply	Cancel

Profile Manager						
🔘 Add 👔 Upload	Last Profile Applied:					
Profile Name 🕇	Download	Rename	Delete			
myprofile	ļ	6	×			
Test1Profile	Ļ	a contraction of the second se	×			
	Appl	y C	Close			

Profile Manager descriptions:

Action	Button	Description
Add New Profile	Add	Adds a new profile from current settings. User must name profile before creation is complete.
Upload Profile	👔 Upload	Allows the user to browse to external storage or workstation to upload profile to TSS 6220.
Last Profile Applied	Last Profile Applied:	Select to see the last profile that was applied.
Apply Profile	Apply	Select a profile from the drop-down menu and click this button. All settings in the selected profile are applied
Rename Profile	Ø	Select a profile from the drop-down menu and click this button. You are prompted for a new name for the profile.
Delete Profile	×	Select a profile from the drop-down menu and click this button. You are prompted to confirm profile deletion.
Download Profile	1	Select a profile from the drop-down menu and click this button. The user will be prompted to select a directory to download the profile.



### 10.3 SNMP MIB Files

The TSS 6220 stores the SNMP MIB files for the currently installed version of software on the unit. These files can be downloaded directly from the TSS 6220 by clicking on the SNMP MBs icon. The screen below will appear where the files can be downloaded and saved off of the unit.

Index of /mib	s/
---------------	----

Name	Last Modified	Size	Туре
Parent Directory/		-	Directory
INET-ADDRESS-MIB.MIB	2017-Oct-17 12:18:30	16.3K	application/octet-stream
SENCORE-CSP-MIB.MIB	2017-Oct-17 12:13:06	86.3K	application/octet-stream
SENCORE-GLOBAL-REG.MIB	2017-Oct-17 12:13:06	2.3K	application/octet-stream
SENCORE-TSS6220-MIB.mib	2017-Oct-17 12:13:04	37.1K	application/octet-stream
SNMP-COMMUNITY-MIB.MIB	2017-Oct-17 12:18:31	15.1K	application/octet-stream
SNMP-FRAMEWORK-MIB.MIB	2017-Oct-17 12:18:31	21.8K	application/octet-stream
SNMP-MPD-MIB.MIB	2017-Oct-17 12:18:31	5.3K	application/octet-stream
SNMP-TARGET-MIB.MIB	2017-Oct-17 12:18:30	22.2K	application/octet-stream
SNMP-USER-BASED-SM-MIB.MIB	2017-Oct-17 12:18:31	38.2K	application/octet-stream
SNMP-VIEW-BASED-ACM-MIB.MIB	2017-Oct-17 12:18:31	33.3K	application/octet-stream
SNMPv2-MIB.MIB	2017-Oct-17 12:18:31	28.6K	application/octet-stream
SNMPv2-SMI.MIB	2017-Oct-17 12:18:30	8.7K	application/octet-stream
SNMPv2-TC.MIB	2017-Oct-17 12:18:30	37.1K	application/octet-stream

To Download: Right-Click, Save Link As or Save Target As

# 10.4 Diagnostics

The "Diagnostics" icon Diagnostics generates a text file for troubleshooting by Sencore support. Click on the Diagnostics icon to generate the file. This file includes the configuration of the system, a log history, licensing and hardware information.

Upon generation of the diagnostic file you are prompted with an "Opening diagnostics.txt" screen. You may open the file to view the contents using the default application (Notepad) or choose to save the file. Click on the selection circle for "Save File" and enter a file name. The file is saved to the computer browsing to the unit.

Success 🛛 😵	Opening diagnostics.txt
The diagnostics file was created. ОК	You have chosen to open:         diagnostics.txt         which is: Text Document         from: http://10.015.18         What should Firefox do with this file?         @ Open with         Notepad (default)         Save File         Do this gutomatically for files like this from now on.         OK



## 10.5 Update the Unit Software Version

Software updates may become available from Sencore to improve performance or add features. Updates to the TSS 6220 are performed through the web interface. Select the Admin tab of the TSS 6220 main menu and locate the Update Unit icon Update Unit at the top of the screen.

#### 10.5.1 Applying Software Updates

A software update file is provided by Sencore and named with a numbered sequence indicating the version. The update file is first downloaded from the Sencore web site or from an ftp site as referred by Sencore support. If downloaded as a zip folder, navigate to the zipped folder and unzip the file by right mouse clicking on the file and selecting "Extract All". Always install or browse to the unzipped folder when applying software updates.



Software updates are governed by Software Subscription Agreements which dictate an expiration date. Software released after the expiration date cannot be loaded. The expiration date is indicated in

the License Information section of the Admin tab. Please see section 10.8 in this chapter on Licensing Configuration for more information.

To apply a software update within the web GUI, select the Admin tab and click on the update unt icon. The current version and any uploaded version is displayed in the Software Versions section. If no version has been uploaded, this field indicates "none." Click on the Upload field and browse to the unzipped folder containing the TSS 6220 update software. Open the file and it will be uploaded and be readied for installation. You will be prompted to approve the installation. Approve and the uploaded software is installed into the TSS 6220. The unit reboots after a software update is complete. The reboot requires that you refresh or re-establish your network connection to the unit.

Action	Button	Description
Upload Software Update	Upload	To upload software updates to the TSS 6220 click this button. The user will be prompted to navigate to an update file. The file will then upload to the TSS 6220. When complete the TSS 6220 with prompt the user to either apply the update or cancel
Delete the Uploaded Software	Delete	Clicking this button prompts the user to confirm the deletion of the software update from the TSS 6220. This will also clear the Uploaded Version status of the Software Versions section.
Update Software to Uploaded Version	Update	Clicking the button starts the software update process. The TSS 6220 will prompt the user to confirm the update. Click Yes to continue or No to cancel.



#### 10.5.2 Rollback - Software Update

The TSS 6220 is capable of reverting back to a previous version of software using the Rollback feature. The TSS 6220 maintains two separate software images; one is the most current version of software with all current settings and the other is the previous version of software with all settings. To perform a rollback, click the Update Unit button and then click the Rollback tab. The TSS 6220 will reboot after the rollback process is complete.

录 Update Unit						
Update	Rollback					
What is This featur installed ve configurati reboot.	Rollback? e will roll the unit software ersion. The unit's settings w on prior to the last update.	back to the previously vill revert to their . Rollback will initiate a				
Previously Installed Version: development-246- g87eabf4 Switch to Previously Installed Version: Rollback						
		Apply Cancol				

 
 Action
 Button
 Description

 Rollback Software
 Rollback
 Clicking this button starts the Rollback process. The TSS 6220 will prompt the user to confirm the rollback or click cancel to stop the process.

#### 10.6 Reboot Unit

The TSS 6220 can be rebooted from the web interface. In order to perform a reboot, click the Reboot button. The TSS 6220 will prompt the user to confirm the reboot. Once the reboot is complete the login screen will appear allowing the web interface to be logged into.

# 10.7 Reset Unit to Factory Defaults

The TSS 6220 settings can be reset to factory defaults. All settings will be returned to the factory defaults except the network management ports TCP/IP settings. All event logs will be cleared. To reset all settings to default, click the Reset to Defaults button. The TSS 6220 will prompt the user to perform the user to account of the New Click the N

confirm the reset as shown. Click the Yes field to proceed with the reset in which all the unit settings will be set back to factory setting. Press No to exist without resetting the unit to factory defaults.



# 10.8 General Configuration

The TSS 6220 can be assigned an alias which is displayed in the upper right-hand corner of the web interface. The alias helps identify which TSS 6220 the operator is logged into. To edit the Unit Alias, click on the

🗊 General Settings		🚅 Configure General Settings		
🔅 Configure General Settings		Alias: Dennis!!		
Unit Alias:		Apply Cano		

click on the Configure General Settings icon. The name can be up to 32 characters. Click the Apply box to enter or edit changes to the alias name.



# 10.9 Network Port Configuration

The Ethernet ports used for web management and stream playout of the TSS 6220 can be configured from the web interface. Select the Admin tab within the TSS 6220 web GUI and locate the Network section as shown below.

	🗖 Network								
$\langle$	Configure Networks Hostname: (none) Defa	ult Gatewa	ay: eth1 Primar,	y Nameserver: 17	2.16.0.86				
	Name 🕇	Mode	IP Address	Subnet Mask	Gateway	MAC	Link Status	Tx Rate	Rx Rate
AD.	∯eth0	DHCP	10.0.15.18	255.255.0.0	10.0.1.3	40:16:7E:79:20:16	1Gbps (Up)	0.005	0.058
1	🔅 eth1	Static	10.99.99.21	255.255.0.0	0.0.0.0	68:05:CA:3A:55:1E	1Gbps (Up)	0.000	0.000

Domain name servers can be configured by clicking the Configure Namerservers button. Choose the default gateway from available unit network ports. IP address and web address entries are accepted as Nameserver addresses.

The network ports of the TSS 6220 can be configured from the web interface. To make changes click the so cog wheel icon at the left in the Network section for the respective port you wish to change.

Configure Networks	
Hostname:	(none)
Default Gateway:	eth1 👻
Primary Nameserver:	172.16.0.86
Secondary Nameserver:	172.16.0.153
	Apply Cancel

	周	Network								•
	÷	Configure Networks Hostname: (none) Defau	ult Gatewa	y: eth0 Priman	y Nameserver: 17	72.16.0.86				
		Name 🕇	Mode	IP Address	Subnet Mask	Gateway	MAC	Link Status	Tx Rate	Rx Rate
$\triangleleft$	្តំរូ	eth0	DHCP	10.0.7.48	255.255.0.0	10.0.1.3	0C:C4:7A:C9:AB:40	1Gbps (Up)	0.003	0.013
100	ŝ	eth1	Static	10.0.0.66	255.255.0.0	10.0.1.4	0C:C4:7A:C9:AB:41	1Gbps (Up)	99.536	0.000

# NOTE: Exercise caution when performing changes to this menu as network communication can be lost with the TSS 6220.

Note the IP address before making changes and prior to changes so you can reestablish a connection to the unit's web GUI.

When selected for changes, the Configure ethX menu is presented and identified by the Network Name. You may set the port to use DHCP to derive IP settings or select static settings and enter the desired IP Address, Subnet Mask and Gateway addresses.

Configure eth0		
Network Name:	eth0	
Mode:	DHCP	-
Static Settings		
IP Address:	10.0.15.18	
Subnet Mask:	255.255.0.0	
Gateway:	10.0.1.3	
	Apply	Cancel

Configure eth1					
Network Name:	eth1				
Mode:	Static -				
Static Settings	ttings				
IP Address:	10.99.99.21				
Subnet Mask:	255.255.0.0				
Gateway:	0.0.0.0				
	Apply Cancel				



Setting	Range	Description
Hostname	Valid characters: A through Z 0 through 9 - (hyphen)	This setting allows the user to define an optional unit Hostname.
Default Gateway	Lists the available Ethernet ports, Eth0, Eth1, etc. for selection	Selects the network interface to use as the network gateway
Nameserver	Four decimal octets XXX.XXX.XXX.XXX	Allows entry of a primary and a secondary nameserver IP address. To disable set to 0.0.0.0
Mode	DHCP Static	Setting to <i>DHCP</i> will allow the network assign an IP address automatically to the TSS 6220 (if supported). Setting to <i>Static</i> allows the user to manually define all TCP/IP settings for the management port.
IP	Four decimal octets: XXX.XXX.XXX.XXX	This option is only available if Static Mode is set. This is the IP address assigned to the management port.
Subnet Mask	255.0.0.0 – 255.255.255.254	This option is only available if Static Mode is set. This is the Subnet Mask assigned to the management port.
Gateway	Four decimal octets: XXX.XXX.XXX.XXX	This option is only available if Static Mode is set. This is the Gateway address assigned to the management port.

Configure Networks and Configure Network Port Settings

# 10.10 Licensing Configuration

Features of the TSS 6220 require a license in order to be functional. Licensing information is shown in the web GUI within the Admin tab and License Information section. This section also provides license management in which new licensing may be applied to add features.

🗊 License Information				٥
🎜 Apply License Key	Software St	upport Agreem	ent Expiration:	2019-06-23
Option		Supported	State	Instances
TSS 6220 - Base Platform		Yes	Licensed	1
TSS 62201 - Playlist License		Yes	Licensed	16
TSS 62202 - File Record License		Yes	Licensed	100
TSS 62203 - File Play License		Yes	Licensed	1
TSS 62204 - Disaster Recovery License		Yes	Licensed	16
TSS 62205 - Single Transmit Delay License		Yes	Licensed	50
TSS 62206 - Multi-Transmit Delay License		Yes	Licensed	16

The License Information section displays all licenses which are available as well as the following status regarding the listings. *Note: The Disaster Recovery feature is currently under development.* 

License Status: Locked or Unlocked



- License Supported: Supported or Unsupported by the installed hardware
- Instances: Number of the license instances when multiple licenses with the same name are applicable

If licenses need to be applied to the TSS 6220 click Apply License Key icon. The menu below appears where the user can copy and paste the provided license key from Sencore. The currently applied license key, if applicable, can be viewed by clicking the View Current Key button.

📑 Enter License Key			
Enter a new license key	here		
		Apply	Cancel

The TSS 6220 software updates are managed by software support agreements (SSAs). A 1-year SSA is automatically included at the time of the unit purchase and issuance of the base software license. The user is able to apply any software update that was created by Sencore before the SSA expiration date. The expiration date is indicated at the far right of the Apply License key row in the GUI as shown/circled below. Software updates created by Sencore after the indicated SSA expiration date cannot be loaded into the TSS 6220.

🔯 License Information	0
Apply License Key	Software Support Agreement Expiration: 2019-01-13
Option	Supported State Instances
TSS 6220 - Base Platform	Yes Licensed 1
TSS 62201 - Playlist License	Yes Licensed 16

Customers may purchase extensions to the Software Support Agreement period to extend the expiration date. Extensions are available to add 1-year, 2-years, or 4-years of software support time.

# 10.11 Date/Time Configuration

The TSS 6220 provides date and time references. The date/time can be set to synchronize with an NTP server or a manual data and time can be defined by the user. The Date/Time setting is found in the web GUI under the ADMIN tab. Locate the Date/Time section and click the **Configure Date / Time** button to configure the date and time. The date/time values are used to timestamp entries in the Alarm and Event logs under the Reporting tab.

鶭 Date / Ti	ime	٢
🔅 Configure	Date / Time	
Update Mode:	Manual	
Current Date:	07/26/2017	
Current Time:	19:03:03	
NTP Server:	0.0.0	

To set the Date/Time manually, click the Configure Date / Time icon and set the Update Mode to "Manual." Enter the current date and time. Click on the Apply field at the bottom. To update the date and time via an NTP server, click on the Configure Date / Time icon and set the Update Mode to "NTP."

🔯 Configure D	ate / Time	
Update Mode:	Manual	~
NTP Server:	0.0.0.0	
Date:	2018-01-08	1111
Time:	15:36:41	
Time Zone:	(GMT+00:00:00) GMT	Ŧ
Note: Changing	time may prompt you to log-in.	
	Apply Ca	incel



Enter the NTP Server address in the address field. Click the Apply field.

👮 Configure Date / Time			
Update Mode:	NTP -		
NTP Server:	172.16.0.153		
Date:	2018-01-08		
Time:	15:41:13		
Time Zone:	(GMT-06:00:00) US/Central		
Note: Changing	time may prompt you to log-in.		
	Apply Cancel		

Setting	Range	Description
Update Mode	NTP Manual	Setting to <i>NTP</i> uses the local network's NTP server to synchronize date and time. <i>Manual</i> allows the user to define a date and time.
NTP Server	Four decimal octets: XXX.XXX.XXX.XXX Domain Name	This is the IP Address or Domain Name of the local NTP Server on the network. This setting is only available if Update Mode is set to NTP.
Date	YYYY-MM-DD	This setting is the user defined date. A calendar widget can be used to select the data by clicking the setting is only available if Update Mode is set to Manual.
Time	00:00:00 – 24:00:00	This setting is the user defined time. The time is based on a 24-hour clock. This setting is only available if the Update Mode is set to Manual.



## 10.12 SNMP Communities

SNMP Communities define whether users have read-only read-write SNMP rights. These two communities are given unique names. The default names for these communities are:

- Read –Only Community: public
- Read- Write Community: private

🗙 SNMP Communi	ties
🔅 Configure SNMP Co	ommunities
Read-Only Community:	public
Read-Write Community:	private

To modify the names of these communities, click on the Configure SNMP Communities button.

➡ SNMP Community Strings		
Read-Only Community:	public	
Read-Write Community:	private	
	Apply Cancel	

### 10.13 SNMP Trap Managers

The SNMP trap managers are recipients of SNMP traps sent from the TSS 6220. The following menu allows the user to configure the recipient's IP addresses. To add and remove recipients of the SNMP traps click the Configure SNMP Managers button.

I SNI	MP 1	ran	Mar	າລດ	ers

Configure SNMP Managers

SNMP Managers

📢 SNMP Managers	
Add Manager	Delete All
SNMP Manager Address	Remove
	0
Apply	Cancel

Page 132 (139)

Action	Button	Description
Add Manager	😳 Add Manager	Clicking this button prompts the user for the IP address of the SNMP trap manager.
Delete All	C Delete All	Clicking this button prompts the user to confirm the deletion of all SNMP trap manager IP addresses. If the user confirms deletion all SNMP trap manager IP addresses will be removed.
Delete Single Entry	9	Highlight a single SNMP trap manager IP address and click this button to delete the entry. A prompt will appear confirming the deletion of IP address.



# 10.14 Syslog Configuration

The TSS 6220 can be configured to send error and event logs formatted in the syslog protocol to a remote user specified Syslog server. Configuration is provided in the Admin tab section of the TSS 6220 web GUI within the Syslog section.

🖪 Syslog		٢
Configure Sy	yolog	
State:	Disabled	
Network Protocol:	UP	
IP Address:	10.0.0.1	
Port:	514	

To configure the Syslog, click on the Configure Syslog icon <sup>the Configure Syslog</sup>. Enter the IP address and port values along with the network protocol. Enable or Disable the Syslog state as desired.

🖪 Configure Syst	og	
State:	Disabled	Ŧ
Network Protocol:	UDP	-
IP Address:	10.0.0.1	
Port:	514	÷
	Apply	Cancel

Action	Range	Description				
State	Enabled	Enable or Disable sending messages to Syslog				
	Disabled	server.				
Network Protocol	UDP	Select which network protocol used to transmit to				
	TCP	the Syslog server				
IP Address	Four decimal octets:	IP of the Syslog server. 0.0.0.0 and				
	XXX.XXX.XXX.XXX	255.255.255.255 are not permitted				
Port	0 - 65535	Destination port of the Syslog server				



# 11 Reporting Panel

The reporting tab provides listings of unit alarms that are currently active alarms and also a listing of logs that show past alarm activities and conditions. To access the Reporting panel, click on the Reporting tab at the top of the screen. This section provides an overview of the alarm and logging features of the TSS 6220 and shows how to configure the operation.

<b>S</b> se	enco	re TSS 6220						
Logged in as:	nimbe			Time: 13:11:47	Disk Usage:	24.9 GB / 3.7 TB	CPU: 14%	👃 Logout
	Play	Schedule Files At min Reporting About						
	Reporti	ng Control Panel						
	Alarms	Logs					💮 Configure	
	State	Name	Location	Last Changed				
	0	Schedule Playout Play Overlap	Schedule (Schedule 3)	09/22/2017 12:	32:31			
	0	Schedule Playout Play Overlap	Schedule (Schedule 4)	09/22/2017 12:	32:02			
	0	Schedule Playout Play Overlap	Schedule (Schedule 7)	09/21/2017 14:	57:31			

The **Reporting** tab in the TSS 6220 contains logs for active alarms currently affecting the unit and an event log. The active alarms are updated periodically in order to reflect the real-time state of the unit. Once an error is cleared it will be cleared from the active alarms window. The event log can be used to view alarm and event history. Both the active alarm and event logs can be configured to hide or change the behavior of alarms and events.

### 11.1 Active Alarms

Clicking on the Alarms button displays the Active Alarms menu. This list displays all of the active alarms currently affecting the unit. There are four columns in the log that display different types of information.

	Play	Schedule Files	Admin	Reporting	About		
	Reportin	ng Control Panel					
Q	Alarms	Logs					
	State	Name				Location	Last Changed
		Schedule Playout Play	Overlap			Schedule (Schedule 3)	09/22/2017 12:32:31

Title	Description
State	This column displays the nature of the alarm. The ③ icon means the log entry is informational and is not an error. The ④ icon means the log entry is an active alarm.
Name	This column displays the description of the error. The function that is experiencing an error condition is described here.
Location	This column displays the hardware or function that is experiencing the active error.
Last Changed	This column displays the data and time the error was raised. This data and time correlates with the Date and Time settings configured in Section <b>Error! Reference source not found.</b> .



#### TSS 6220 Alarm Conditions:

- **Port Link Loss Error**: Error occurs when the NIC on the Streaming Server Platform NIC is providing a network link for an enabled output IP stream and the NIC loses physical link to the network
- **Port Link Loss Error Ends:** Error condition clears when the NIC regains the link or unit reconfiguration has the NIC no longer providing a network link for an output IP stream or PCAP
- **MPEG/IP Transmit Unicast Receiver Not Found**: Error condition occurs when the streamer is outputting a unicast and cannot communicate with a receiver
- **MPEG/IP Transmit Unicast Receiver Found**: Error clears when the unicast receiver communication occurs
- Schedule Playout Play Overlap: Error condition occurs when the schedule playout time of a stream overlaps or runs into the time scheduled for another stream
- Schedule Conflict: Error condition occurs when one or more streams in a schedule have the same start time
- Dropped Packets Error: Error condition indicates dropped packets on the output
- TS Sync Loss: Record Error condition indicates inability to synchronize to incoming TS stream
- In Disaster: Indicates input, time delay, control conditions meet for Disaster Recovery output to be active

## 11.2 Event Logs

Clicking on the Logs button displays the Event Log menu. This list displays all of the events and alarms that have affected the unit. The TSS 6220 stores logs up to four days. If the unit is rebooted or powered off and on, the event logs are cleared. The logs can be cleared manually by clicking the Clear button. The logs can be downloaded as a .tsv file and saved to an external location by clicking the Countral button. There are five columns in the log that display different types of information.

Play	Schedule	Files	Admin	Reporting	About	
Reporti	ng Control P	anel				
Alarms	s Log	gs				
🤹 Refre	esh 📑 Cle	ar 📑	Download			
Severity	Timestamp		Transition	Location		Message
0	09/22/2017 1	2:32:31	0	Schedule (Sc	chedule 3)	Scheduled Play overlap
0	09/22/2017 1	2:32:02	٢	Schedule (Sc	chedule 4)	Scheduled Play overlap
0	09/22/2017 1	2:31:50	٢	Schedule (Sc	chedule 4)	Scheduled Play overlap - Cleared
0	09/21/2017 1	9:00:56	٢	Schedule (Sc	chedule 1)	Scheduled Play overlap - Cleared
0	09/21/2017 1	8:28:24	٢	Schedule (Sc	chedule 1)	Scheduled Play overlap
0	09/21/2017 1	8:27:46	٢	Schedule (Sc	chedule 1)	Scheduled Play overlap - Cleared
0	09/21/2017 1	5:05:50	٢	Schedule (Sc	chedule 1)	Scheduled Play overlap
	09/21/2017 1	4:57:39		eth1		Link Loss OK



Title	Description
Severity	This column displays the nature of the alarm. The (a) icon means the log entry is informational and is not an error. The (b) icon means the log entry is an active alarm.
Timestamp	This column displays the data and time the error was raised or cleared. This data and time correlates with the Date and Time settings configured in Section <b>Error! Reference source not found.</b> .
Transition	This column displays when an alarm transition from a bad to good state. When an error is raised the icon is displayed. When an error is cleared the icon is displayed. When an event takes place the <i>f</i> icon is displayed.
Message	This column displays the description of the error or event. The function or hardware that experienced the event or error is described here.
Location	This column displays the hardware or function that experienced the alarm or event.

#### TSS 6220 Event Conditions:

- **Unit Booted**: The Streaming Server Platform shall notify the user when the unit is booted.
- Unit Shutdown Event: Indicates the unit shutdown
- Update Succeeded Event: Notifies when a software update has succeeded.
- Update Failed Event: Notifies when a software update has failed.
- NTP Updated Event: Notifies when the system time has been updated by an NTP server.
- Date/Time Changed Event: Notifies when the system time has been changed by the user.

# 11.3 Configuring the Logs

The TSS 6220 allows the user to configure alarms and events. Events and alarms can be hidden or set to send SNMP traps. To configure these options, click the Configure button while in the Logs section of the Reporting tab.

The **conditions** tab allows the user to configure the alarms reported by the TSS 6220. Check the boxes of the listed conditions you wish to include in the logs and/or

Set Viewer Time Offset       ±00:00 +       HR         Conditions       Events         Name ↑       Location ↑       Log ✓       Severity       Alarm ✓       SNMP Trap         Dropped Packets Error       eth0       Ø       Error       Ø       Ø       Error       Ø         Dropped Packets Error       eth1       Ø       Error       Ø       Ø       Error       Ø       Ø         Dropped Packets Error       eth1       Ø       Error       Ø <t< th=""><th>Sonfigure Conditions and Ex</th><th>vents</th><th></th><th></th><th></th><th></th><th></th></t<>	Sonfigure Conditions and Ex	vents					
Conditions         Events           Name ↑         Location ↑         Log ✓ Severity         Alarm ✓ SNMP Trap           Dropped Packets Error         eth0         Error         Image: SnMP Trap           Dropped Packets Error         eth1         Error         Image: SnMP Trap           MPEG/IP Transmit Unicast Receiver         Playout MPEG/IP (Stream 1)         Error         Image: SnMP Trap           MPEG/IP Transmit Unicast Receiver         Schedule (Schedule 1)         Error         Image: Schedule (Schedule 3)         Error         Image: Schedule (Schedule 4)         Error         Image: Schedule (Schedule 5)         Error         Image: Schedule (Schedule 5)         Error         Image: Schedule (Schedule 5)         Image: Error         Image: Schedule (Schedule 7)         Image: Error	Set Viewer Time Offset: ±00:00	♣ HR					
Name ↑         Location ↑         Log ✓         Severity         Alarm ✓         SMMP Trap           Dropped Packets Error         eth0         ✓         Error         ✓                 ShMP Trap	Conditions Events						
Dropped Packets Error       eth0       Error       Image: Provide the ethn of th	Name 🕇	Location 1	Log 🗹	Severity	Alarm 🗹	SNMP Trap	
Dropped Packets Error       eth1       etror       error       error         MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 1)       error       error       error         MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 3)       error       error       error         MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 3)       error       error       error         MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 4)       error       error       error         MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 5)       error       error       error         MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 5)       error       error       error         MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 7)       error       error       error         MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 7)       error       error       error         Port Link Loss Error       error       error       error       error       error       error         Schedule Playout Ray Overlap       Schedule (Schedule 1)       error       error       error       error         Schedule Playout Play Overlap       Schedule (Schedule 3)       error       error       error	Dropped Packets Error	eth0	$\checkmark$	Error	~		^
MPEG/IP Transmit Unicast Receiver       Playout MPEG/IP (Stream 1)       Image: Error       Image: Im	Dropped Packets Error	eth1	$\checkmark$	Error	$\checkmark$		
MPEGIP Transmit Unicast Receiver       Schedule (Schedule 3)       Error       Image: Schedule (Schedule 3)       Error       Image: Schedule (Schedule 3)       Image: Schedule (Schedule 4)       Image: Schedule (Schedule 5)       Image: Schedule (Schedule 5)       Image: Schedule (Schedule 5)       Image: Schedule (Schedule 5)       Image: Schedule (Schedule 7)       Image: Schedule 7)	MPEG/IP Transmit Unicast Receiver	Playout MPEG/IP (Stream 1)	$\checkmark$	Error	$\checkmark$		
MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 4)       Error       Image: Schedule (Schedule 4)         MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 5)       Error       Image: Schedule (Schedule 5)         MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 5)       Error       Image: Schedule (Schedule 5)         MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 7)       Error       Image: Schedule (Schedule 7)         Port Link Loss Error       Error       Image: Schedule (Schedule 7)       Error       Image: Schedule Schedule (Schedule 7)         Schedule Playout Play Overlap       Schedule (Schedule 1)       Error       Image: Schedule (Schedule 3)       Error         Schedule Playout Play Overlap       Schedule (Schedule 3)       Error       Image: Schedule (Schedule 4)       Image: Schedule (Schedule 4)	MPEG/IP Transmit Unicast Receiver	Schedule (Schedule 1)	$\checkmark$	Error	$\checkmark$		
MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 4)              Error	MPEG/IP Transmit Unicast Receiver	Schedule (Schedule 3)	$\checkmark$	Error	$\checkmark$		
MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 5)       Error       Image: Schedule (Schedule 7)         MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 7)       Error       Image: Schedule 7)         Port Link Loss Error       Error       Image: Schedule 7)       Error       Image: Schedule 7)         Port Link Loss Error       Error       Image: Schedule 7)       Error       Image: Schedule 7)         Schedule Playout Play Overlap       Schedule (Schedule 1)       Error       Image: Schedule 7)         Schedule Playout Play Overlap       Schedule (Schedule 3)       Error       Image: Schedule 7)         Schedule Playout Play Overlap       Schedule (Schedule 4)       Error       Image: Schedule 7)	MPEG/IP Transmit Unicast Receiver	Schedule (Schedule 4)	$\checkmark$	Error	$\checkmark$		=
MPEG/IP Transmit Unicast Receiver       Schedule (Schedule 7)       Error       Error         Port Link Loss Error       Error       Error       Image: Construction of the	MPEG/IP Transmit Unicast Receiver	Schedule (Schedule 5)	$\checkmark$	Error	$\checkmark$		
Port Link Loss Error	MPEG/IP Transmit Unicast Receiver	Schedule (Schedule 7)	$\checkmark$	Error	$\checkmark$		
Port Link Loss Error     Error     Error     Image: Constraint of the second	Port Link Loss Error		$\checkmark$	Error	$\checkmark$		
Schedule Playout Play Overlap     Schedule (Schedule 1)     Image: Frror     Image: Frror       Schedule Playout Play Overlap     Schedule (Schedule 3)     Image: Frror     Image: Frror       Schedule Playout Play Overlap     Schedule (Schedule 4)     Image: Frror     Image: Frror	Port Link Loss Error		$\checkmark$	Error	$\checkmark$		
Schedule Playout Play Overlap Schedule (Schedule 3)	Schedule Playout Play Overlap	Schedule (Schedule 1)	$\checkmark$	Error	$\checkmark$		
Schedule Playout Play Overlap Schedule (Schedule 4)	Schedule Playout Play Overlap	Schedule (Schedule 3)	$\checkmark$	Error	$\checkmark$		
~	Schedule Playout Play Overlap	Schedule (Schedule 4)	$\checkmark$	Error	$\checkmark$		-
Annual Annua							

alarms. You may also choose which conditions to include or generate an SNMP trap when available.



The **Events** tab allows the user to configure the events reported by the TSS 6220. Each column and its function are described below. A user configured time offset can also be applied to allow viewing the logs in a local time zone. Check the box for the events you want included in the logs or to include or generate an SNMP trap, when available.

Configure Conditions and Events					
Set Viewer Time Offset ±00:00	♣ HR				
Conditions Events					
Name 🕇	Location 1	Log 🗹	SNMP Trap		
Date/Time Changed	Unit	$\checkmark$			
NTP Updated	Unit	$\checkmark$			
Software Update Failed	Unit	$\checkmark$			
Software Update Succeeded	Unit	$\checkmark$			
Unit Booted	Unit	$\checkmark$			
		Apply	Cancel		
		Арріу	Calicei		

Title	Description
Name	This column displays the name of the error or condition. This is informational data; no options can be set here.
Location	This column displays the hardware or function that the alarm or event applies to. This is informational data; no options can be set here.
Log	Checking the box in this column creates an entry in the event log in the case this error or event is raised. If this box is unchecked this error or event will be hidden and not logged if raised.
Log Severity	This column is only available in the <b>Conditions</b> tab This option allows the user to set the severity of the error to Info or Error. If Info is selected in the drop-down box the <b>(a)</b> icon will be displayed in the event log. If Error is selected the <b>(d)</b> icon will be displayed in the event log.
Alarm	This column is only available in the <b>Conditions</b> tab This option allows the user to enable or disable this alarm in the Active Alarms log. If checked the alarm will be displayed in the Active Alarms log if raised. If this box is unchecked this error will be hidden.
SNMP Trap	This column allows the user to send an SNMP Trap if this alarm is raised. If this box is checked an SNMP Trap is sent when this alarm is raised. If this box is unchecked an SNMP is not sent.



# 12 About Panel

The About tab provides information regarding the TSS 6220 including system information, hardware and software options, Sencore contract information, and third-party software information. Click on the About tab to access the About Panel.

Play	Record	Schedule	Delay	Disaster Recovery	Files	Admin	Reporting	About		
About Control Panel										
System Information										
Software Version:         1.3.0.RC1           Unit Serial Number:         0000007 R01           UUID:         0000000-0000-0000-0CC47AC9AB40										
Options										
TSS 6210 (TSS 6210 Base Platform)										
TSS 6220 (TSS 6220 - Base Platform)										
TSS 62201 (TSS 62201 - Playlist License)										
TSS 62202 (TSS 62202 - File Record License)										
TSS 62203 (TSS 62203 - File Play License)										
Ⅲ TSS 62204 (TSS 62204 - Disaster Recovery License)										
TSS 62205 (TSS 62205 - Single Transmit Delay License)										
Ⅲ TSS 62206 (TSS 62206 - Multi-Transmit Delay License)										
S Contact Information										
200 W Sencore Or Sixux Falls, SD 57107 United States 05-978-000 http://www.sencore.com										
T	Third-Party Software Information									

Under the About tab, there are no user definable parameters. Below is a description of the information found in the About Panel.

#### System Information

The System Information section provides the following information regarding the specific TSS 6220.

Software Version: Indicates the current software version. Unit Serial Number: Indicates the unit's serial number UUID: Indicates a unique unit identification number, typically the MAC address.



#### Options

The Options section provides a list of the unit's hardware and software licensing options. Click on the + symbol at the left of each listing for additional drop down informational lines and details of each option. The listings provide useful information regarding the hardware versions and unit configuration. The listings differ depending on the configuration and options of the TSS 6220.

#### **Contact Information**

The Contact Information section provides Sencore company and contact information.

#### Third Party Software Information

The Third-Party Software Information section lists all the TSS 6220's third party software. Information includes the package identification, software version, license numbers and copyright information. Click the box 💷 to show a complete listing.

